

Università degli Studi di Padova

UNIVERSITÀ DEGLI STUDI DI PADOVA Department of Information Engineering

Ph.D. School of Information Engineering Curriculum: Information and Communication Technologies Cycle: 35°

Analysis of multi-sampled pulsewidth modulation and its application in grid-tied converters

Supervisor: Prof. Simone Buso Co-Supervisor: Prof. Paolo Mattavelli

Ph.D. student: Ivan Petrić

Thesis is written with the financial contribution of Fondazione Cassa di Risparmio di Padova e Rovigo

September 2022

Ph.D. Thesis — Ivan Petrić

Abstract

Power electronic systems are essential parts of most modern technologies that depend on converting electrical energy from one form to another. One of such applications is the electric power grid, where more and more grid-tied converters are present, bringing a true paradigm shift in the generation, transmission, distribution, and consumption of electricity.

To enable their highly-sophisticated usage as energy-processors, power electronic converters are actively controlled. For a set hardware design, it is the task of the control system to enable highest dynamic performance, which determines the overall system capabilities. The actuation of any power electronic converter is determined by the modulation strategy, which dynamically adjusts its switching states in order to achieve the desired system behavior. Nowadays, power electronic control systems are typically implemented within digital control platforms, as they offer much higher levels of flexibility and unmatched regulation capacities with respect to analog control systems. However, significant demerits of digital control are the associated delays, which are not present when using analog controllers.

The focus of this dissertation is the multi-sampled pulsewidth modulation (MS-PWM), which relies on executing the control action more than twice per modulation period with a target of achieving analoglike dynamic performance while retaining all the benefits of the digital control systems. This dissertation brings the analysis of some fundamental aspects of this modulation strategy, as well as an investigation of its applicability in grid-tied power electronic converters.

Besides the positive impact of MS-PWM on reducing the control delays, it is known that several nonlinearities arise due to the switching ripple being introduced in the control system. This has resulted in most reported MS-PWM applications to rely on ripple-removal filters, which have a strong negative impact on the dynamic performance. In this dissertation, particular attention is placed on analyzing the MS-PWM nonlinearities caused by jump discontinuities of the modulating waveform. Besides the two known types, another one is discovered and shown to bring a mechanism for limit cycle oscillations. The resulting effects are analyzed, statistically modelled, and a solution for their cancellation is proposed. By examining the impact of the control system on the discontinuities, it is shown how it affects which nonlinearity is introduced and to what extent. This part of the dissertation is focused on providing a methodology for the control system design that minimizes the impact of the MS-PWM nonlinearities, without having to completely remove the switching ripple. It is concluded that the appropriate filtering is crucial for the performance of MS-PWM with lower multisampling factors while, for high-enough number of samples, the effects caused by the discontinuities are strongly reduced. The analysis is verified in simulations and experimentally.

The dissertation continues by exploring noise propagation in digitally-controlled PWM-based power electronic systems. Motivated by the fact that, in MS-PWM, the feedback signal is by definition oversampled, focus is placed on investigating whether the noise impact can be strongly suppressed, while retaining the high dynamic performance. First, it is shown that the MS-PWM systems feature an inherent decimation mechanism, which limits the noise attenuation capabilities. Accordingly, new models for noise propagation in systems with decimation are derived and digital anti-aliasing filters are proposed to reduce the related aliasing. Extensive experimental measurements validate the derived models and show that, with the appropriate digital filters, MS-PWM can bring strong noise suppression without heavily impacting the system dynamics.

Finally, this dissertation examines the potential of MS-PWM to achieve robust stability of gridtied converters. Namely, due to control delays, power converters introduce negative damping at highfrequencies, which may result in harmonic instability, therefore compromising the system operation. To address this, many control strategies have been proposed to introduce active damping (AD) to the system, however, their effectiveness is compromised at very high frequencies. In this dissertation, it is explored how MS-PWM inherently brings the required damping, in an extended frequency range, without introducing additional complexity to the control system. Experimental impedance measurements and tested grid-connected scenarios validate that MS-PWM provides robust stability, outperforming the typically-used AD methods. Additionally, noise sensitivity is analyzed and the derived models show that MS-PWM outperforms AD in this respect as well. These findings are supported by experimental measurements and simulations.

Acknowledgments

I feel fortunate to have been surrounded by so many great people throughout my life. I have been supported by and learned from my entire family, numerous professors, and many friends, some of whom are with me from early childhood and some of whom I have met on this exciting academic path. With such a wind at my back, focusing on the research was greatly eased. It is inadequate to leave out any of you from here and I am truly thankful for having had you with me. However, writing a comprehensive acknowledgment would require many pages, which is why I shall limit myself to mentioning some individuals that had the most direct impact on this dissertation.

First of all, I would like to express my deepest gratitude to my supervisors Prof. Simone Buso and Prof. Paolo Mattavelli. Your immense knowledge gave me the confidence for the scientific explorations that have resulted in this dissertation. Thank you for your attentive guidance, critical opinions, and wise advice that have helped me grow as a scientist and a person. With all challenges of the period behind us, discussions with you were something that brought me great joy and that I always looked forward to.

I wish to further express my appreciation to all the people of the Power Electronics Group of the University of Padova. Prof. Tommaso Caldognetto, thank you for your selfless knowledge transfer and help with a part of my research activities. Prof. Giorgio Spiazzi, thank you for being a member of the annual review committees. Special acknowledgments go to my friends and colleagues, Ph.D. students, with whom I had many wonderful moments and interesting technical discussions: Giovanni Bonanno, Stefano Cabizza, Nicola Zanatta, and Ružica Cvetanović. Special thanks go to Ružica for her help in the final stage of this thesis and the great motivation brought by her strive for knowledge.

My immense gratitude goes to all the people of the power electronics group of the University of California, Berkeley, where I have spent most of the last year as a visiting scholar. Particularly, I would like to thank Prof. Robert Pilawa-Podgurski for hosting me and including me in the exciting and stimulating research activities of his group. Many thanks go to my dear friends with whom I have collaborated mostly: Rod Bayliss, Nathan Brooks, and Rahul Iyer. Our brainstorming sessions, discussions on theory, work in the laboratory, and conference travels were experiences that I will always cherish.

This dissertation has its roots in my time spent at the School of Electrical Engineering (ETF), the University of Belgrade. I would like to thank all my professors, colleagues, and friends who have made this period so enriching and unforgettable to me. Particularly, I would like to express my gratitude to Prof. Slobodan Vukosavić, whose lectures were the main motivation for me to start the research in power electronics, and whose guidance during my first scientific steps was unique and invaluable.

I would like to strongly emphasize the role of my friend Igor Lopušina in my academic development. The challenge of keeping up with you and your exquisite mind is what pushed me to learn so many things.

I also wish to acknowledge my friends and colleagues from the Power Electronics, Machines and Control (PEMC) Research Group of the University of Nottingham. You have made my first year away from home interesting and memorable. Prof. Michele Degano, thank you for your mentorship, friendship, and serenity that is still an inspiration to me. Miloš Lukić, thank you for the wonderful time in Nottingham as well as for all the talks that we had since then.

Here, I would like to express my deepest appreciation to Aleksandra Bijeljac, for her unselfish and wholehearted help with countless activities related to this dissertation and the events surrounding it.

Finally, I want to give special thanks to my parents and sister Zoran, Ana, and Saša, for their unconditional care and devotion. Dad, thank you for being my moral compass and for your time and patience in helping me with all those mathematics and physics school assignments. Mom, thank you for your warmth and tireless commitment to revising so many lessons with me. Salči, thank you for always being protective towards me and for all the beautiful piano music that made studying at home so enjoyable.

> Thank you! — Grazie! — Hvala! Kotor, September 2022 Ivan Petrić

To Miloš Janković

Ph.D. Thesis — Ivan Petrić

Table of contents

Al	bstra	ict				iii
A	cknov	wledgn	ments			\mathbf{v}
\mathbf{Li}	List of Figures xiii					
Li	st of	Table	es			xvii
1	Intr	oducti	ion			1
	1.1	Motiva	vation and objectives			. 1
	1.2	Review	w of previous research			. 4
	1.3	Disser	rtation outline and summary of main contributions			. 8
2	Con	ntrol sy	ystem architecture			11
	2.1	Eleme	ents of digital PWM-based control systems			. 12
	2.2	Pulsev	width modulation			. 13
		2.2.1	Digital PWM resolution			. 14
		2.2.2	Classification based on modulating waveform properties $\ . \ . \ . \ .$. 15
		2.2.3	Classification based on carrier's waveform			. 18
		2.2.4	Small-signal US-PWM modeling			. 20
			2.2.4.1 Sampling instants			. 20
			2.2.4.2 Modulation delay			. 20
	2.3	Bandv	width limitations of PWM-based control systems			. 23
		2.3.1	Small-signal limitations			. 23
		2.3.2	Large-signal limitations			. 24
		2.3.3	Filtering the switching ripple in MS-PWM			. 26
3	Exp	erime	ental set-ups			29
	3.1	Power	converters			. 29
	3.2	Contro	ol platforms			. 31

4	Nor	nlinear	ities cau	sed by the switching ripple	33
	4.1	Introd	luction .		33
	4.2	Conve	erters und	er test	34
		4.2.1	Controll	er design	34
		4.2.2	Impact	of the control system on the modulating waveform shape $\ldots \ldots \ldots$	36
	4.3	Aliasii	ng in digi	tally controlled power converters	38
	4.4	Ripple	e modulat	ion	40
		4.4.1	Gain of	the pulse width modulator $\hfill \ldots \hfill \ldots \h$	40
		4.4.2	Calculat	ing the modulator transcharacteristic for NS-PWM \hdots	42
	4.5	MS-P	WM nonl	inearities caused by modulating waveform discontinuities $\ldots \ldots \ldots$	46
		4.5.1	Multi-sa	mpled control loop settings	46
		4.5.2	Types o	f intersections between the modulating waveform and the carrier	47
			4.5.2.1	Critical duty cycles and critical modulating segments $\ldots \ldots \ldots \ldots$	48
			4.5.2.2	Counter-phase vertical crossings	50
			4.5.2.3	In-phase "vertical" crossings	52
		4.5.3	Duty cy	cle jittering caused by the in-phase "vertical" crossings $\ldots \ldots \ldots \ldots$	53
			4.5.3.1	Explanation of the jittering mechanism	53
			4.5.3.2	Statistical approach for the jitter modeling $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	54
			4.5.3.3	Algorithm for jittering prevention	55
			4.5.3.4	Approximate in-phase conditions for current control loops with one step	
				delay and low-pass feedback filters	57
			4.5.3.5	Examples of operation with duty cycle jittering	62
		4.5.4	MS-PW	M modulating waveform and its discontinuities	67
			4.5.4.1	Calculating the modulator transcharacteristic for MS-PWM $\ . \ . \ . \ .$	67
			4.5.4.2	Discontinuity graphs	69
		4.5.5	Analysis	s of MS-PWM transcharacteristics	74
			4.5.5.1	Measures of MS-PWM nonlinearity	74
			4.5.5.2	Nonlinearity graphs	75
		4.5.6	Experin	nental verifications	77
			4.5.6.1	Impact of MS-PWM nonlinearities in dc-dc converters	79
			4.5.6.2	Impact of MS-PWM nonlinearities in ac-type converters	82
	4.6	Summ	ary		86
5	Ana	alysis c	of noise]	propagation in multi-sampled control systems	87
	5.1	Introd	luction .		87
	5.2	Analy	zed multi	-sampled control system	88
		5.2.1	Convert	er under test	90
		5.2.2	Controll	er design	90

	5.3	Noise	properties and linear propagation modeling	92
		5.3.1	Sources of noise and feedback acquisition errors	92
		5.3.2	Analysis of white noise propagation	93
		5.3.3	Noise measurement organization	95
	5.4	Impac	t of DPWM decimation	97
		5.4.1	Aliasing caused by DPWM decimation	97
		5.4.2	Qualitative approach to anti-aliasing digital filter design)0
	5.5	Exten	ded models that include decimation effects $\ldots \ldots \ldots$)2
		5.5.1	Open-loop systems)3
			5.5.1.1 Model derivation $\ldots \ldots \ldots$)3
			5.5.1.2 Validation in simulations)5
		5.5.2	Closed-loop systems)9
			5.5.2.1 Model derivation $\ldots \ldots \ldots$)9
			5.5.2.2 Validation in simulations	13
	5.6	Exper	imental verification of noise propagation	17
		5.6.1	Benchmark simulation settings	19
		5.6.2	Results for the PI current loop	19
		5.6.3	Results for the PID voltage loop	21
		5.6.4	Comparison between MS-PWM and MR-DS-PWM methods	26
	5.7	Summ	ary	29
6	5.7 Imp	Summ Dedanc	e shaping for robust stability of grid-connected VSCs 15	29 81
6	5.7 Imp 6.1	Summ Dedanc Introd	ary 12 e shaping for robust stability of grid-connected VSCs 13 uction 13	29 81 31
6	5.7 Imp 6.1 6.2	Summ pedanc Introd Contro	ary	29 81 31
6	5.7Imp6.16.2	Summ pedanc Introd Contro 6.2.1	ary	29 31 33 35
6	5.7Imp6.16.26.3	Summ Dedanc Introd Contro 6.2.1 Imped	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 14 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 14	29 31 33 35 36
6	5.7Imp6.16.26.3	Summ Dedanc Introd Contro 6.2.1 Imped 6.3.1	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 14 Equivalent time delay and controller design 14 cance-based stability analysis of grid-tied VSCs 14 Norton equivalent circuit 14	29 31 33 35 36 36
6	5.7Imp6.16.26.3	Summ Dedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2	ary	29 31 33 35 36 36 37
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 14 cl quivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 14 Norton equivalent circuit 14 Passivity criterion 14 sis and experimental verification of admittance properties 14	29 31 33 35 36 36 37 39
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 14 bl of grid-following VSCs 14 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 14 Norton equivalent circuit 14 Passivity criterion 14 sis and experimental verification of admittance properties 14 Measurement set-up 14	29 31 33 35 36 36 37 39 39
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ Dedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 13 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 15 Norton equivalent circuit 14 Passivity criterion 15 sis and experimental verification of admittance properties 15 Measurement set-up 15 Benchmark results for DS-PWM without AD 14	29 31 33 35 36 36 37 39 39 39
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3	ary 11 e shaping for robust stability of grid-connected VSCs 12 uction 13 bl of grid-following VSCs 13 Equivalent time delay and controller design 13 ance-based stability analysis of grid-tied VSCs 13 Norton equivalent circuit 14 Passivity criterion 14 sis and experimental verification of admittance properties 14 Measurement set-up 14 Benchmark results for DS-PWM without AD 14 Results for MS-PWM without AD 14	29 31 33 35 36 36 37 39 39 42
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3 6.4.4	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 14 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 14 Norton equivalent circuit 14 Passivity criterion 14 Sis and experimental verification of admittance properties 14 Measurement set-up 14 Results for MS-PWM without AD 14 Results for DS-PWM with AD 14	29 31 33 35 36 36 37 39 39 42 43
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3 6.4.4 6.4.5	ary 11 e shaping for robust stability of grid-connected VSCs 12 uction 13 ol of grid-following VSCs 13 Equivalent time delay and controller design 13 ance-based stability analysis of grid-tied VSCs 13 Norton equivalent circuit 14 Passivity criterion 15 sis and experimental verification of admittance properties 14 Measurement set-up 14 Results for MS-PWM without AD 14 Results for DS-PWM with AD 14 Comparison of effectiveness 14	29 31 33 35 36 36 37 39 39 42 43 48 51
6	 5.7 Imp 6.1 6.2 6.3 6.4 6.5 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3 6.4.4 6.4.5 Passiv	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 13 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 13 Norton equivalent circuit 14 Passivity criterion 14 sis and experimental verification of admittance properties 14 Measurement set-up 14 Results for MS-PWM without AD 14 Results for DS-PWM with AD 14 Comparison of effectiveness 14 ity implications in grid-connected scenarios 14	29 31 33 35 36 37 39 42 43 48 51 52
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3 6.4.3 6.4.4 6.4.5 Passiv 6.5.1	ary 11 e shaping for robust stability of grid-connected VSCs 12 uction 13 ol of grid-following VSCs 13 Equivalent time delay and controller design 13 sance-based stability analysis of grid-tied VSCs 13 Norton equivalent circuit 14 Passivity criterion 15 sis and experimental verification of admittance properties 14 Measurement set-up 14 Results for MS-PWM without AD 14 Results for DS-PWM without AD 14 Results for DS-PWM without AD 14 Analysis of the equivalent impedance network 14	229 31 333 335 336 337 339 339 339 339 339 339 342 433 345 552 54
6	 5.7 Imp 6.1 6.2 6.3 6.4 	Summ pedanc Introd Contro 6.2.1 Imped 6.3.1 6.3.2 Analy 6.4.1 6.4.2 6.4.3 6.4.3 6.4.4 6.4.5 Passiv 6.5.1	ary 11 e shaping for robust stability of grid-connected VSCs 13 uction 14 ol of grid-following VSCs 13 Equivalent time delay and controller design 14 ance-based stability analysis of grid-tied VSCs 14 Norton equivalent circuit 14 Passivity criterion 14 sis and experimental verification of admittance properties 14 Measurement set-up 14 Results for MS-PWM without AD 14 Results for DS-PWM without AD 14 Results for DS-PWM without AD 14 Analysis of the equivalent impedance network 14 Analysis of the equivalent impedance network 14 6.5.1.1 Experimental measurement of anti-resonant circuit parameters 14	299 31 333 335 336 337 339 339 339 422 433 348 448 551 552 554 554

Conclusions			171
6.7	Summary		170
6.6	Noise sensitivity	v of passivizing current control strategies \hdots	165
	6.5.2.3	Some considerations on validity of single-frequency models $\ . \ . \ . \ .$	164
	6.5.2.2	Impact of AD strategies and comparison of effectiveness with MS-PWM .	162
	6.5.2.1	Comparison between MS-PWM and DS-PWM without AD	156

7 Conclusions

List of Figures

2.1	Illustration of a digitally controlled power electronic system	11
2.2	Block diagram of a single-stage closed-loop digital control system. $\ldots \ldots \ldots \ldots$	12
2.3	Illustration of the PWM process.	13
2.4	Classification of the pulsewidth modulation based on the modulating waveform properties.	15
2.5	Illustration of a digital control system operating with: (a) SS-PWM and (b) DS-PWM. $\ .$	16
2.6	Illustration of a digital control system operating with MS-PWM. \ldots	17
2.7	Illustration of a digital control system operating with MR-DS-PWM	18
2.8	Typically used PWM carrier waveforms	19
2.9	Illustration of the mechanism behind the modulation delay. \ldots \ldots \ldots \ldots \ldots \ldots	21
2.10	Small-signal block diagram of US-PWM modulators	21
2.11	Small-signal block diagram of a digital control system	23
3.1	Converter topologies used in this thesis.	30
3.2	The experimental set-up B configured for the grid-connected application.	32
4.1	A digital current control system of a two-level half-bridge buck converter	35
4.2	S-domain small-signal block diagram of current-controlled buck-type converters	35
4.3	Block diagram of the analog current control system.	37
4.4	Illustration of the chosen sampling instants for SS-PWM, DS-PWM, and MS-PWM. \hdots .	38
4.5	Illustration of NS-PWM for a modulating waveform with the triangular ripple. \ldots .	40
4.6	Illustrations for calculating the modulator's gain.	41
4.7	Illustration of a current control loop, for a buck converter with a constant output voltage.	43
4.8	Illustration of NS-PWM waveforms.	44
4.9	Examples of analytically obtained NS-PWM transcharacteristics for 4 values of τ_D	45
4.10	Illustration of MS-PWM waveforms for $N = 4$	47
4.11	Types of intersections between the modulating waveform m and the carrier w	48
4.12	Types of zones that may appear in MS-PWM transcharacteristics	49
4.13	Illustration of the critical modulating segments and their discontinuities	50
4.14	Illustration of MS-PWM operation with $N = 4$ and counter phase vertical crossings	51

4.15	Illustration of MS-PWM operation with ${\cal N}=4$ and a single in-phase "vertical" crossing	52
4.16	Example of an operating condition that may result in the jitter amplification	53
4.17	Flowchart of the algorithm proposed for cancelling the duty cycle jittering.	55
4.18	Illustration of the critical modulating segments for the positive slope of $w. \ldots \ldots \ldots$	58
4.19	Illustration of the anti-symmetry between modulating waveforms	59
4.20	The in-phase condition boundary lines for the PI current control loops	61
4.21	Illustration of the impact of the filter $G_{fb}(z)$ on the intersections between m and w	62
4.22	Simulation results of modulating waveforms for different tested filter configurations	63
4.23	Simulation results of the reference current step responses for $N = 6. \ldots \ldots \ldots \ldots$	65
4.24	Experimental results of the inductor current spectra for: (a) $N = 4$ and (b) $N = 8$	66
4.25	Experimental demonstration of the anti-jittering algorithm adaptive limits.	66
4.26	Calculated MS-PWM transcharacteristics for $N = 64$ and 4 values of τ_D	68
4.27	Illustration of the approximated procedure for calculating m , based on NS-PWM	69
4.28	Illustration of the time delay impact on the sampled modulating waveform. \ldots	70
4.29	Discontinuity graphs for $N = 4$ and $D_c = 0.5.$	71
4.30	Discontinuity graphs for $N = 6$ and $D_c = 0.33$.	72
4.31	Discontinuity graphs for $N = 8$ and $D_c = 0.25$.	72
4.32	Discontinuity graphs for $N = 8$ and $D_c = 0.5.$	73
4.33	Demonstration of the impact of τ_D on the MS-PWM nonlinearities for $N = 4$	74
4.34	Nonlinearity measures for $N = 4$ and $D_c = 0.5$	76
4.35	Nonlinearity measures for $N = 6$ and $D_c = 0.33$	77
4.36	Nonlinearity measures for $N = 8$ and $D_c = 0.25$	78
4.37	Nonlinearity measures for $N = 8$ and $D_c = 0.5$	79
4.38	Examples of transcharacteristics around $D = 0.33$, for $N = 6$ and 4 values of τ_D	80
4.39	Operation of the tested converter for $N = 4$ and two values of τ_D	81
4.40	Experimental verification of calculated transcharacteristics. $\ldots \ldots \ldots \ldots \ldots \ldots$	83
4.41	Current distortion for DS-PWM and MS-PWM with $N = 4$ and two values of τ_D	84
4.42	Current distortion for various values of N	85
51	Multi compled control system of a da da norma convertor	80
5.1 5.9	Simplified block diagram of the control system	09 80
0.2 5.2	Submatic of the buck converter with the direct current or voltage control	09
5.4	Ellustration of linear processing of a discrete time stackastic signal <i>n</i>	90
0.4 5.5	mustration of linear processing of a discrete-time stochastic signal n	93
5.5 5.6	Plack diagram of the poice measurement set up	90
5.0 5.7	Illustration of the triangular DDWM re compliant its input signal twice per switching period	91
0.1 5 0	must ation of the triangular DF with re-sampling its input signal twice per switching period. Comparison between the simulated magnitude response from n to i_{-} and $\frac{1}{2}U_{-}^{-1}$	90
J.O	Comparison between the simulated magnitude response from n to i_L and $ \pi_{n,i} $.	99
0.9	musuration of the oversampling impact on the white holse r 5D	99

5.10	Impact of the PI and the PID controllers on the total noise power above f_{pwm} 101
5.11	Frequency responses of cascades of different controllers and feedback filters 102
5.12	Open-loop oversampled filtering, followed by an M-time down-sampler
5.13	Equivalent system for the open-loop noise propagation analysis
5.14	Impact of the open-loop decimation for the derivative filter
5.15	Impact of the open-loop decimation for the P controller without filters
5.16	Impact of the open-loop decimation for the P controller with $G_{DLPF}(z)$
5.17	Impact of the open-loop decimation for the PID controller without filters. \ldots \ldots \ldots 108
5.18	Impact of the open-loop decimation for the PID controller with $G^3_{DLPF}(z)$ 108
5.19	Closed-loop system of interest for noise propagation analysis. $\dots \dots \dots$
5.20	Equivalent system for the closed-loop noise propagation analysis
5.21	Impact of the closed-loop decimation for the PI controller without filters
5.22	Impact of the closed-loop decimation for the PI controller with $G_{DLPF}(z)$
5.23	Impact of the closed-loop decimation for the PID controller without filters. \ldots \ldots \ldots 115
5.24	Impact of the closed-loop decimation for the PID controller with $G_{DLPF}(z)$
5.25	Impact of the closed-loop decimation for the PID controller with $G^3_{DLPF}(z)$
5.26	Impact of the closed-loop decimation for the PID controller with $G_{MAF}(z)$
5.27	Oscilloscope screen capture showing switching components over 2 switching periods. \dots 118
5.28	Comparison between the variances and the spectra for the current loop without filters 120
5.29	Comparison between the variances and the spectra for the current loop with $G_{DLPF}(z)$. 121
5.30	Experimental results showing time-domain impact of noise for the tested current loops. $~$. 122 $$
5.31	Comparison between the variances for the voltage loop without filters and with $G_{DLPF}(z)$.123
5.32	Comparison between the variances and the spectra for the voltage loop with $G^3_{DLPF}(z)$. 124
5.33	Comparison between the variances for the voltage loop with $G_{MAF}(z)$
5.34	Spectral results for all tested feedback configurations of the voltage loop
5.35	Experimental results showing time-domain impact of noise for the tested voltage loops $.\ 126$
5.36	Timing illustration for MR-DS-PWM with the MAF and one step computation delay. $\ . \ . \ 126$
5.37	Comparison of noise attenuation for the tested strategies
61	Illustration of a grid-following VSC 134
6.2	S-domain representation of a grid-following VSC and its Norton equivalent circuit 135
6.3	Impact of handwidth on the relative passivity index
6.4	Illustration of the set-up for admittance measurements 140
6.5	Admittance measurements for DS-PWM: (a) $L = 2.5$ mH $\alpha = 0.1$: (b) $L = 1.5$ mH $\alpha = 0.2.142$
6.6	Relative conductance for DS-PWM: (a) $L = 2.5$ mH, $\alpha = 0.1$; (b) $L = 1.5$ mH, $\alpha = 0.2$. 143 Belative conductance for DS-PWM: (a) $L = 2.5$ mH, $\alpha = 0.1$; (b) $L = 1.5$ mH, $\alpha = 0.2$.
6.7	Admittance measurements for MS-PWM: $N - 4$ $L - 25$ mH $\alpha - 0.1$ 144
6.8	Admittance measurements for linearized MS-PWM: $N = 4$, $L = 2.5$ mH, $\alpha = 0.1$. 144
6.0	Admittance measurements for MS-PWM: $N = 8$ $L = 2.5$ mH $\alpha = 0.1$ 145
0.0	114111000100 11000000000000000000000000

6.10	Admittance measurements for MS-PWM: $N = 16$, $L = 2.5$ mH, $\alpha = 0.1$	146
6.11	Admittance measurements for MS-PWM: $N = 32, L = 2.5$ mH, $\alpha = 0.1. \ldots \ldots$	146
6.12	Comparison of experimentally measured conductances: $L = 2.5$ mH, $\alpha = 0.1$	147
6.13	Relative conductance for DS-PWM with D-AD: $L = 1.5$ mH, $\alpha = 0.2$	148
6.14	Impact of the AD derivative gain on the conductance for DS-PWM with M-D-AD	149
6.15	Relative conductance for DS-PWM with DD-AD: $L = 1.5$ mH, $\alpha = 0.2$	150
6.16	Comparison of the operating point impact on the conductance properties. \ldots \ldots \ldots	151
6.17	Comparison of conductances for all tested control strategies	151
6.18	Impedance network at the frequency of the grid anti-resonance	152
6.19	Nyquist plots of the minor-loop gain $Y_i(j\omega)Z_g(j\omega)$ for the tested grid impedances	153
6.20	Circuit used for measuring passive damping of the tested anti-resonant networks	156
6.21	Illustration of the grid-connected set-up	157
6.22	Real part of the VSC input admittance Y_i for DS-PWM	158
6.23	Grid-connected tests: $N=2, \alpha=0.2, L=1.5$ mH, $V_{in}=400$ V, and $v_{pcc,rms}=230$ V.	158
6.24	Grid-connected tests: $N=2, \alpha=0.2, L=1.5$ mH, $V_{in}=250$ V, and $v_{pcc,rms}=110$ V.	159
6.25	Grid-connected tests: $N=2, \alpha=0.2, L=2.5$ mH, $V_{in}=250$ V, and $v_{pcc,rms}=110$ V.	160
6.26	Grid-connected tests: $N=2, \alpha=0.1, L=1.5$ mH, $V_{in}=250$ V, and $v_{pcc,rms}=110$ V.	160
6.27	Grid-connected tests: $N=4,\alpha=0.2,L=1.5$ mH, $V_{in}=250$ V, and $v_{pcc,rms}=110$ V.	161
6.28	Grid-connected tests: $N=32,\alpha=0.2,L=1.5$ mH, $V_{in}=250$ V, and $v_{pcc,rms}=110$ V	161
6.29	Grid-connected tests: $N \in \{2, 16\}, \alpha = 0.2, L = 2.5$ mH, $V_{in} = 400$ V, $v_{pcc,rms} = 230$ V	163
6.30	Experimental demonstration of passivity implications in grid-connected scenarios	164
6.31	Comparison of Nyquist plots for various modeling strategies.	166
6.32	Output power spectral densities due to current and voltage noise sensing. \ldots	168
6.33	Variance of i_L up to $\frac{f_{pwm}}{2}$	169

List of Tables

2.1	Small-signal gain and delay of US-PWM modulators	22
3.1	Hardware parameters of the set-up A	29
3.2	Hardware parameters of the set-ups B.1 and B.2	31
3.3	Hardware parameters of the set-up B.3	31
4.1	Summary of the dc-ac converter results	83
5.1	Controller parameters	91
5.2	Control strategies tested in this section	128
6.1	Comparison of measured conductances for the tested control strategies.	152

Chapter 1

Introduction

1.1 Motivation and objectives

The rise of civilization has been strongly linked with our capabilities of harnessing energy from the surroundings and converting it into a suitable form. From the earliest applications of burning biomass for the purposes of heating and cooking, via creating machinery that helped us surpass our physical limitations, to the highly-sophisticated era of digital information processing, modern society owes its current state to the technological developments related to energy. Starting from the 20th century, participation of electricity in the energy transfer and final consumption has significantly increased and, with the increasing relevance of sustainability, this trend is only expected to continue [1].

The fundamental infrastructure for a widespread use of the electrical energy is the electric power grid, which consists of systems for generation, transmission, and distribution. The electric power grid is experiencing one of the greatest revolutions of the 21st century. Large, centralized, power plants based on fossil fuels are being replaced by distributed generation using renewable energy sources. These sources are by their nature intermittent, hence, storage of the electrical energy is gaining almost the same importance as its generation. Transmission and distribution networks are partly changing from ac grids to dc grids and are becoming significantly more complex in order to accommodate some novel, challenging, services such as, for example, the electric vehicle fast charging stations. Even the end-consumption is changing from dominantly passive to almost entirely electronically controlled loads. Traditional, centralized, systems are expected to be gradually replaced by many smaller, quasi-independent, "smart-grid" blocks that would incorporate localized generation, storage, and consumption.

The enabling technology and the driving force of this change is power electronics. Whether to interface a wind-turbine to the grid, store energy using batteries or hydrogen, replace large capacitor banks for reactive power compensation, or to provide many other functions, the use of grid-connected electronic power converters is unavoidable.

To enable the required flexibility in energy processing, grid-tied converters are actively controlled, with the control systems being divided in several hierarchical levels [2]. Higher levels that, amongst all, determine the needed power flows, adaptation to particular operating conditions, and reactive power and harmonic compensation, are in the end all serviced by the foundation layer, i.e. the "zero-level" control. This control layer is in charge of regulating the currents or voltages of individual converters, using appropriate feedback or feedforward systems. Dynamic performance of these control loops has a determining impact on the overall system performance, particularly regarding stability, resonance damping, and power quality¹.

For a hardware-defined power converter's switching frequency, limited mainly by the losses and the electromagnetic interference (EMI), it is the task of the zero-level control to achieve as high dynamic performance as possible, while maintaining stability, required output waveform quality, and a satisfactory robustness against uncertainties. The zero-level control is intimately related to the power electronic hardware stage and, in parallel with designing the appropriate controller structure, the most important step is determining the suitable modulation (switching) strategy. This dissertation is indeed focused on analyzing a specific modulation strategy, rather than on a particular controller structure design.

In most power electronic systems, digital control is nowadays a standard solution as it offers a much higher flexibility compared to analog control circuits. This is especially important for applications that require sophisticated control laws, controller auto-tuning, communications, various monitoring tasks, additional protection levels, and similar [3].

It is known that the highest dynamic performance is obtained using nonlinear techniques that instantaneously impact the switching state of the converter [4]. Amongst these, the most widely implemented one is the hysteresis controller. Despite its superior dynamic capabilities, a demerit of the hysteresisbased control is the variable switching frequency, which is often unacceptable, due to problems with filtering and excitation of resonances. To solve this issue, the hysteresis control with frequency stabilization algorithms is the topic of extensive research [3]. While commonly implemented using analog circuits, the fully-digital hysteresis control is starting to gain a significant interest, however, it is still not a standardized solution [3]. Another widely-investigated nonlinear control strategy is the model predictive control (MPC) [5]. MPC determines the switching state of the converter as a result of minimizing a certain cost function over a finite prediction horizon, using constrained optimization methods. While offering a relatively simply-posed multi-objective control, several aspects limit its use in power electronics. These include a non-standardized choice of cost functions, incompatibility with tools developed for linear control theory, and a high computational burden in conjunction with a high sampling rate requirement in order to achieve an output waveform quality comparable to other control strategies. Moreover, MPC is highly sensitive to parameter and model uncertainties.

Undoubtedly, most power electronic control systems rely on the pulsewidth modulation (PWM) technique to determine the switching state of the converter [3]. PWM fixes the modulation period, while the state durations are determined by the controller output, i.e. the modulating waveform. A fixed

¹Not to limit the application space to grid-tied converters only, dynamic capabilities of the inner current or voltage control loops have a determining influence on the performance of all power electronic systems, from switched-mode power supplies for CPUs to traction inverters in electric drives.

switching frequency and a well-known output harmonic content allow PWM-based systems to have simple filtering schemes, as well as to avoid the risk of triggering system resonances. This modulation scheme allows the effective use of both linear and nonlinear control structures, derived using frequency-domain or time-domain methods. PWM is also used together with predictive, e.g. dead-beat, controllers [6]. Moreover, PWM behaves closely to a linear wideband amplifier, which effectively enables analyses using the standard control theory.

Traditionally, in digital PWM applications, the feedback is sampled and the controller output is updated once or twice per switching period, which introduces a significant modulation delay [3]. This delay, together with times needed for analog-to-digital conversion and algorithm execution, strongly deteriorates dynamic performance of the digital PWM. However, with the advancement of the technology related to microcontrollers, digital signal processors (DSPs), and field-programmable gate arrays (FP-GAs), increasingly more information can be processed within a certain time interval, using standard control platforms. In regard to processing power, novel wide bandgap devices, based on Silicon Carbide (SiC) or Gallium Nitride (GaN), enable very high switching frequencies [7]. However, in many higher power or voltage applications, the switching frequencies are expected to remain quite low compared to information processing capabilities. Hence, to fully exploit the emerging capacities of digital control platforms, one of the possible directions is going towards analog-like controllers in digital systems [3].

This dissertation follows the trend of investigating smart digital control solutions, by focusing on the *multi-sampled pulsewidth modulation* (MS-PWM). A step towards overcoming the bandwidth limitations of digital control loops is made by adopting the MS-PWM control, in which the feedback is sampled and the control action is executed with a rate higher than double the switching frequency. This reduces the modulation delays and enables analog-like dynamic performance [3].

MS-PWM control was explored in the past and it was shown that it offers improvement of both small-signal and large-signal dynamic capabilities. Besides custom hardware control platforms, its implementation was successfully demonstrated in commercially available DSPs. However, MS-PWM is still a "young" technique with many aspects that require further analysis before it can be widely implemented on an industrial level.

For example, with MS-PWM the switching ripple is unavoidably introduced in the feedback signal. Properties of MS-PWM operated converters, without entirely filtering out the switching ripple, are still not completely clarified. It is known that the sampled switching ripple may cause vertical crossings between the modulating waveform and the PWM carrier, which brings some nonlinearities to the system. However, a clear procedure for determining which nonlinearity will be present and to what extent, depending on the control system and hardware parameters, was not present in the literature. This has brought a motivation to dedicate a part of this research to the analysis of the related MS-PWM nonlinearities. Besides the already-reported types, another nonlinear effect is identified, explained, and statistically modeled in this thesis. Classification of all MS-PWM nonlinearities is given, their impact on a converter's operation is analyzed, and a methodology is provided for their minimization via the appropriate control system design.

Another relevant question is the noise sensitivity of MS-PWM control systems. In return for enabling high dynamic capabilities, most related control strategies bring a significant sensitivity to the measurement noise, which correspondingly limits their practical applicability. However, the fact that MS-PWM is based on oversampling the feedback and that its delay reduction does not come from a derivative action has brought an inspiration to examine whether MS-PWM may jointly offer strong noise suppression and high dynamic performance. In this thesis, it is shown that MS-PWM control systems feature an inherent decimation effect caused by the PWM carrier, which limits the noise suppression. To address this, an analytical model is derived for noise propagation in multi-sampled digital control systems. It is shown analytically and verified experimentally that, with the proposed anti-aliasing digital filter design, MS-PWM may indeed offer both strong noise suppression and fast dynamics.

The final motivation behind this dissertation was to further investigate the applicability of MS-PWM in grid-connected converters, relying on the fundamental properties analyzed beforehand. Namely, the increasing presence of grid-connected converters gives rise to stability issues of power grids. Due to dynamic interactions between controlled converters and passive filters, resonance destabilization may occur. One of the problems, related to high-frequency harmonic instability, comes from the negative damping introduced due to the control delays. This has attracted a significant attention and has driven research in the field of passivity-based control, which focuses on adding damping to the system. However, the developed damping strategies bring certain negative aspects, like ineffectiveness at very high frequencies. In this thesis, it is shown that MS-PWM outperforms these strategies and uniquely offers robust high-frequency stability by inherently reducing the control delays.

1.2 Review of previous research

The idea behind this review is to bring an insight into the development of the multi-sampled pulsewidth modulation, both from the chronological perspective, as well as by grouping together the related, distinct, research paths. Results achieved by other authors that are published in parallel with the writing of this thesis are placed in this section as well. The original contributions of this thesis, and related publications, are highlighted in the following section.

Digital control in power electronics has gained interest starting from the late 1960s [8]. The considered applications mostly involved medium- and high-power inverters in variable speed drives and uninterruptible ac power supplies, i.e. power converters with switching frequencies limited to a few tens of kHz. A comprehensive review of these early works is available in [9]. Regarding digital control in high switching frequency converters, e.g. in voltage regulator modules (VRMs), extensive research has started in the late 1990s and the early 2000s [10–14]. Moreover, at this time, a great research effort was placed on investigating the fundamental principles and specific properties of digital control in power electronics, rather than just its application.

Regarding the pulsewidth modulation, first, analog, solutions were based on the "naturally sampled"

PWM (NS-PWM). In his fundamental work [15], *Bowes* has introduced the "regularly sampled", or "uniformly sampled", PWM (US-PWM) to power electronics, by first processing the analog modulating waveform with the zero-order-hold (ZOH) circuit. In [15], the sample-and-hold action was performed once per switching period, which was termed the "symmetric" modulation. This strategy has brought a linear relationship between the sampled modulating waveform and the switching pulse duration. US-PWM with two samples per switching period was termed the "asymmetric" modulation. The discrete-time sample-and-hold nature of US-PWM lent itself suitable for digital control techniques, particularly in the early years, when the processing capabilities were severely limited [9]. However, it was shown that the introduced ZOH action brings a delay, which is not present in the case of NS-PWM [16,17]. Therefore, for applications that required a very high dynamic performance, analog control with NS-PWM was for long time the only viable strategy.

The background of MS-PWM in power electronics is seemingly perplexed and, when put in perspective, the idea behind it is as old as digital control itself. For instance, the implementation of the digital NS-PWM approximation was reported in [9], although, quite clearly ahead of feasibility for its widespread use. With the advancement of digital control platforms, the idea to reduce the delay associated with the ZOH, by mimicking the NS-PWM, has started to develop more firmly. In [18], *Walker* was the first one to specifically discuss the implementation and some important properties of the digital approximation of NS-PWM. In that paper, the modulation was termed the "re-sampled uniform" PWM and some important practical remarks were reported for the first time. These included handling the, later termed, *counter-phase* and *in-phase vertical crossings*, caused by the discontinuous nature of the modulating signal, in order to prevent the *pulse-skipping* and the *multiple-switching* phenomena. It was shown that the re-sampled US-PWM reduces the modulation delay. Furthermore, it was pointed out that this modulation strategy is highly suitable for multilevel converters with multiple phase-shifted carriers.

A significant expansion of knowledge on MS-PWM was brought by a series of publications from *Corradini et al.* [19–27]. There, the modulation was first termed the multi-sampled PWM, which is used in this thesis as well. The analysis was narrowed down to the two-level modulation and, for the first time, some unique MS-PWM properties that do not have their counterparts in NS-PWM were analyzed. The associated publications have brought the small-signal model of MS-PWM, based on the describing function (DF) approach. This has proven that MS-PWM reduces the modulation delay and, for high multisampling factors, approaches NS-PWM. Besides the dynamic modeling, a great effort was dedicated to analyzing the impact of the sampled switching ripple, which was shown to cause the vertical crossings even in steady-state conditions. It was revealed that the vertical crossings introduce a nonlinear behavior, which impairs the converter's transient response capabilities by reducing the modulation gain or even bringing uncontrollable operating points, termed the dead-bands². These nonlinearities were analyzed through the modulator's static transcharacteristic, whose calculation is presented in [24]. The related publications have also analyzed different carrier waveforms and have demonstrated that the triangular

 $^{^{2}}$ It should be elucidated that the dead-bands are not the same as the pulse-skipping and cannot be avoided by a simple logic in the controller. This is often misconstrued in the literature, even in recent publications [28,29].

carrier modulation is superior for MS-PWM, due to its higher linearity. For suppressing the impact of the modulating waveform discontinuities, in these publications, the groundwork was placed for either completely filtering out the switching ripple [25,26] or dynamically re-adjusting the modulating waveform update instants [23]. In [25] it was shown that MS-PWM has a great potential for grid-tied applications as well.

In [24], it was clearly presented that the MS-PWM nonlinearites are reduced as the multisampling factor is increased. Moreover, in this thesis it will be shown that higher multisampling factors also bring other positive aspects. Therefore, concluding that benefits obtained by increasing the sampling frequency quickly saturate [30] is not generally valid.

Besides the vertical crossings, presence of the switching ripple in the modulating waveform brings other nonlinear effects, which are more evident for high multisampling factors and are practically shared with NS-PWM [31, 32]. For example, in [33], the analysis was provided for determining the maximal bandwidth that retains the smooth modulation in case of highly-oversampled systems. From the early research stages, it was recognized that certain feedback filtering is necessary for the robust implementation of MS-PWM. Starting from the work in [19], it was clear that a compromise must be made between reducing the switching ripple content and keeping a high dynamic performance. In the above-listed publications by *Corradini et al.*, impacts of the moving average filter (MAF), the repetitive filter (RF), and the ripple predictor were analyzed. The application of these filters, and their combinations with phase-leading elements, was revisited in a recent work [34]. A short overview of the relevant ripple filtering strategies is placed in Chapter 2 of this thesis.

In a highly-oversampled class-D audio amplifier application, analyzed in [35], it was shown that, for sawtooth carriers, the ripple impact may be compensated by summing the carrier with the PWM output. This was proven to be equivalent to pre-distorting the carrier based on the output filter. However, this strategy has not found a vast application space and it does not cover the triangular carrier modulation.

An interesting possibility for counteracting the modulating waveform discontinuities and, hence, linearizing the MS-PWM operation, is to replace the ZOH with the first-order-hold (FOH) [36,37]. In theory, this positively impacts the shape of the modulating waveform, bringing the operation closer to NS-PWM. However, this has not yet been widely investigated and some clear demerits are the necessity for a custom hardware control platform as well as an increased high-frequency gain, which amplifies the feedback noise.

In [38, 39], multi-rate system analyses were provided and it was shown that MS-PWM may offer an improved output harmonic content, particularly in applications with very low switching frequencies. Note, however, that it was shown in [34] that MS-PWM may actually increase the output distortion. In this thesis, it is verified that the output waveform quality is indeed deteriorated for low multisampling factors, without ripple filtering, due to the emphasized nonlinear effects related to the vertical crossings. These findings are supported by a recent publication [40], where resonant filters are added to the controller to compensate the impact of the counter-phase vertical crossings on the output waveform distortion. MS-PWM has recently gained more interest for grid-tied converter applications, where the system stability may become compromised due to the interactions between various power electronic systems. One of the first studies on the positive impact of MS-PWM on damping the LCL filter resonance is reported in [41]. In this thesis, research effort was dedicated to investigating the impedance properties of grid-connected converters with MS-PWM control systems. Corresponding findings are supported by later publications that deal with the positive impact of MS-PWM in circulating current control loops of parallel inverters [42] and the impedance properties of systems that employ both MS-PWM and active damping techniques [43].

Regarding direct comparison with other high-performance digital control strategies, in [44], it was shown that a linear single-loop MS-PWM current controller outperforms an oversampled fixed switching frequency finite-control-set model predictive controller (FCS-MPC). It was shown that, compared to MS-PWM, even the improved oversampled MPC implementation suffers worse output waveform quality, higher computational complexity, and higher sensitivity to model uncertainties.

In [3,45], for a voltage source inverter (VSI) current control application, a linear single-loop MS-PWM controller was outperformed by a double-sampled dead-beat controller and a fully-digital implementation of the nonlinear hysteresis controller. However, this does not invalidate the MS-PWM itself, due to some specific limitations of the other two strategies. Namely, the implemented dead-beat, although belonging to the group of predictive controllers, is in fact a proportional controller with a feedforward action based on the output voltage, realized using the double-sampled PWM. Due to its model-based design, the response is significantly deteriorated with the presence of parameter mismatch or unmodeled dynamics. Another down side of the dead-beat control is that, in its basic implementation, it requires an additional measurement, which complicates the required hardware and increases the noise sensitivity. If the output voltage is estimated, sensitivity to model and parameter uncertainties is further increased and the dynamic response is slightly deteriorated. Moreover, the dead-beat control is particularly sensitive to the dead time and requires its compensation. Extremely fast dynamic response of the dead-beat control is aided by the feedforward action, which can be applied to the MS-PWM systems as well. Indeed, in [46–49], it was shown that the MS-PWM dead-beat implementation results in improved disturbance rejection, reduced sensitivity to parameter mismatch and unmodeled dynamics, and an improved large-signal response. Regarding the hysteresis control, this type of bang-bang nonlinear strategy guarantees the fastest possible current response [3]. However, its fully-digital implementation requires very high multisampling factors, high bandwidth sensing, external analog current reference generation, an additional frequency stabilization algorithm, compensation of the dead time and sampling delays, and more [50]. Therefore, although providing an unmatched dynamic response, its implementation requires specific control hardware and a highly optimized design.

In multilevel and interleaved power converter applications, harmonic cancellation may be achieved when using multiple properly phase-shifted carriers [51]. In balanced conditions, the phase-shifted PWM (PS-PWM) results in the multiplication of the switching ripple frequency, which allows for MS-PWM implementation without introducing ripple to the feedback. For such cases, most of the specific MS-PWM nonlinearities are avoided and the modulation properties are quite similar to the single- or doublesampled PWM. An early application of MS-PWM in a three-level converter, with 4 sampling instants that coincide with the average current value, is reported in [52]. Other applications with the phase-shifted MS-PWM and the multisampling factor determined by the increased ripple frequency can be found amongst [28, 53–57]. It should be mentioned that, in response to transients, even in perfectly balanced conditions, the vertical crossings may appear. Moreover, for unbalanced operation with the phase-shifted MS-PWM, which is often met in, for example, flying capacitor multilevel (FCML) converters [58], the ideal frequency multiplication is not achieved and the switching ripple is sampled. MS-PWM properties in related conditions are not yet widely investigated in the literature.

Some other publications that are closely related to MS-PWM and the topics covered in this thesis are reported here. In [59], the multi-sampled space vector modulation (MS-SVM) was introduced and shown to be able to suppress the low-order harmonics in low-switching frequency current source converters. The positive impact of MS-SVM on the output waveform quality was also reported in [60]. In [61], the generation of limit cycle oscillations (LCOs) was analyzed for the MS-PWM control of a buck converter. A recent work on predicting the counter-phase vertical crossings in MS-PWM systems was published in [62]. In [63, 64], it was analyzed how the oversampled filtering, with the decimation to the doublesampled PWM, can be used to strongly suppress the aliasing and the feedback noise. The oversampling also offers a great advantage in parameter estimation and condition monitoring. These topics are outof-scope of this thesis and some related applications can be found in [29].

1.3 Dissertation outline and summary of main contributions

Besides the introduction, this dissertation consists of the following parts.

Chapter 2: Control system architecture. This chapter is practically an extension of the introduction. First, the structure of the digital control system is explained and relevant blocks and signals, used throughout the thesis, are defined. Then, the chapter continues with the important aspects, properties, and definitions related to the pulsewidth modulation. A brief overview of the existing knowledge on MS-PWM, which this thesis builds upon, is presented.

Chapter 3: Experimental set-ups. This chapter describes the hardware set-ups and the control platforms that are used for all experimental verifications in the thesis.

Chapter 4: Nonlinearities caused by the switching ripple. This is the first chapter that features original contributions of this thesis. Based on the previous work, it is known that the switching ripple introduces several nonlinearities to MS-PWM control systems. Consequently, most MS-PWM implementations rely on completely removing the switching ripple using digital filters. However, these filters have a strong impact on the overall performance; hence, there is a motivation to further analyze the MS-PWM operation with a non-negligible switching ripple being present in the feedback. The chapter begins by briefly revisiting the nonlinear phenomena that are present in, but are not uniquely

related to, MS-PWM. This includes the aliasing, which may appear in all digital control systems, and the ripple modulation, which may appear in NS-PWM and MS-PWM. Then, this chapter explores the nonlinearities caused by the jump discontinuities in the modulating waveform, which are only present in MS-PWM. Besides the known effects caused by the counter-phase vertical crossings, another nonlinear mechanism is discovered and analyzed. Its highly-detrimental impact, manifested by duty cycle jittering, is explained and statistically modelled. Then, the focus of the chapter is placed on investigating the connection between control systems settings and the exhibited MS-PWM nonlinearities. The main original contributions found in this chapter are:

- Discovery and analysis of the MS-PWM infinite-gain zones, which bring a limit cycle oscillation mechanism.
- Analytical and numerical methods for determining which discontinuity-related MS-PWM nonlinearity is present and to what extent, depending on the control system parameters.
- Framework for control system design focused on suppressing the impact of MS-PWM nonlinearities, without needing to completely remove the switching ripple.

The material from this chapter is published in two journal manuscripts [65, 66].

Chapter 5: Analysis of noise propagation in multi-sampled control systems. This chapter investigates noise sensitivity of multi-sampled PWM-based power electronic systems. The main motivation is to analyze whether MS-PWM can jointly enable a strong noise suppression and a high dynamic performance. The chapter begins by briefly explaining the main sources of noise that are present in power electronic control systems. Then, a known linear analytical model for white noise propagation is presented and shown to be valid only for single-sampled and double-sampled PWM. For MS-PWM, it is explained that the modulator down-samples the digital modulating signal, which causes aliasing and limits the achievable noise suppression. Correspondingly, design of anti-aliasing digital filters is proposed to reduce the impact of the modulator-related decimation. The chapter continues with the derivation of noise propagation models that include decimation in both open-loop and closed-loop system configurations. The derived models are experimentally verified and the proposed anti-aliasing digital filters are shown to enable strong white noise suppression, without significantly impacting the dynamic performance. This chapter also features a comparison of noise attenuation capabilities of MS-PWM and a commonly implemented technique for oversampled filtering. The main original contributions found in this chapter are:

- Validation that, in terms of noise propagation, the multi-sampled digital pulsewidth modulator introduces decimation.
- Derivation of white noise propagation models that take into account decimation in open-loop and closed-loop systems.

• Proposed design of anti-aliasing filters that enable MS-PWM systems to strongly suppress the feedback noise, with a low impact on the dynamic performance.

The material from this chapter is published in one conference [67] and one journal manuscript [68]. Another journal manuscript [69] is submitted at the time of writing.

Chapter 6: Impedance shaping for robust stability of grid-connected VSCs. This chapter analyzes the effectiveness of MS-PWM in preventing the high-frequency harmonic instability of grid-tied voltage source converters (VSCs), caused by delays of the current control system. The investigation is based on the admittance (impedance)-based stability approach; more specifically, on the passivity analysis. The chapter begins by analyzing the impact of the control system on the VSC admittance properties. Correspondingly, the chapter proceeds with a direct comparison between MS-PWM and typically-used active damping (AD) strategies in effectiveness of passivizing the VSC admittance. The admittance measurement set-up, designed to enable high-precision measurements in a wide frequency range, is explained and it is shown that MS-PWM inherently achieves passivity in a very wide frequency range, while the remaining non-passive zones are by all means negligible. Based on conclusions from Chapter 4, the impact of the MS-PWM nonlinearities is examined and it is shown that the resulting effects are not detrimental for the passivity properties. Moreover, it is verified that the digital filter design from Chapter 4 can be used to linearize the VSC behavior. As for the AD strategies, the admittance measurements verify that their effectiveness is limited to medium frequencies, while being detrimental at very high frequencies. The implications of the admittance properties are verified by testing the stability of specific grid-connected scenarios. As predicted by the analysis, only MS-PWM retains stability for all tested cases, while AD strategies cause instability. The final part of this chapter examines noise sensitivity of the tested passivizing control strategies. Models, derived based on the analysis from Chapter 5, are supported by experimental measurements. It is shown that AD strategies are not highly sensitive, however, MS-PWM brings a superior performance. Based on the design procedure from Chapter 5, the anti-aliasing digital filters are tested for MS-PWM and it is demonstrated that they enable strong noise suppression, while having a negligible impact on the admittance properties. The main original contributions found in this chapter are:

- Investigation of the impact of MS-PWM on passivizing the VSC admittance, also considering the MS-PWM nonlinearities.
- Comparison in passivizing effectiveness of MS-PWM and AD control strategies.
- Analysis and modeling of noise sensitivity of AD control strategies.

The material from this chapter is published in two journal manuscripts [70, 71].

Besides the papers listed above, additional publications on topics closely related to this thesis include a conference [72] and two journal [73, 74] manuscripts.

Chapter 2

Control system architecture



Figure 2.1: Illustration of a digitally controlled power electronic system.

A power electronic system (PES) is illustrated in Fig. 2.1. It comprises the switching network, based on the power semiconductor devices (switches), which interfaces different electrical systems via passive filters. In the given illustration, two electrical systems are present; however, a PES can interface an arbitrary number of systems, which is typically referred to as multi-port conversion. Depending on the switching network topology and the switch technology, PESs can enable energy flow in an arbitrary direction. The energy flow depends on the state of the switching network, which can be manipulated by using controllable power devices (e.g. transistors) [75].

A fundamental step in designing the power electronic control system is the choice of the actuation strategy. Namely, based on the required energy flow, the state of the switching network must be dynamically adjusted. This requires an appropriate transformation from the controller outputs to the switching signals that determine the switching network state. This is typically referred to as the modulation strategy of PESs. As noted in the introduction, there are several ways to control the switching network state. The one analyzed in this thesis is the pulsewidth modulation (PWM).

PESs are most-often controlled in closed-loop configurations, in order to compensate disturbances and nonlinearities, and to improve the dynamic performance [75]. Nowadays, high-performance control systems for power electronic converters are mostly realized within digital control platforms. This is due to the simplicity, flexibility, reproducibility, and adaptability of digital controllers, as well as due to many added functionalities that are unavailable when using analog systems [3,4]. The digital closed-loop control is what enables the use of PESs as true energy-processors, creating an unbreakable link between the power electronics and most of the existing and emerging technologies [1].

This chapter begins by introducing the structure of the digital PWM-based control system, which will be used throughout this thesis, with only minor variations in each chapter. Then, the pulsewidth modulation is explained in more detail and is classified based on different possible implementations. Some of the important PWM properties are briefly summarized by recalling the pioneering work and the main motivation for the multi-sampled PWM is elaborated accordingly.

This thesis is focused on analyzing fundamental properties of MS-PWM. As the main accent is placed on the modulation itself, it was found suitable to analyze it within systems with relatively simple structures, where the resulting effects are easily observed. The aspired outcome is to motivate the implementation of MS-PWM in arbitrary power electronic control systems, with much higher levels of complexity, while being fully aware of its potentialities and limitations.

As an example, a natural extension of this research is the analysis of joined properties of multiple phase-shifted multi-sampled pulsewidth modulators that may be employed in series or parallel stacked converters [51, 55, 76–78]. Related research is being conducted at the time of writing [74], while this thesis focuses on applications that require a single modulator.

2.1 Elements of digital PWM-based control systems



Figure 2.2: Block diagram of a single-stage closed-loop digital control system.

A digital, single-stage, PWM-based, closed-loop control system is shown in Fig. 2.2. The switching state of the converter is determined by the logic switching signal x(t). It is assumed that x is a square waveform with a fixed period, referred to as the switching (modulation) period T_{pwm} . The switching

frequency is labeled as $f_{pwm} = \frac{1}{T_{pwm}}$. Choice of the switching frequency depends on the hardware design, and is limited by the power semiconductor technology, thermal constraints due to switching losses, electromagnetic interference (EMI), and more [75].

The regulated output y(t) is fed-back to the control system. Its transition from the continuous time domain to the digital domain is performed by an analog-to-digital converter (ADC), with the sampling frequency $f_s = N f_{pwm}$, where N is the multisampling (oversampling¹) factor. The illustrated digital domain comprises a feedback filter $G_{fb}(z)$, a reference set-point r[k], and a controller $G_c(z)$, which processes the error signal e[k]. The operating frequency of the digital part is determined by the sampling frequency. The controller's output $m_s[k]$, in the form of an impulse train, is used as the input of the digital pulsewidth modulator (DPWM). The update period of m_s , $T_s = \frac{1}{f_s}$ is interchangeably referred to as the sampling and the control period². The DPWM block serves as an interface between the digital and the continuous time domains. Its inherent zero-order hold action transforms $m_s[k]$ into the piecewise constant modulating waveform m(t), which is compared with the DPWM's carrier w(t), resulting in the switching signal x(t).

2.2 Pulsewidth modulation



Figure 2.3: Illustration of the PWM process: (a) Comparison between the modulating waveform m and the carrier w, which results in the switching signal x; (b) Spectra of the modulating waveform and the switching signal. The bandwidth of interest for replication is labeled with f_{bw} .

Let us here recall some basic operating principles of pulsewidth modulators.

The goal of PWM is to re-create the low-frequency content of its input reference (modulating) waveform m, using the high-frequency switching square-waveform x. The period of x, T_{pwm} , is determined by the carrier w. The shape of w can be arbitrary and the most-widely used ones are analyzed in Section 2.2.3.

¹Oversampling is the traditional term, used in digital signal processing applications. The term multisampling is more specific to the analyzed pulsewidth modulation in power electronic applications.

 $^{^{2}}$ Throughout this thesis, most of the analyzed digital control systems are single-rate, meaning that all digital blocks operate with the rate determined by the sampling frequency. However, in a general case, the digital system can have an arbitrary number of operating frequencies. Multi-rate digital systems will be analyzed in a smaller extent, as a benchmark or in cases where they bring unique properties.

The high state (on state) of x is obtained when m > w, while m < w results in its low state (off state)³. The switching waveform x exhibits a variable pulsewidth (on state duration), depending on the difference between m and w. Within one switching period, the pulsewidth of x relative to T_{pwm} is referred to as its duty cycle D. The pulsewidth of x determines its low-frequency spectrum, such that it corresponds to the one of the modulating waveform.

Besides recreating the low-frequency spectrum of m, the PWM generates harmonic components around the multiples of f_{pwm} , which are referred to as the switching harmonics and their sidebands. Note that x is not necessarily periodic, depending on the ratio between the carrier and the modulating signal frequencies [15]. This fact is not crucial for the analysis found in this thesis; hence, for simplicity it is always assumed that the PWM process is periodic. Generation of the PWM output spectrum, for a fixed frequency f_{pwm} , is analyzed using the double Fourier series expansion and can be read in [15,79–81]. An illustration of the PWM process described above is shown in Fig. 2.3.

Providing a fixed switching frequency is one of the main advantages of PWM in power electronic applications, as it results in a deterministic harmonic content localized around multiples of f_{pwm} , which simplifies the hardware design process with respect to both thermal considerations and passive filtering. Additionally, PWM allows the effective use of linear controllers, which leads to a relatively simple control system design.

2.2.1 Digital PWM resolution

When implemented digitally, the carrier waveform is generated using an integer counter with a clock rate $f_{clk} = \frac{1}{T_{clk}}$. The total number of clock counts sets the switching period as $T_{pwm} = N_{clk}T_{clk}$. The time resolution of a DPWM, $\Delta t_{min} = T_{clk}$, is defined as the smallest feasible variation of the pulsewidth of x [4]. The duty cycle resolution is defined as $\Delta D_{min} = \frac{\Delta t_{min}}{T_{pwm}} = \frac{1}{N_{clk}}$. In case N_{clk} is a power of two, the DPWM resolution can be expressed by an equivalent number of bits as $n_{bit} = \log_2(N_{clk})$. The DPWM resolution determines how coarse of an actuator the converter is, as it translates to the smallest possible variation of the converter's output voltage. For a small-enough ΔD_{min} , the DPWM quantizer behaves like a linear block [82].

In closed-loop systems, if the DPWM resolution is not high enough with respect to the ADC resolution, limit-cycle oscillations (LCOs) are bound to appear [4,82–84]. In modern digital control platforms, f_{clk} is typically in the order of 100 MHz, which, to put in a perspective, results in a DPWM resolution of over 10 bits for applications with $f_{pwm} < 100$ kHz. For extremely high switching frequencies, which are not the focus of this thesis, increasing the clock rate to a sufficiently high value would significantly increase the power consumption of the control platform. Therefore, the high clock rate counters are often replaced by tapped delay-lines or some hybrid combination of both [14,82,85,86].

 $^{^{3}}$ It is assumed that, during one switching period, an appropriate logic is implemented to allow only one transition (switching action) from the high to the low state (turn off) and from the low to the high state (turn on).

2.2.2 Classification based on modulating waveform properties



Figure 2.4: Classification of the pulsewidth modulation based on the modulating waveform properties. The used acronyms are: naturally-sampled (NS), uniformly-sampled (US), single-sampled (SS), double-sampled (DS), multi-sampled (MS).

The pulsewidth modulators can be separated in two main groups based on the properties of the modulating waveform [27]. The PWM classification, used in this thesis, is illustrated in Fig. 2.4.

The first group, known as the naturally-sampled PWM (NS-PWM), relies on using the analog modulating waveform m, without any additional processing [15]. Its name comes from the fact that m is being sampled by the comparator itself, at the instants when it becomes equal to w. NS-PWM can only be realized within analog control systems. A significant merit of NS-PWM is that it, in a small-signal sense, it behaves as a pure gain, without introducing any delays [17]. In this thesis, the terms analog modulation and NS-PWM will be used interchangeably.

The second group is known as the uniformly-sampled PWM (US-PWM). In US-PWM, the modulating waveform m is obtained by processing the controller's output m_s with a zero-order hold (ZOH). US-PWM can be realized within analog control systems; however, it is inherent to digital system where the ZOH is part of the DPWM. An important difference with respect to NS-PWM is that, in US-PWM, there is always a certain amount of delay between the instant at which m_s is updated and the instant at which its held version m intersects with w. This introduces delay (both small-signal and large-signal), which is analyzed in Section 2.2.4. US-PWM can be divided in 3 distinct types, depending on the ratio between the ZOH sampling frequency and the switching frequency⁴.

The single-sampled PWM (SS-PWM) is obtained if the sampling frequency is equal to the switching frequency, i.e. $f_s = f_{pwm}$ and N = 1. The double-sampled PWM (DS-PWM) is obtained for $f_s = 2f_{pwm}$, i.e. N = 2. An illustration of a control system with SS-PWM and DS-PWM is shown in Fig. 2.5. The figure also illustrates sampling of the output y, which is assumed to have a triangular shape. SS-PWM and DS-PWM are singled-out, for an intuitive reason. Namely, within one switching period, the switching

⁴It is important to note that the term "sampled" refers to the modulating waveform; hence, in digital systems, the ZOH sampling rate is equal to the controller update rate, which is the same as f_s for single-rate systems such as the one in Fig. 2.2.

signal x can be affected by changing its rising and falling edges. SS-PWM can only modulate the two edges together, while DS-PWM can impact both of them independently. As it will be shown in the following section, whether one or two edges of x can be modulated depends also on the type of carrier used. For many effects analyzed in this thesis, conclusions do not qualitatively differ for SS-PWM and DS-PWM. In such cases, they will be jointly referred to as the (S/D)S-PWM. When needed to avoid confusion between the sampling of the modulating waveform and the sampling of the feedback signal, SS-PWM and DS-PWM will also be referred to as the single-update and double-update PWM [3].



Figure 2.5: Illustration of a digital control system operating with: (a) SS-PWM and (b) DS-PWM. The feedback signal y is assumed to feature a triangular ripple shape.

The multi-sampled PWM (MS-PWM) is obtained when the sampling frequency is higher than twice the switching frequency, i.e. $f_s = N f_{pwm}$, where N > 2 [18]. An illustration of a control system with MS-PWM is shown in Fig. 2.6. Updating the controller's output more than 2 times in a modulation period means that not every sample of m_s determines an edge of x; however, as shown in Section 2.2.4, increase of N reduces the modulation delay.



Figure 2.6: Illustration of a digital control system operating with MS-PWM. The feedback signal y is assumed to feature a triangular ripple shape.

Finally, let us mention the case of multi-rate digital control systems. A typical related scenario would be to implement filtering and control at different rates, by either adding an interpolator or a decimator after G_{fb} in Fig. 2.2. One example is the implementation of the oversampled filtering followed by the control action that is performed at a decreased rate. Note again that the control rate determines the type of US-PWM, i.e. SS-PWM, DS-PWM, or MS-PWM⁵. To differentiate these multi-rate (MR) cases from the single-rate ones, they will be labeled as MR-SS-PWM, MR-DS-PWM, and MR-MS-PWM. An example of MR-DS-PWM with a decimation between the filter and the controller is illustrated in Fig. 2.7.

There are two fundamental differences between the above-mentioned PWM types. First, a difference between NS-PWM and US-PWM is in their dynamic response capabilities. NS-PWM offers a faster large-signal response without introducing any small-signal delay [17]. This is the exact motivation for the implementation of MS-PWM. The second difference puts together SS-PWM and DS-PWM. Namely, given that y is periodic with T_{pwm} , up to two samples per period can be aligned with its average value, resulting in the cancellation of high-frequency harmonic content in y_s and, consequently, also m. Contrarily, for NS-PWM and MS-PWM, the switching harmonics are unavoidably introduced in the closed-loop system. This results in specific nonlinearities, some of which are discussed in Section 2.3.2 and some that are the focus of Chapter 4.

 $^{^{5}}$ Motivation for downsampling the filtered feedback can vary, e.g. to still use the standard SS-PWM or DS-PWM [63], or simply due to the fact that the control execution time limits the update rate. Oversampled filtering followed by a decimation to a lower-rate MS-PWM is analyzed in [72].



Figure 2.7: Illustration of a digital control system operating with MR-DS-PWM. The feedback signal y is assumed to feature a triangular ripple shape.

2.2.3 Classification based on carrier's waveform

Different shapes of w result in specific modulation schemes; the three typically used types are the trailingedge, leading-edge, and symmetric dual-edge (triangular) modulators. These are illustrated in Fig. 2.8.

Let us start by introducing a convention that will be used throughout this thesis. The time axis is normalized by T_{pwm} so that all modulation properties are examined within the range $\frac{t}{T_{pwm}} \in [0, 1)$. Also, the carriers are normalized and limited to $w \in [0, 1]$. This results in the modulating waveform m being unitless and also bounded to range $m \in [0, 1]$. The carrier parametrization that will be used throughout this thesis is:

$$w(t,\zeta) = \begin{cases} 1 - \frac{1}{\zeta} \frac{t}{T_{pwm}}, & \text{if } 0 < \frac{t}{T_{pwm}} < \zeta \\ \frac{1}{1-\zeta} \left(\frac{t}{T_{pwm}} - \zeta \right), & \text{if } \zeta < \frac{t}{T_{pwm}} < 1 \end{cases},$$
(2.1)

where the trailing-edge, leading-edge, and triangular modulators are obtained for ζ equal to 0, 1, and 0.5, respectively.

The trailing-edge modulation, shown in Fig. 2.8a, uses a positive-slope sawtooth carrier. It relies on fixing the rising edge of x while the falling edge is modulated by m. The leading-edge modulation, shown in Fig. 2.8b, uses a negative-slope sawtooth carrier. It modulates the rising edge of x while the falling edge is fixed. The trailing- and leading-edge modulators can be grouped together as the single-edge ones, as they only modulate one edge of x while the other one is fixed by the corresponding infinite slope of w. For the single-edge modulators, the switching signal x is always aligned with either the beginning or the end of the modulation period. The symmetric dual-edge modulation, shown in Fig. 2.8c, uses the triangular carrier with equal rise and fall times, allowing both edges of x to be modulated.

Let us here point to a connection between the carrier type and the sampling rate of US-PWM. Namely, for a single-edge modulator, only one value of m is needed to fully determine its output x. For


Figure 2.8: Typically used PWM carriers: (a) trailing-edge, (b) leading-edge, and (c) symmetric dualedge (triangular). Illustrations are given for analog m, corresponding to NS-PWM.

the triangular modulator, two, possibly different, values of m are used. Therefore, a naturally-imposed choice is to use SS-PWM for the single-edge modulators and DS-PWM for the dual-edge ones. It will be shown in Chapter 5 that modulators indeed do exhibit a nonlinear re-sampling behavior, with a rate that depends on the number of modulated edges of x.

2.2.4 Small-signal US-PWM modeling

2.2.4.1 Sampling instants

Let us start by defining the sampling instants of m_s . With respect to the carrier parametrization in (2.1), the array of N normalized sampling instants within one switching period is chosen as:

$$t_{sample}^{l} = T_{pwm} \left(l - 1 \right) \frac{1}{N} , \quad l \in [1, N].$$
 (2.2)

The reason behind this choice is evident for SS-PWM and DS-PWM. Namely, for the single-edge carriers with N = 1 and the triangular carriers with N = 2, the sampling instants in (2.2) correspond to the peaks or valleys of w. These values of w are suitable for the update of m_s as they do not introduce any limitation on the applicable value of D. Moreover, these instants guarantee that the modulated edge of x is determined by a horizontal crossing between m and w [27], which prevents nonlinearities that are discussed in Chapter 4. For MS-PWM, additional sampling instants are uniformly distributed across T_{pwm} . This choice is arbitrary, motivated by the pioneering work in [24].

Note that, due to delays of the digital control systems, there is always a time shift between the instant at which the feedback is sampled and the instant at which the controller's output is updated. These delays arise due to the ADC acquisition time and the execution time of the digital blocks in Fig. 2.2. Hence, to allow the precise update instants in (2.2), m_s is often updated one T_s after the feedback is sampled [3]. This will be referred to as the one step computation delay. Alternatively, feedback sampling is sometimes re-scheduled in advance to the instants in (2.2) [87]. Moreover, even without the delays, sampling the feedback at instants in (2.2) is not always the optimal choice, which is briefly discussed in Section 4.3.

2.2.4.2 Modulation delay

As mentioned in Section 2.2.2, in US-PWM, the modulation delay is present due to a time difference between the instants at which m_s is sampled and the ones at which the switching actions occur. An illustration of this time difference is shown in Fig. 2.9, for the triangular carrier and two values of N. The modulation delay impacts the large-signal [47] and the small-signal [17] response.

Consider a small perturbation being applied to m_s , such that $m_s = M_s + \hat{m}_s$. Small-signal modulator modeling relies on determining the resulting response of the duty cycle, i.e. $d = D + \hat{d}$ [4]. As explained in [4], the duty cycle D is inherently a discrete-time variable, even for analog PWM; however, its continuous-time representation d(t) is often used as a control input in the averaged modeling⁶ [88]. Hence, the modulator modeling corresponds to finding an expression between the discrete-time $m_s[k]$ and the continuous-time d(t) [27]. Small-signal modeling of SS-PWM and DS-PWM can be read in [17,89,90]. A generalization that covers MS-PWM is derived in [24].

The s-domain pulse-to-continuous transfer function $G_{dpwm}(s) = \frac{\hat{d}(s)}{\hat{m}_s(s)}$ is derived by finding the delays

 $^{^{6}}$ Regardless of the modeling strategy, the modulation delay is present and impacts the converter operation [4].



Figure 2.9: Illustration of the time difference between the sampling of m_s and the corresponding switching action, for DS-PWM (N = 2) and MS-PWM (N = 4).

between the sampling of m_s and the rising (τ_D^{on}) and falling edges (τ_D^{off}) of x [89]. These delays depend on the operating point and the oversampling factor [27]. An illustration of the US-PWM small-signal block diagram is shown in Fig. 2.10, and the corresponding modulator's transfer function is:

$$G_{dpwm}(s) = \frac{\hat{d}(s)}{\hat{m}_s(s)} = \zeta e^{-s\tau_D^{on}(D,N)} + (1-\zeta)e^{-s\tau_D^{off}(D,N)},$$
(2.3)

where ζ is the parameter from (2.1) that determines the carrier type, $s = \sigma + j\omega$ is the complex variable of the Laplace transform, j is the imaginary unit, and ω is the angular frequency.

Figure 2.10: Small-signal block diagram of US-PWM modulators.

In [24], it has been shown that for the trailing-edge, leading-edge, and triangular carriers, the frequency response of $G_{dpwm}(s)$ can always be represented in the following form:

$$G_{dpwm}(j\omega) = \frac{\hat{d}(j\omega)}{\hat{m}_s(j\omega)} = A(j\omega, D, N) \cdot e^{-j\omega\Delta t_{dpwm}(D,N)},$$
(2.4)

where A is a real function and Δt_{dpwm} represents the small-signal modulation delay. The gain A and Ph.D. Thesis — Ivan Petrić - 21 - the delay Δt_{dpwm} are expressed in Table 2.1, for all analyzed carriers.

	0 0 1	
Carrier type	$A(j\omega, D, N)$	$\Delta t_{dpwm}(D,N)$
Trailing-edge	1	$(D - q_N[D]) T_{pwm}$
Leading-edge	1	$(1 - D - q_N[1 - D]) T_{pwm}$
Triangular (N even)	$\cos\left(\omega T_{pwm}\left(\frac{D}{2} - q_N\left[\frac{D}{2}\right] - \frac{1}{2N}\right)\right)$	$\frac{T_{pwm}}{2N}$
Triangular (N odd)	$\cos\left(\omega T_{pwm}\left(\frac{D}{2} - q_N\left[\frac{D}{2} + \frac{1}{2N}\right]\right)\right)$	$\frac{T_{pwm}}{2N}$

Table 2.1: Small-signal gain and delay of US-PWM modulators.

The time-quantization parameter $q_n\left[\frac{t}{T_{pwm}}\right] = \frac{floor\left(N\frac{t}{T_{pwm}}\right)}{N}$ is defined as in [27] and corresponds to allocating the nearest preceding sampling instant.

First, it should be noted that, for all carriers, the gain A is either identically equal to, or very close to 1. Additionally, regardless of the carrier type, the introduced phase lag is decreased with the increase of N, which is expected due to the delay reduction between the sampling of m_s and the resulting switching actions. Clearly, for high values of N, MS-PWM approaches the NS-PWM and the introduced phase lag becomes negligible. The maximal achievable N depends on the available processing power relative to the switching frequency. Some application scenarios are more likely to allow high values of N (e.g. high power converters) than others (e.g. high switching frequency converters). Another important fact is that, for the trailing-edge and the leading-edge modulators, the time delay is a discontinuous function of the steady state duty cycle D. For the triangular modulators, the time delay is equal to $\frac{T_s}{2}$, which points to one of its advantages in terms of a superior linearity in MS-PWM applications [27]. Another advantage of the triangular modulator, as analyzed in [27], is the drastically lower nonlinearity induced by the switching ripple in m; the relevant effects are analyzed in Chapter 4.

Relying on the conclusions from [27], due to its superiority in MS-PWM applications, the triangular modulator is used throughout this thesis. Given that A is very close to unity, the triangular US-PWM modulator is often modelled as a transport delay in the Laplace s-domain:

$$G_{dpwm}(s) = \frac{\hat{d}(s)}{\hat{m}_s(s)} \approx e^{-s\frac{T_{pwm}}{2N}} = e^{-s\frac{T_s}{2}}.$$
(2.5)

The delay model from (2.5) does not capture the operating point dependency from (2.3). However, it brings a good prediction in a wide frequency range, making it suitable for most applications. A significant mismatch arises at frequencies close to the Nyquist frequency (NF) [73], which is relevant for the applications analyzed in Chapter 6.

Let us here state two important facts. First off, even though the US-PWM incorporates a ZOH, its small-signal model *does not* correspond to the one of the ZOH, which is often misinterpreted in the literature. Secondly, an important assumption used to derive the MS-PWM modulator's small signal model is that the ripple in m is small [24]. Analysis of MS-PWM operation with a non-negligible switching ripple in m is analyzed in Chapter 4.

Finally, let us mention another promising PWM carrier, which is not analyzed in this thesis. This is the asymmetric dual-edge carrier, which may bring a faster large-signal response as well as a phase-leading small-signal behavior. Its NS-PWM implementation is first analyzed in [91], while its fully-digital US-PWM implementation is recently published in [92]. The digital implementation is still limited to custom hardware, e.g. Field-Programmable Gate Array (FPGA).

2.3 Bandwidth limitations of PWM-based control systems

Bandwidth capabilities of power electronic systems are increased with higher switching frequencies. This is due to faster reaction times as well as smaller required passive filters at the output. However, the switching frequency is constrained by hardware specifications and is not a control design parameter.

Therefore, it is often a control task to achieve the maximal bandwidth for a given switching frequency. This brings many benefits such as a faster torque response and operating speed of electric drives [93,94], harmonic distortion compensation in grid-tied converters, uninterruptible power supplies (UPS), and active filters [3,25,95,96], as well as minimization of passive filters [97,98] that leads to cost reduction and higher power densities. In this section, the main bandwidth limitations in pulsewidth modulated power electronic systems are explained.

2.3.1 Small-signal limitations



Figure 2.11: Small-signal block diagram of the system from Fig. 2.2.

Let us start by analyzing the small-signal representation of the closed-loop control system from Fig. 2.2. The corresponding block-diagram is shown in Fig. 2.11, where all signals represent the small-signal perturbations while the symbol "^" is omitted. The DPWM interfaces the digital domain with the analog domain, and its small-signal transfer function is shown in the previous section. As an example, let us assume that the plant G_p is obtained using the state-space averaging technique [99]; hence, it relates the applied continuous-time duty cycle d with the converter's fed-back output y (e.g., a current through some inductor or a voltage across some capacitor). Its order depends on the output filter of the converter.

Using the standard tools of the linear control theory, the closed-loop system from Fig. 2.11 can be

analyzed in terms of its reference tracking performance, disturbance rejection, and stability margins. An important thing to notice is that the analyzed system features both the continuous and the discrete time domain. Depending on the required analysis, the transformation can be performed either from the continuous to the discrete time domain $(\mathcal{Z} \{ \mathcal{L}^{-1} \{ G_{dpwm}(s) G_p(s) \} \})^7$, or from the discrete to the continuous time domain $(\mathcal{L} \{ \mathcal{Z}^{-1} \{ G_{fb}(z) G_c(z) \} \})$. Symbols \mathcal{L} and \mathcal{Z} label the Laplace and the Z transforms, respectively.

For simplicity, let us here neglect the sampler and assume that the frequency response of all individual blocks in Fig. 2.11 is known. The stability margins can be assessed using the open-loop transfer function $W_{ol}(j\omega)$:

$$W_{ol}(j\omega) = G_{fb}(j\omega)G_c(j\omega)G_{dpwm}(j\omega)G_p(j\omega).$$
(2.6)

Assuming a fixed crossover frequency ω_c and using $G_{dpwm}(j\omega)$ from (2.4), the phase margin can be found from (2.6) as:

$$PM = \pi + \arg\{W_{ol}(j\omega_c)\} = \pi + \arg\{G_p(j\omega_c)G_{fb}(j\omega_c)G_c(j\omega_c)\} - \omega_c\Delta t_{dpwm}.$$
(2.7)

From (2.7), it is clear that the small-signal stability is impacted by the modulation delay. Moreover, in practice, additional delays are always present due to a limited bandwidth of the sensing circuitry and the times related to the ADC acquisition, signal propagation, and the control execution. In general, for control systems with (S/D)S-PWM, the introduced small-signal phase lag is the limiting factor and the achievable bandwidths are often determined by the modulation delay [3]. Note that the phase lag can be partially compensated by adding phase-leading actions to the controller [3, 100], however, with a limited impact at high frequencies and at the expense of an increased noise sensitivity.

2.3.2 Large-signal limitations

For NS-PWM and MS-PWM with high enough N, the modulation delay is negligible. However, even without any other phase-lagging blocks in Fig. 2.11, the bandwidth cannot be set to an arbitrarily high value. Namely, all PWM-based systems feature large-signal limitations that arise for two reasons [33, 101–106]. The first one is the overlap between the bandwidth of the modulating signal m and the switching harmonic sidebands generated by the PWM [104, 105]. The second one is related to maintaining the modulation smooth by limiting the slew rate of m [33, 101–103, 106].

For the first type of large-signal limitations, let us consider the following application scenario, which may appear in both NS-PWM and US-PWM systems. The case of interest is when the modulating signal m is multi-tone, i.e. it contains a dominant fundamental-frequency (or dc) component as well

 $^{^{7}}$ As shown in [4], the impulse-invariant discretization of the averaged continuous-time dynamics is equivalent to using the direct discrete-time modeling for time-invariant topologies (e.g. buck converters).

as other harmonics. These harmonics are introduced in m either by the tracking reference or a feedback/feedforward reaction to a disturbance. A typical example may be found in an active filtering application [95]. The desired PWM behavior is that of a linear wideband amplifier, i.e. such that the bandwidth of m is faithfully replicated in its output. As noted previously, and illustrated in Fig. 2.3, by comparing m with w, PWM generates its output with the spectrum that contains components at frequencies of m, w, as well as their linear combinations (with integer coefficients). It is clear that, for mwith components at high-enough frequencies, the PWM sidebands may cause an overlap. This creates an interference that resembles aliasing; hence, a criterion similar to the sampling theorem can be used to limit the bandwidth of m such that an acceptable amount of error is introduced by the modulation process.

Finding the exact bandwidth limitation requires computing the output spectrum of the switching signal x, using Bessel's functions for a given multi-tone m, and setting a limit to the introduced "aliasing" error [105]. The resulting expressions are very complex; however, a simple rule-of-thumb limitation can be found, in a manner similar to the Carson's rule [107] used in the telecommunication theory. Based on the analysis from [104], it can be shown that, for m that contains a dominant component at frequency f_1 and series of harmonics with the maximal frequency f_m^{max} and relatively small magnitudes, the lowest-frequency sideband harmonic with a non-negligible magnitude appears at the frequency $f_x^{min} = f_{pwm} - f_1 - f_m^{max}$. The bandwidth limitation is obtained by setting $f_x^{min} > f_m^{max}$, which leads to $2f_m^{max} + f_1 < f_{pwm}$. This guarantees some gap is placed between the bandwidth of m and the strong switching harmonic sidebands. A more conservative but memorable condition is obtained when f_m^{max} is set equal to f_1 :

$$f_m^{max} < \frac{1}{3} f_{pwm}.$$
(2.8)

This limitation is often met in high-power applications, where the switching frequencies are very low. There, the carrier-based PWM is often replaced by pre-calculated switching patterns [108].

The second large-signal limitation, relevant for NS-PWM and MS-PWM, is related to maintaining the modulation smooth in the presence of switching harmonics in the modulating waveform [33,101,102,106]. To avoid considering the impact of the sampler, let us refer to the NS-PWM, while the same conclusions are brought for MS-PWM with high values of N [33]. Consider the block diagram from Fig. 2.2, where the control system is realized in the analog domain $(f_s \to \infty)$. The switching harmonics, that appear due to PWM, propagate to the converter's output, after being filtered by the plant. These harmonics are then fed-back, processed by the controller and, finally, they appear in the modulating waveform m. As noted in [33,101,102,106], a condition for the smooth modulation is satisfied if the slope of m never exceeds the slope of w at their intersections. Otherwise, the modulator exhibits a chaotic, unstable operation. The magnitude of the switching ripple in m and, consequently, its slope, depend on the filtering characteristic of the entire closed-loop system and are amplified with higher controller gains. Therefore, the maximally-allowed slew rate of m limits the controller and, consequently, the achievable

bandwidth. It is important to note that the slope of m also depends on the operating point. The procedure for calculating the maximal controller gain is reported in [33].

To conclude, the bandwidth of PWM-based control systems depends on the small-signal and the large-signal effects. Which one of them will be dominant and cause a limitation depends on the sampling strategy and the physical parameters of the converter. For (S/D)S-PWM, the bandwidth is mainly limited by small-signal stability margins, which are often compromised by delays. Additionally, the bandwidth of m is limited by the allowed interference with the switching sidebands. For NS-PWM and MS-PWM, the bandwidth is generally limited by the large-signal effects, either the interference of m with the switching sidebands, or due to the slope of m with respect to w. For MS-PWM, additional nonlinearities arise due to the switching ripple in m [24, 66]; these are the main focus of the analysis in Chapter 4.

2.3.3 Filtering the switching ripple in MS-PWM

Based on the analysis from the previous section, it is clear that the presence of the switching ripple in the modulating waveform is detrimental for the operation. The switching ripple that appears in the converter's output is determined by the topology, the switching frequency, and the passive filters, which are all related to the hardware design process. Within the control domain, the switching ripple can be suppressed using appropriate filters, e.g. $G_{fb}(z)$ in Fig. 2.2.

Ripple filtering directly impacts the slope of m, which has a positive effect on enabling a smooth modulation. Moreover, in the case of MS-PWM, as discussed in Chapter 4, the switching ripple in m brings another set of nonlinearities [24, 66]. For this reason, in many MS-PWM applications, the switching ripple is completely filtered out [19, 25, 26, 34, 109]. The design goal is to remove the ripple content from m, while imposing as low as possible impact on the dynamic response. While an arbitrary filter may be used, let us here point to some structures whose application is reported in the literature.

First, let us start with the finite impulse response (FIR) moving average filter (MAF). It relies on using the N latest feedback samples to obtain the averaged output over one T_{pwm} :

$$G_{MAF}(z) = \frac{1}{N} \sum_{n=0}^{N-1} z^{-n}.$$
(2.9)

The MAF is widely used due to its simplicity, capability to effectively remove the switching ripple, as well as to provide a strong noise attenuation [63, 68], which is analyzed in Chapter 5. However, its big demerit is the significant added phase lag, which compromises the small-signal stability. Namely, the phase response of the MAF is close to a transport delay equal to one half of the averaging period, i.e. $e^{-s\frac{T_{pwm}}{2}}$. This value is equal to the modulation delay for SS-PWM; hence, using the MAF completely cancels the dynamic improvements obtained with MS-PWM. It should be noted that MR-SS-PWM and MR-DS-PWM are often implemented with the oversampled MAF [63, 100, 110, 111]. This is done to suppress the feedback acquisition errors, caused by an imperfect sampling synchronization, parasitic oscillations, and various sources of noise [63]. It is clear that the difference between the average of N samples and the actual dc component of the output variable depends on the value of N.

To reduce its impact on the dynamic response, the MAF can be simply modified by using less samples for averaging. In [25], the following filter, based on two samples, is used:

$$G_{MAF,r}(z) = \frac{1}{2} \left(1 + z^{-\frac{N}{2}} \right).$$
(2.10)

The filter in (2.10) introduces a smaller phase lag, however, the ripple is not completely removed (the even switching harmonics remain) and the strong capability for noise suppression is lost.

Additionally, the averaging delay can be partially compensated by adding a phase-leading structure in cascade with the filter, which is analyzed in [34, 63, 100, 109]. This involves either directly applying a derivative action [63, 109] or a linear predictor [34, 100]. Both methods are structurally similar and bring an increased noise sensitivity.

A repetitive filter for ripple removal is proposed in [26]:

$$G_{RF}(z) = \frac{(1+R)\left(1 - \left(z^{-N} - \frac{1}{N}\sum_{n=1}^{N}z^{-n}\right)\right)}{1 - \left(z^{-N} - \frac{1}{N}\sum_{n=1}^{N}z^{-n}\right) + R},$$
(2.11)

where the gain R determines the settling time. This structure brings a negligible phase lag below the switching frequency, which is favorable regarding the small-signal dynamics. However, as for the most repetitive filters, its settling time is relatively long. In [26], it is explained that the gain R must be set relatively low, i.e. the ripple reconstruction must be slow so that the filter would not compromise stability. There, the settling time is set to (10 - 20) switching periods. Therefore, the repetitive filter is only suitable when the steady-state is kept for a long time. If it is frequently perturbed, the effectiveness of the filter is strongly limited. Moreover, it does not bring any low-pass filtering action; hence, it does not suppress the feedback noise.

Let us also mention a technique that removes the switching ripple based on prediction [19]. It relies on reconstructing the ripple component and subtracting it from the feedback. In this way, a very simple ripple removal is obtained, with practically zero delay; however, there is a high sensitivity to parameter mismatch.

In case the described filters are not suitable, a simple infinite impulse response (IIR) low-pass filter may be used to smooth the ripple in m with a low impact on the dynamic response. However, care must be taken not to trigger some of the MS-PWM specific nonlinearities, analyzed in Chapter 4.

In conclusion, there is a trade-off between suppressing the large-signal effects, by filtering the switching ripple, and maintaining the small-signal dynamic improvements obtained with MS-PWM. Therefore, there is a large degree of flexibility and space for optimization in an MS-PWM control system design.

Chapter 3

Experimental set-ups

This thesis is focused on analyzing some general properties of the multi-sampled pulsewidth modulation, regardless of the specific converter topology. For this reason, it was found that the two-level buck-type converters are suitable for an in-depth analysis of the multi-sampled modulation, without introducing an extra layer of complexity related to the topology itself.

3.1 Power converters

For all results in this thesis, two hardware systems were used, both configured as two-level buck-type converters.

The first one, referred to as the set-up A, was used in the early research stage, for part of results presented in Chapter 4, which are published in [65]. It was formed using one leg of a three-phase inverter, connected to LC output filter and resistive load. The schematic of the set-up A corresponds to Fig. 3.1a and the hardware parameters are shown in Table 3.1.

Description	label	value	unit
Nominal input voltage	V_{in}	200	V
Switching frequency	f_{pwm}	20	kHz
Filter inductance	Ĺ	0.6	mH
Output capacitance	C	30	μF
Output load resistance	R	30	Ω

Table 3.1: Hardware parameters of the set-up A.

The set-up B consists of a full-bridge buck converter, compatible with applications as a bidirectional dc-dc or ac-dc converter. The used transistor devices are FGA60N65SMD IGBTs. The input voltage supply was formed using the regenerative power system Keysight RP7962A. The switching frequency was chosen based on the used IGBTs and the inductor cores and set to 20 kHz.

For dc-dc tests in Chapters 4 and 5, the set-up B was configured as a half-bridge buck converter with LC output filter and resistive load, shown in Fig. 3.1a. This is referred to as the set-up B.1. For ac-dc



Figure 3.1: Converter topologies used in this thesis: (a) half-bridge buck converter with an *RLC* load, used in the set-ups A and B.1; (b) full-bridge buck converter with an *RLC* load, used in the set-up B.2; (c) full-bridge buck converter connected to an ac source, used in the set-up B.3.

tests in Chapter 4, the set-up B was configured as a full-bridge inverter/rectifier operating with bipolar modulation. The used coupled inductor was split between the two legs of the converter. The full-bridge configuration is again used with LC filter and resistive load, as shown in Fig. 3.1b. This is referred to as the set-up B.2. The parameters of the set-ups B.1 and B.2 are shown in Table 3.2.

For the grid-connected application in Chapter 6, the full-bridge configuration was connected via an

Description	label	value	unit
Nominal power	S_n	3	kVA
Nominal input voltage	V_{in}	400	V
Switching frequency	f_{pwm}	20	kHz
Filter inductance	L	$\{1.2, 1.5\}$	mH
Output capacitance	C	20	μF
Output load resistance	R	47	Ω
Dead time	t_{dt}	500	ns

Table 3.2: Hardware parameters of the set-ups B.1 and B.2.

inductive filter to a sinusoidal voltage source, formed using the Chroma 6460 programmable ac supply. This is referred to as the set-up B.3, which is illustrated in Fig. 3.1c and whose parameters are shown in Table 3.3.

Table 3.3: Hardware parameters of the set-up B.3

Description	label	value	unit
Nominal power	S_n	3	kVA
Nominal input voltage	V_{in}	400	V
Nominal PCC voltage	$v_{pcc,RMS}$	230	V
Nominal admittance	Y_n	56.7	mS
Fundamental frequency	f_1	50	Hz
Switching frequency	f_{pwm}	20	kHz
Filter inductance	L	$\{1.5, 2.5\}$	mH
Dead time	t_{dt}	500	ns

A photo of the experimental set-up B, configured for the grid-connected application (set-up B.3) is shown in Fig. 3.2.

3.2 Control platforms

For control system implementation, two digital platforms were used.

In the set-up A, the converter was controlled using a digital signal processor (DSP) from Texas Instruments C2000 series, TMS320F28379D. This will be referred to as the DSP control platform.

For all configurations of the set-up B, the control system was implemented on an NI sbRIO-9606, which is based on a Xilinix Zynq 7020 all programmable system on chip (AP-SoC). This system will be referred to as the FPGA control platform. Controlled variables are sensed using a custom interfacing board, which is re-configurable to enable current measurements (based on a shunt-resistor) and voltage measurements (based on a voltage divider). The interfacing board is built based on the original design from [112]. The board uses conditioning circuits with a high common mode rejection capability, 12 bit



Figure 3.2: The experimental set-up B configured for the grid-connected application: (1) Keysight RP7962A, used as the input power supply; (2) converter and the control board; (3) point of common coupling with the inductor and additional filters used to form various grid impedances; (4) Chroma 6460, used as the programmable ac supply.

ADC module AD9226 by Analog Devices, and digital isolators that enable the FPGA platform to read the digital signal values at very high rates. The maximally achieved sampling rate is 40 MHz. The FPGA is programmed in LabVIEW and features a DPWM with the counter clock rate equal to 160 MHz, communication and protection systems, synchronization and control loops.

Chapter 4

Nonlinearities caused by the switching ripple

4.1 Introduction

The positive impact of multi-sampled PWM on the dynamic performance is evident from the smallsignal model in (2.5). However, the underlying modeling approach from [24] relies on the assumption that the switching ripple in m is small enough. For MS-PWM, in general, this cannot be ensured without specific ripple removal filters, some of which are described in Section 2.3.3. Due to the several mentioned drawbacks of these filters, there is a motivation to further analyze the MS-PWM operation with a non-negligible switching ripple content in m.

The focus of this chapter is the analysis of the modulator nonlinearities that are caused by the switching ripple. Three sets of analyzed nonlinearities are those due to *aliasing*, *ripple modulation*, and *modulating waveform discontinuities*. The aliasing problems are present only in digital control systems. The effects caused by the ripple modulation [31–33, 88, 113, 114] are found in both NS-PWM and MS-PWM. The modulating waveform discontinuities are uniquely related to MS-PWM; moreover, it will be shown that their impact is emphasized for lower oversampling factors. The first two types of nonlinearities are only briefly discussed here, being covered by the previously published work. The third one is analyzed in detail, and the corresponding results are one of the main contributions of this thesis.

Regarding the effects caused by the modulating waveform discontinuities, the focus of this chapter is first placed on explaining each specific nonlinearity that may arise. Their impact is manifested through the reduced-gain [24], zero-gain [24], and infinite-gain [65] zones of the modulator. The discovery and analysis of the infinite-gain zones is one of the original contributions of this thesis. It is shown that the infinite-gain zones may result in limit cycle oscillations, bringing a highly detrimental impact on a converter's operation. This effect is statistically modelled and an algorithm is proposed to cancel out the induced oscillations. Additionally, this chapter brings a generalized methodology suitable to predict which nonlinearity will appear, depending on the control system parameters. The corresponding findings are published in [65, 66]. The main results of this chapter can be summarized as:

- Analysis and classification of the MS-PWM nonlinearities caused by the modulating waveform discontinuities. Specifically, the infinite-gain zones are introduced for the first time.
- A methodology for predicting the appearance of MS-PWM nonlinearites and quantifying the resulting effects.
- A framework for designing the control system so as to suppress or even completely cancel the discontinuity-related MS-PWM nonlinearities, without needing to remove the switching ripple.

4.2 Converters under test

The phenomena described in this chapter are directly related to the shape of the switching ripple in the modulating waveform, which depends on the converter topology and the control system structure. In this chapter, the experimental results are given for two-level current-controlled buck-type converters. Specifically, the hardware set-ups A, B.1, and B.2, described in Section 3.1, are used. The corresponding schematics are shown in Fig. 3.1.

Numerical results reported in this chapter correspond to these systems; however, the presented methodology can be extended to other topologies and controller structures, operating with multi-sampled pulsewidth modulation, without any particular challenges.

4.2.1 Controller design

For the analysis in this chapter, single-stage current control systems are considered. Current control was found suitable to address the modulator's properties in the presence of strong switching harmonics. The cut-off frequency of the output LC filter is chosen to be sufficiently low, such that the output voltage dynamics are negligible. An illustration of the implemented digital control system is shown in Fig. 4.1, assuming the half-bridge buck configuration.

Let us start the analysis in s-domain, which is often a viable approach for the controller design [3]. A simplified s-domain representation of the system from Fig. 4.1, where the feedback filter and the modulation delay are neglected, is shown in Fig. 4.2. As the buck-type converters are large-scale linear, in all expressions the small-signal label "^" will be omitted.

For the analyzed two-level converters, there is only one active switching signal x. The switched node voltage of the converter v_x is determined by the switching signal and the input voltage, i.e. for the halfbridge configuration it is equal to $v_x(t) = x(t)V_{in}$, while for the full-bridge with bipolar modulation it is equal to $v_x(t) = (2x(t) - 1)V_{in}$. It is assumed that the input voltage dynamics are negligible. Using the averaged modeling, the one-cycle average of v_x is related to the input voltage using the continuous duty cycle d(t) as the control variable [75,99]. Therefore, neglecting the modulation block, the controller can



Figure 4.1: A digital current control system of a two-level half-bridge buck converter. The inductor current i_L is sampled by the ADC, processed by a digital filter G_{fb} , and subtracted from the reference $i_{L,r}$. The resulting error signal e is processed by the controller G_c to obtain m_s . The modulating waveform m is obtained after the ZOH processing of m_s within the DPWM.



Figure 4.2: S-domain small-signal block diagram of current-controlled buck-type converters, without including the feedback filter and the modulation delay.

be seen to output the required duty cycle of x, which determines the applied switched node voltage as $v_x = d \cdot K + V_x^0$, where K is the gain and V_x^0 is the dc offset¹. For the half-bridge configurations, $K = V_{in}$ and $V_x^0 = 0$. For the full-bridge configurations with bipolar modulation, $K = 2V_{in}$ and $V_x^0 = -V_{in}$. The switched node voltage, together with the disturbance v_o , determines the inductor current response as:

$$G_p(s) = \frac{i_L(s)}{v_x(s) - v_o(s)} = \frac{1}{sL}.$$
(4.1)

Let us now present the typically used controller structures, for the inner current control loops, corresponding to Fig. 4.2. Given that the analyzed plant model in (4.1) is an integrator, a zero steady-state error can be achieved using just proportional control, i.e.:

$$G_c^P = k_p, \tag{4.2}$$

where k_p is the proportional gain. For the set crossover frequency f_c , the proportional gain is calculated as $k_p = 2\pi f_c \frac{L}{K}$. As explained in Section 2.3, the maximal crossover frequency is limited by the control delays (which are left out of the small-signal block diagram in Fig. 4.2) and the switching frequency [3,33].

¹The offset does not impact the small-signal analysis, which is why it is left out of the block diagram.

For dc-dc converters, to improve robustness against the model uncertainties and other disturbances, an integrator is often added to form the proportional-integral (PI) controller [3]:

$$G_{c}^{PI} = k_{p} + k_{i} \frac{1}{s} = \frac{k_{p}s + k_{i}}{s} = k_{p} \frac{s + \omega_{i}}{s},$$
(4.3)

where k_i is the integral gain and $\omega_i = \frac{k_i}{k_p}$ is the angular frequency of the zero added by the integral action. An often used PI controller design strategy is to calculate k_p like for the P controller, while the zero of the integral gain is placed such that its impact at the crossover frequency is negligible [3], e.g. $\omega_i = \frac{k_i}{k_p} < 0.1 \cdot 2\pi f_c$.

In case of converters with a sinusoidal ac operating point, the integrator can be replaced with a resonator [115,116]:

$$G_c^{PR} = k_p + k_r \frac{s}{s^2 + \omega_1^2},$$
(4.4)

where k_r is the resonant gain and ω_1 is the angular frequency of the sinusoidal reference. As the resonant action corresponds to the integral action observed in a rotating reference frame [3], the same design approach as for k_i of the PI controller is often used, i.e. $\frac{k_r}{k_n} < 0.1 \cdot 2\pi f_c$.

The analysis found in this chapter can be directly extended to any linear controller; however, these structures (P, PI, PR) and their combinations are most often used for single-stage current-controlled converters. An important property of these controllers is that, at high frequencies ($\omega \rightarrow \infty$), their frequency response converges to the one determined by the proportional gain. For this reason, they will be referred to as the high-frequency proportional-dominant controllers.

4.2.2 Impact of the control system on the modulating waveform shape

From Fig. 4.1, it can be seen that the current ripple propagates to m_s after being processed by the sensing circuits, the ADC, the feedback filter, and the controller. It is clear that all of these blocks, together with the ZOH of the DPWM, have an impact on the shape of ripple in the modulating waveform.

Let us start by analyzing the impact of the controller itself, using the standard structures introduced in the previous section. Other blocks, such as the feedback filter, will be used to modify the shape of mthroughout this chapter. Additionally, the impact of the controller will be first examined assuming an analog control system, shown in Fig. 4.3, such that the continuous-time m is observed. The sampledand-held version of m, present in MS-PWM, is related to this continuous-time waveform [24], making it useful for bringing some general conclusions before going into the sampled nature of the system.

Let us separate the inductor current as $i_L(t) = \langle i_L \rangle + i_L^{ac}(t)$, where $\langle i_L \rangle$ is its average value over one T_{pwm} and i_L^{ac} is its ripple component.

In order to provide results that have a general value, the following reasoning is made. For the analyzed two-level buck converters with constant output voltage, the inductor current ripple is triangular, with slopes that depend on V_{in} , L, f_{pwm} , and the steady-state duty cycle D [75]. Its peak-to-peak value Δi_L^{p-p} can be found as:



Figure 4.3: Block diagram of the analog current control system used to examine the impact of the controller on the modulating waveform ripple.

$$\Delta i_L^{p-p} = f(D) \frac{V_{in}}{L f_{pwm}},\tag{4.5}$$

where f(D) is determined by the topology configuration. For the half-bridge buck-type converters f(D) = D(1-D), while for the full-bridge buck-type converters with bipolar modulation f(D) = 2D(1-D) [75].

In the same way, the modulating waveform is separated in two parts as $m(t) = \langle m \rangle + m^{ac}(t)$. Since the switching frequency is always above the feasible control bandwidths, the ripple component i_L^{ac} propagates to the modulating waveform as if the system is in the open-loop configuration. Starting with the proportional controller in (4.2), it is clear that:

$$m^{ac}(t) = -k_p i_L^{ac}(t). (4.6)$$

For P controllers, m^{ac} completely retains the shape of i_L^{ac} . In case of PI or PR controllers with gains designed as in the previous section, the impact of the integral or the resonant action is bandlimited well-below the crossover frequency; hence, also below the switching frequency. Consequently, for these high-frequency proportional-dominant controllers, the modulating waveform ripple is well approximated by (4.6). Using (4.5) and the relation between the proportional gain and the crossover frequency of the system shown in Fig. 4.2, it is possible to correlate the steady-state peak-to-peak magnitude of the modulating waveform Δm_{p-p} with the designed relative crossover frequency $f_{c,r} = \frac{f_c}{f_{pwm}}$:

$$\Delta m_{p-p} = k_p \Delta i_L^{p-p} = 2\pi f_c \frac{L}{K} \Delta i_L^{p-p} = 2\pi f_c \frac{L}{K} f(D) \frac{V_{in}}{L f_{pwm}} = 2\pi f_{c,r} D(1-D).$$
(4.7)

From (4.7) it is clear that the modulating waveform ripple is simply proportional to the designed value of $f_{c,r}$, independently from the specific values of V_{in} and L. Moreover, it is equal for the half-bridge and the full-bridge buck converters. The expression (4.7) is a very close approximation for all mentioned high-frequency proportional-dominant controllers; hence, it can be concluded that these controller structures will bring very similar properties related to nonlinearities caused by the modulating waveform ripple. For other linear controllers, such as those involving a derivative action, the modulating waveform ripple is strongly affected by the controller; hence, their analysis must be conducted independently.

Based on the considerations above, it is concluded that the investigation of the ripple-induced nonlinearities can be performed in a scalable way, by varying only the relative crossover frequency of control loops with the high-frequency proportional-dominant controllers. This is found useful for bringing generalized conclusions, regardless of the exact values of f_{pwm} , V_{in} , L, k_r , and k_i . Note that, for LC output filters where the resonant frequency is near the crossover frequency, the plant cannot be modeled as a simple integrator from (4.1). Hence, the actual crossover frequency differs from $\frac{k_p K}{2\pi L}$. In those cases, the subsequent analysis is dependent on the designed value of k_p and not the obtained value of f_c .

4.3 Aliasing in digitally controlled power converters



Figure 4.4: Illustration of the sampling instants from (2.2), for the inductor current control loop with negligible output voltage dynamics for: (a) single-sampled (SS-PWM), (b) double-sampled (DS-PWM), and (c) multi-sampled (MS-PWM) with N = 4.

In digitally controlled power converters, aliasing-related issues may arise as the switching ripple is sampled [26, 34]. The switching harmonics cannot be fully removed using analog anti-aliasing filters, unless completely deteriorating the closed-loop dynamics. Therefore, without high oversampling, the bandwidth of the feedback signal is always above the Nyquist frequency and the aliasing is bound to occur. This is particularly evident in the inner inductor current control loops, which experience the strongest switching harmonic content [3].

For SS-PWM and DS-PWM, the feedback sampling instants can be chosen to align with the onecycle average of i_L , which results in the direct removal of the switching ripple component [3]. For the analyzed two-level inductor current control loops, neglecting any parisitic resistances and the output voltage dynamics, these instants coincide with the mid-points of the switching signal x(t). This kind of sampling synchronization is referred to as the *center-pulse sampling*.

For SS-PWM and the triangular carrier, neglecting the presence of any delays, the mid-points of x coincide with the peaks or valleys of the carrier, depending on when the modulating waveform is updated. An example is shown in Fig. 4.4a, where the modulating waveform is updated once per T_{pwm} at each peak of w(t), with the corresponding center-pulse instant being found at the valley of w(t). However, the presence of dead times and actuation delays always causes x to be, at least slightly, shifted. Moreover, with the parasitic resistances and a non-constant output voltage, the average value of i_L no longer coincides with the mid-points of x. In practice, for SS-PWM, these effects result in a low-frequency acquisition error that can be suppressed by re-scheduling the sampling instants accordingly [117].

For DS-PWM and the triangular carrier, it is often assumed that synchronizing the sampling instants with the peaks and valleys of the carrier results in the center-pulse sampling. However, even without any dead times or delays, the switching signal may be asymmetric with respect to the carrier, resulting in the switching ripple being sampled. This is illustrated in Fig. 4.4b. Therefore, for DS-PWM, besides the low-frequency aliasing, the switching harmonics may also be introduced in the feedback signal unless the sampling instants are dynamically adjusted. This effect brings one of the motivations for highly oversampling the feedback and then averaging it over one switching period, i.e. MR-DS-PWM [63].

For MS-PWM, N > 2 samples are taken within a switching period; hence, the ripple is unavoidably sampled. In case the average of N consecutive samples differs from $\langle i_L \rangle$, a low-frequency aliasing error is introduced. An example of MS-PWM with N = 4 and the sampling instants from (2.2) is shown in Fig. 4.4c. Intuitively, the low-frequency aliasing is less emphasized for higher oversampling factors, as the average value of the sampled feedback comes closer to the one-cycle average value $\langle i_L \rangle$.

In case the digital control system features outer loops that generate a dc current reference, the aliasing issues may be less critical as the resulting dc offset would be compensated by the outer controller. In case the current reference is an ac signal, the aliasing may cause a distortion [34,66]. A detailed analysis of the aliasing-related problems is out-of-scope of this thesis; however, it is an important and under-investigated topic related to MS-PWM.

4.4 **Ripple modulation**

In this section, the ripple modulation effect is explained by first introducing the concept of the modulator's gain. The ripple modulation is well-documented for NS-PWM and, as it will be shown later on in this chapter, its impact remains practically unchanged for MS-PWM with high values of N. The following analysis assumes the implementation of NS-PWM, to simply introduce general notions that are later extended to the analysis of the MS-PWM nonlinearities.

4.4.1 Gain of the pulsewidth modulator

A way to describe the pulsewidth modulator as an actuator is through its static transcharacteristic [24]. This function relates the steady-state value of the PWM output duty cycle D with the average value of the modulating waveform over one switching period, $\langle m \rangle$. The slope of the transcharacteristic, labeled as $k_{mod} = \frac{dD}{d \langle m \rangle}$, will be referred to as the modulator's gain. It is important not to misconstrue this static gain with the one from the small-signal model, which does not take ripple into account.

It is well known that the gain of NS-PWM depends on the switching ripple in m(t); this effect is termed the *ripple modulation* [31–33, 88, 113, 114]. A detailed analysis and derivations for the gain of single edge modulators in the presence of the switching ripple can be read in [32, 113, 114].



Figure 4.5: Illustration of NS-PWM for a modulating waveform with the triangular ripple.

In this thesis, detailed derivations are omitted and only a simple illustration of the effect is presented. An example of NS-PWM operation with triangular ripple in m is shown in Fig. 4.5. The duty cycle D of the converter is represented using the sum of the up-count (positive slope of w) duty cycle D_u and the down-count (negative slope of w) duty cycle D_d :

$$D = D_u + D_d = \frac{\Delta t_{on,u}}{T_{pwm}} + \frac{\Delta t_{on,d}}{T_{pwm}},\tag{4.8}$$

- 40 -



Figure 4.6: Illustrations for calculating the modulator's gain in case of m with: (a) no ripple, (b) triangular ripple with slope β .

where $\Delta t_{on,u}$ is the on-time of the switching signal x(t) during the positive slope of w and $\Delta t_{on,d}$ is the on-time during the negative slope of w. The introduced quantities, D_u and D_d , are referred to as the *partial duty cycles*. To illustrate the impact of the switching ripple, two distinct cases are shown in Fig. 4.6, where the modulating waveform is slightly perturbed such that $\langle m_2 \rangle = \langle m_1 \rangle + d \langle m \rangle$. Focusing on the positive slope of w, the result of applying the perturbation $d \langle m \rangle$ is a change of the corresponding partial duty cycle dD_u .

First, let us observe the case without the switching ripple in m, seen in Fig. 4.6a. It is straightforward to conclude that:

$$\frac{\mathrm{d}\mathrm{D}_{\mathrm{u}}}{\mathrm{d}<\mathrm{m}>} = \frac{1}{tg(\gamma)},\tag{4.9}$$

where γ defines the slope of w. In this case, the perturbation d < m > yields an equal response for both slopes of the triangular carrier, i.e. $dD_u = dD_d$. Therefore, it can be concluded that the resulting modulator's gain is equal to $k_{mod} = \frac{2}{|tg(\gamma)|}$. For the normalized triangular carrier, the counter slope is equal to $tg(\gamma) = 2$, which results in $k_{mod} = 1$. For the trailing-edge and leading-edge carriers, the slope of w is decreased to $|tg(\gamma)| = 1$ and only one edge of x is being modulated; hence, again $k_{mod} = 1$. It is clear that the unity gain will be present also for US-PWM, if the switching ripple is not sampled or is completely removed by filtering.

Now, let us observe the case where m contains the triangular switching ripple, as seen in Fig. 4.6b. Again focusing on the positive slope of w, a few geometrical identities allow us to find the response to the perturbation d < m >. The slope of m, just before its intersection with w, is defined by the *tangent* of the angle β . The set of equations necessary to find the modulator's gain is:

$$\Delta = \frac{d < m >}{tg(\gamma)}$$

$$\frac{\Delta - dD_{u}}{y} = \frac{\Delta}{d < m >}$$

$$\frac{y}{dD_{u}} = tg(\pi - \beta) = -tg(\beta).$$
(4.10)

From (4.10), it is straightforward to find the resulting response of D_u :

$$\frac{\mathrm{dD}_{\mathrm{u}}}{\mathrm{d}<\mathrm{m}>} = \frac{1}{tg(\gamma) - tg(\beta)} = \frac{1}{tg(\gamma)} \frac{1}{1 - \frac{tg(\beta)}{tg(\gamma)}},\tag{4.11}$$

which reduces to (4.9) for the slope $tg(\beta)$ equal to zero, i.e. for m without the switching ripple. For the triangular carrier and m that contains switching ripple, correlating (4.11) with k_{mod} is not trivial as dD_u and dD_d may differ, even in their signs. This fact is mentioned to result in a time variant modulator's gain for the triangular carriers [32]. However, the principles regarding the gain dependency on the slope of m at the intersection with w remain similar.

Let us introduce here, for the first time, the terms *in-phase* and *counter-phase* operation. The modulator operation is referred to as the *in-phase* if the derivative of m(t), just before its intersection with w(t), has the same sign as the derivative of w(t), i.e. $tg(\gamma) \cdot tg(\beta) > 0$. The modulator operation is referred to as the *counter-phase* if the derivative of m(t), just before its intersection with w(t), has the opposite sign to the derivative of w(t), i.e. $tg(\gamma) \cdot tg(\beta) < 0$. From (4.11), it is clear that, with respect to the partial duty cycle D_u , the *counter-phase* operation decreases the gain, while the *in-phase* operation increases it. As explained in Section 4.2.2, for the high-frequency proportional-dominant controllers, the modulating waveform ripple is just an inverted and scaled version of the feedback variable's ripple. For buck converters with a constant output voltage, the switching ripple in i_L is determined by the integral of the switching signal x; hence, without any filters or delays, m and w are always in the *counter-phase* relation (as seen in Fig. 4.5). However, with the *in-phase* one. Similar conclusions are brought in [113].

As the switching ripple shape and, hence, the slope $tg(\beta)$, depend on the duty cycle, it is clear that the *ripple modulation* causes the modulator's gain to be operating point dependent. In [31] it is reported that this effect can be partially compensated for the single-edge modulators, by pre-distorting the carrier.

4.4.2 Calculating the modulator transcharacteristic for NS-PWM

In this section, the procedure for calculating the modulator transcharacteristic is presented for NS-PWM. The extension to MS-PWM is given in Section 4.5.4.1.

To calculate the modulator's static transcharacteristic, it is necessary to find the steady-state waveform m(t) with respect to w(t) inside one switching period of the converter.

For this, it is first necessary to find the ripple component m^{ac} . This waveform can be calculated analytically by solving a set of differential equations that describe the closed-loop control system. For the supposed steady-state duty cycle D, the converter's circuit is solved for all topological states and the ripple in the output waveform is found. If the control loop features any feedback filtering, the corresponding set of differential equations is used to find the filtered ripple component. The filtered output ripple is then used to solve the set of controller-related equations taking into account the negative feedback sign, which results in m_1^{ac} . Finally, an arbitrary amount of time delay $t_d = \tau_D T_{pwm}$ is added



Figure 4.7: (a) Illustration of a current control loop, for a buck converter with a constant output voltage. The control system features a proportional gain and a delay t_d . (b) The corresponding switching ripple components.

to obtain m^{ac} . As mentioned in the previous section, the time delay can impact the slope of m at the intersection with w and, hence, also the modulator's gain. It will be shown in the following sections that the time delay also plays a crucial role in determining the types of nonlinearities that arise with MS-PWM.

The procedure described above is illustrated in Fig. 4.7 for the proportional current control, without feedback filters. It is assumed that the output voltage is constant and that the converter operates in steady-state. It is important to notice that the resulting ripple component m^{ac} is found relative to the applied switching signal x and its shape does not depend on the position of the actual switching instants,

determined by the intersections of m and w. For this reason, the time is labeled as t' in Fig. 4.7b, to emphasize that it is not synchronized with the carrier but with the turn-on instant t_{on} of x. As an illustration corresponding to Fig. 4.7 with $\tau_d = 0$, Fig. 4.8 is given. It is clear that $t' = t - t_{on}$.



Figure 4.8: Illustration of NS-PWM waveforms, corresponding to Fig. 4.7 for $t_d = 0$.

The next step is to determine the average value of m and its position with respect to w. The triangular carrier waveform is obtained by substituting $\zeta = 0.5$ in (2.1):

$$w(t) = \begin{cases} 1 - 2\frac{t}{T_{pwm}}, & \text{if } 0 \le t \le \frac{T_{pwm}}{2} \\ 2\frac{t}{T_{pwm}} - 1, & \text{if } \frac{T_{pwm}}{2} < t < 1 \end{cases}.$$
(4.12)

Let us define the slope of the modulating waveform as $s_m(t) = \frac{\mathrm{d}m(t)}{\mathrm{d}t} = \frac{\mathrm{d}m^{ac}(t)}{\mathrm{d}t}$. As the switching instants are determined by the intersections between two continuous functions, m and w, the following identities hold:

$$m(t_{on}) = w(t_{on})$$

$$m(t_{on} + DT_{pwm}) = w(t_{on} + DT_{pwm})$$

$$m(t_{on} + DT_{pwm}) = m(t_{on}) + I$$

$$I = \int_{t_{on}}^{t_{on} + DT_{pwm}} s_m(t) dt = \int_0^{DT_{pwm}} s_m(t') dt'.$$
(4.13)

It is important to note that the integral I is completely determined without knowing the value of t_{on} . Actually, being an integral of the slope, this value is determined by $I = m^{ac}(t' = DT_{pwm}) - m^{ac}(t' = 0)$. Using (4.12) and (4.13), the turn on instant can be found as:

$$t_{on} = \frac{T_{pwm}}{2} \left((1-D) + \frac{1}{2}I \right).$$
(4.14)

Once the value of t_{on} is calculated, the modulating waveform m(t) is found as:

$$m(t) = m(t_{on}) + \int_{t_{on}}^{t_{on}+t} s_m(t) dt = w(t_{on}) + \int_{t_{on}}^{t_{on}+t} s_m(t) dt = 1 - 2\frac{t_{on}}{T_{pwm}} + \int_{t_{on}}^{t_{on}+t} s_m(t) dt.$$
(4.15)

From (4.15), the average value $\langle m \rangle$, for the initially set value of D, may be found. This pair determines one point of the modulator transcharacteristic. The complete transcharacteristic is obtained by repeating the procedure above for all $D \in [0, 1]$. For the leading- and trailing-edge carriers, the only change that needs to be made is to replace the carrier parameter ζ in (2.1).

As an illustration of the *ripple modulation* in NS-PWM, Fig. 4.9 shows the transcharacteristics obtained using the procedure above, for several values of τ_D . The calculations were performed for the current controlled buck converter, as in Fig. 4.7a, with the proportional gain determined as $k_p = 2\pi f_c \frac{L}{V_{in}}$ to set the crossover frequency to $f_c = 0.1 f_{pwm}$. As explained in Section 4.2.2, for the proportional current control, the exact hardware parameters do not change the modulating waveform properties as long as the analysis is performed with respect to the relative crossover frequency. An interesting thing to note in Fig. 4.9 is that, for the analyzed modulation with the triangular carrier and the triangular switching ripple, for both $\tau_D = 0$ and $\tau_D = 0.5$, the modulator's gain is constant and equal to 1, like for the case of m without ripple.



Figure 4.9: Examples of analytically obtained NS-PWM transcharacteristics for 4 values of the time delay. Calculations are performed for the proportional inductor current control loop shown in Fig. 4.7a, for $f_c = 0.1 f_{pwm}$.

The analysis and illustrations in this section were given for NS-PWM. For MS-PWM with high values of N, the sampled feedback becomes very close to the analog waveform. Hence, similar effects of the *ripple modulation* are expected [31,32], which is validated in Section 4.5.4.1.

Note also that the above-described procedure for calculating m allows to determine the maximal bandwidth that satisfies the condition for smooth modulation, described in Section 2.3.2.

4.5 MS-PWM nonlinearities caused by modulating waveform discontinuities

This section discusses the modulator nonlinearities specific to MS-PWM control. Depending on the types of intersections between the modulating waveform and the carrier, the nonlinearities are manifested as reduced-gain, zero-gain, and infinite-gain zones in the static modulator transcharacteristic. The nonlinear behavior is caused by the modulating waveform jump discontinuities, whose pattern depends on the switching ripple, the controller gains, the multisampling factor, and the high-frequency response of the control system. Provisions for minimizing these nonlinearities are presented and a few application examples of converters operating in each of the nonlinear zones are given. For converters with a dc operating point, the nonlinearities can lead to an undesirable response to transients and to an increased jitter of the duty cycle. For converters with an ac operating point, the nonlinearities can also lead to the output waveform distortion.

This section begins by classifying the types on intersections between m and w and explaining their general impact on a converter's operation. This part is completely independent of the converter's topology or the control system's architecture. Subsequently, the quantitative analysis is performed for two-level, current-controlled, half-bridge and full-bridge buck converters. As mentioned at the beginning of this chapter, the same methodology can be applied to other converter topologies or feedback variables.

4.5.1 Multi-sampled control loop settings

The control system that will be referred to in this section corresponds to the one of Fig. 4.1. The triangular carrier is used and the modulating waveform update instants are equal to those in (2.2).

In general, with multiple update instants throughout the switching period, more than 2 intersections between m and w may occur. To prevent an increased number of transistor commutations, the switching action is determined based on the first intersection between m and w, for each slope of the carrier. This is equivalent to allowing the turn on only during the negative slope of the carrier, and the turn off only during the positive one. Multiple intersections can be caused by a transient or, in the steady-state, by the switching ripple, as is explained in the following sections. It should be mentioned that, for large-signal disturbances, it may be favorable to allow multiple switching actions, which would result in a faster nonlinear response and a variable switching frequency during transients [91, 118, 119].

Operation of the MS-PWM modulator for N = 4, with the chosen update instants and the logic that prevents multiple switching actions is illustrated in Fig. 4.10.



Figure 4.10: Illustration of the modulating waveform m(t), the carrier w(t), and the switching signal x(t) for one switching period of MS-PWM with N = 4. The chosen update instants are illustrated as arrows. The illustration also shows that multiple switching is avoided by allowing only the turn-off to occur during the positive slope of the carrier.

4.5.2 Types of intersections between the modulating waveform and the carrier

An important difference between MS-PWM, NS-PWM, and (S/D)S-PWM² is that, for MS-PWM, an update of m can cause a vertical crossing with w, i.e. a switching action may occur when $m \neq w$. For NS-PWM, even if m is not constant over T_{pwm} , it is a continuous function; hence, it is always true that m = w at a given switching instant. For MS-PWM, the discontinuous nature of m gives rise to various types of intersections with w, each of which brings a unique impact on the modulator transcharacteristic and, hence, k_{mod} [27,66]. The modulator transcharacteristic can be defined in the same way as for NS-PWM. The average value of the modulating waveform over one switching period, < m >, is found as the average of N controller outputs m_s starting from when w = 1, i.e. in the range $\frac{t}{T_{pwm}} \in [0, 1)$. In Fig. 4.10, this corresponds to $< m > = \frac{m_{s,1}+m_{s,2}+m_{s,3}+m_{s,4}}{4}$. It will be shown that an MS-PWM transcharacteristic generally features distinct zones with specific gains, which are referred to as the nonlinearity zones. All types of intersections between m and w are shown in Fig. 4.11 and the corresponding nonlinearity zones are illustrated in Fig. 4.12.

For (S/D)S-PWM, the update of m occurs when w is equal to 0 or 1. This always yields horizontal crossings between m and w and, as explained in Section 4.4.1, results in the unity modulator's gain $k_{mod} = 1$.

For MS-PWM, an intersection between m and w can be either horizontal or vertical. In Fig. 4.11a, horizontal crossings are shown. For cases when horizontal crossings occur for both slopes of w, the

²With update instants that coincide with peaks and valleys of the carrier.



Figure 4.11: Types of intersections between the modulating waveform m and the carrier w: (a) horizontal crossings; (b) counter-phase vertical crossings; (c) in-phase "vertical" crossings. The closest update of m to the intersection with w is labeled with a blue dot. The intersection that defines the switching signal x is labeled with a red cross.

modulator's gain is $k_{mod} \approx 1$. As for the NS-PWM modulators, the gain is not exactly equal to 1, due to the ripple modulation³ described in Section 4.4. The operation with double horizontal crossings will be referred to as the linear operation of the modulator, as the duty cycle is determined by a unique modulating waveform and both edges of x are equally modulated by applying a small perturbation d < m >. In Figs. 4.11b and 4.11c, vertical intersections are shown. It is clear from Fig. 4.10 that vertical intersections between m and w may only occur at instants when the modulating waveform is updated and the carrier is not equal to 0 or 1.

4.5.2.1 Critical duty cycles and critical modulating segments

For the triangular carrier parametrized in (4.12) and the sampling instants chosen as in (2.2), the vertical intersections may occur only when the normalized value of w is equal to $\frac{2i}{N}$, where $1 \le i < \frac{N}{2}$ and i is an integer. Assuming that both switching instants occur for a similar value of w, the duty cycle in the

³In Section 4.5.4.1, it will be shown that for high values of N, the impact of the ripple modulation on k_{mod} is practically the same as for NS-PWM. For low values of N, the ripple modulation is not easily distinguished from the discontinuity-related effects, described in this section, which also impact the modulator's gain.



Figure 4.12: Types of zones that may appear in MS-PWM transcharacteristics: (a) linear zone corresponding to double horizontal crossings; (b) reduced-gain zone corresponding to a single counter-phase vertical crossing; (c) zero-gain (dead-band) zone corresponding to double counter-phase vertical crossings; (d) infinite-gain (jitter) zone corresponding to the in-phase "vertical" crossings. Extensions of the nonlinearity zones are labeled with δ .

presence of vertical crossings is close to the value of $\frac{2i}{N}$. This precise operating point, labeled as D_c , is defined as the *critical duty cycle*. It will be shown that the nonlinearity zones in the transcharacteristics are located around these values of D. For a given N, the *critical duty cycles* are found as:

$$D_c = \frac{2i}{N} \quad , \quad 1 \le i < \frac{N}{2},$$
 (4.16)

Note that the exact operating points, for which the vertical intersections may occur, span a certain range around D_c .

The critical modulating segments are the two adjacent segments of m before and after the update instant closest to the intersection between w and m, i.e. those shown in Fig. 4.11. For each operating point, there are two pairs of critical modulating segments, one for the positive and one for the negative slope of the carrier.

Finally, let us define the *discontinuities* of the critical modulating segments for both positive $(\Delta m_{c,u})$

and negative $(\Delta m_{c,d})$ slopes of w as:

$$\Delta m_{c,u} = m(T_{c,u} + \epsilon) - m(T_{c,u} - \epsilon)$$

$$\Delta m_{c,d} = m(T_{c,d} - \epsilon) - m(T_{c,d} + \epsilon)$$
(4.17)

where ϵ is an infinitely small positive number and $T_{c,u}$ and $T_{c,d}$ are the modulating waveform update instants closest to the intersection with the carrier.

For further classification, let us again recall the *in-phase* and *counter-phase* operating regimes. The in-phase operating regimes occur when the difference between values of the *critical modulating segments* has the same sign as the slope of w. The counter-phase operating regimes occur when the difference between values of the *critical modulating segments* has the opposite sign compared to the slope of w. It will be shown that, regarding vertical crossings, different behavior is obtained for the counter-phase and the in-phase regimes. Note that, the signs in (4.17) are chosen so that a positive value of $\Delta m_{c,u,d}$ always corresponds to the in-phase condition.

An illustration of the critical modulating segments and their discontinuities is shown in Fig. 4.13.



Figure 4.13: Illustration of the critical modulating segments and their discontinuities for (a) counterphase and (b) in-phase operation. The modulating waveform update instants closest to the intersection with the carrier are labeled as $T_{c,u}$ for the positive slope of w and $T_{c,d}$ for the negative slope of w. The turn-on and turn-off instants are labeled with t_{on} and t_{off} , respectively.

4.5.2.2 Counter-phase vertical crossings

In Fig. 4.14, the counter-phase vertical crossings are shown. A counter-phase vertical crossing prevents the modulation of the corresponding edge of x(t), which reduces the modulator's gain [27].

The operation with a single counter-phase vertical crossing is shown in Fig. 4.14a. Let us assume that the steady-state operation is determined by m(t) drawn using the dark blue color. By applying a perturbation d < m >, the modulator's gain determines the response dD = dD_u + dD_d. For the analyzed case, the rising edge of x is not modulated at all (dD_d = 0). Hence, neglecting the ripple modulation, the modulator's gain is effectively halved, i.e. $k_{mod} \approx \frac{1}{2}$. The corresponding nonlinearity zone in the modulator transcharacteristic, referred to as the reduced-gain zone, is illustrated in Fig. 4.12b.



Figure 4.14: Illustration of MS-PWM operation with N = 4 and (a) single and (b) double counter phase vertical crossing. For a small change d < m >, the edge of x determined by a vertical crossing is not being modulated.

Now let us consider the case shown in Fig. 4.14b. There, both edges of x are locked due to double vertical crossings and the perturbation of m does not at all affect the applied D. In this case, the converter practically works in open-loop conditions, with $D = D_c$, and the control system is ineffective, i.e. $k_{mod} = 0$. The corresponding nonlinearity zone in the modulator transcharacteristic, referred to as the dead-band or the zero-gain zone, is illustrated in Fig. 4.12c.

The impact of the reduced- and, especially, the zero-gain zones on a response to small- or large-signal disturbances is very difficult to analyze in a general form and a case-by-case study is needed. For dc-dc converters, several examples of deteriorated transient responses are given in [24, 27]. In Section 4.5.6.2, it will be shown that, for converters with an ac operating point, these nonlinearity zones result in the

output distortion.

It should be noted that, for the MS-PWM implementation in certain DSP platforms, an additional logic must be implemented to prevent the *pulse-skipping* phenomenon [18, 72]. Namely, if the PWM output is toggled only when m = w, it is necessary to predict a vertical crossing and force the switching action to occur. In case FPGAs are used, the *pulse-skipping* is easily prevented by using the logic comparison $m \ge w$.

4.5.2.3 In-phase "vertical" crossings



Figure 4.15: Illustration of MS-PWM operation with N = 4 and a single in-phase "vertical" crossing. For a small change d < m >, the response of the falling edge of x is determined by the corresponding critical modulating segment discontinuity $\Delta m_{c,u}$.

Consider the case shown in Fig. 4.15, where the in-phase "vertical" crossing occurs just after the falling edge of x. With the logic that prevents multiple commutations, the in-phase "vertical" crossings are actually turned into horizontal crossings; however, this operation brings another nonlinear phenomenon. Namely, for the illustrated case, by applying a small positive perturbation d < m >, the falling edge of x is no longer triggered by the same modulating segment, but by the following one. Hence, the resulting response dD_u is not dominantly affected by the value of d < m > but by the critical modulating segment discontinuity $\Delta m_{c,u}$, which can result in a large gain of the modulator. The in-phase "vertical" crossings can appear for both slopes of w, or in combination with a horizontal or a counter-phase vertical crossing.

In the following section it will be shown that, due to the discontinuity of the modulating waveform, the in-phase operation yields a whole range of duty cycles for which a steady-state operation cannot be found. For those duty cycles, the transcharacteristic cannot be defined; however, to indicate its impact, the corresponding nonlinearity zone will be represented by a vertical line, i.e. $k_{mod} \rightarrow \infty$, as illustrated in Fig. 4.12d. These nonlinearity zones will be referred to as the infinite-gain or jitter zones, because they result in LCOs.

4.5.3 Duty cycle jittering caused by the in-phase "vertical" crossings

Discovery and analysis of the effects caused by the in-phase "vertical" crossings are amongst the original contributions of this thesis. Therefore, this entire section is devoted to the analysis of this specific discontinuity-related nonlinearity.



4.5.3.1 Explanation of the jittering mechanism

Figure 4.16: Example of an operating condition that may result in the jitter amplification, for MS-PWM with N = 4.

The jitter amplification phenomenon, described in this section, is a result of the in-phase discontinuity of m. The mechanism behind it is illustrated in Fig. 4.16, for N = 4.

Consider an initial operating point, shown in Fig. 4.16, where the falling edge of x is triggered by the intersection between w and the modulating segment $m_{off,1}$. This determines the duty cycle $D_1 = D_d + D_{u,1}$. In case the closed-loop control results in a small change of the controller's output, such that a higher duty cycle is required (a value between $D_d + D_{u,1}$ and $D_d + D_{u,2}$), the modulating waveform is incremented by a certain amount d < m >. As illustrated in Fig. 4.16, a very small positive change of the segment $m_{off,1}$ causes the intersection between m and w to be determined by the following segment $m_{off,2}$, which changes the up-count partial duty cycle from $D_{u,1}$ to $D_{u,2}$. Since the applied duty cycle is higher than the required one, the controller will react by decreasing the value of m, which may cause the segment $m_{off,1}$ again to intersect with w in the following switching period. This causes limit cycling (duty cycle jittering) by an extent that is determined by the discontinuity of these critical modulating segments: $x = f(\Delta m_{c,u}, t)$. As it will be shown in the following sections, the critical modulating segments can exhibit a large discontinuity, resulting in a very high modulator's gain under this condition. Let us here assert that, around each D_c , the in-phase discontinuity of the critical modulating segments determines the height of the jitter zone δ_{∞} in an MS-PWM transcharacteristic (see Fig. 4.12d.). Without a formal proof, this relation comes from that fact that, for a fixed switching ripple, this discontinuity defines a range of duty cycles that cannot be achieved in steady-state. Referring to Fig. 4.16, this range is equal to $D_{u,2} - D_{u,1} = \frac{\Delta m_{c,u}}{2}$. In an MS-PWM transcharacteristic, this corresponds to the respective jitter zone height being equal to $\delta_{\infty} = \frac{\Delta m_{c,u}}{2}$. For the case when the in-phase regime is present for both slopes of the carrier, the accumulated height⁴ of the jitter zones, around the observed D_c , can be estimated as:

$$\Delta D_{\infty} = \delta_{\infty,u} + \delta_{\infty,d} = \frac{\Delta m_{c,u}^{in-p}}{2} + \frac{\Delta m_{c,d}^{in-p}}{2},\tag{4.18}$$

where Δm_c^{in-p} labels the in-phase discontinuity. As noted before, the in-phase operation does not always occur simultaneously for both slopes of w, in which case only one Δm_c^{in-p} in (4.18) is present.

4.5.3.2 Statistical approach for the jitter modeling

To gain a quantitative insight into the effects that jittering imposes on a converter's output, a simple statistical approach is proposed. The principle behind the described effect is that a small change of the modulating segment can lead to a large change of the switching pulse width. In Fig. 4.16, the mechanism is shown for the jittering of the falling edge of the switching impulse, but it can equally appear for the rising edge or for both the edges.

Consider again the operating condition shown in Fig. 4.16. Assuming that the rising edge of the switching impulse is fixed (the corresponding value of the modulating segment is equal to m_{on}), the applied duty cycle depends on the value of the critical modulating segment that first intersects with the carrier during the positive slope of w. The relevant critical modulating segments are $m_{off,1}$ and $m_{off,2}$, which correspond to the duty cycles equal to $D_1 = D_d + D_{u,1}$ and $D_2 = D_d + D_{u,2}$, respectively. Assuming that the controller tries to impose a duty cycle value exactly in between D_1 and D_2 , it can be supposed that there is an equal probability that the switching will occur at $t_{off,1}$, resulting in D_1 , or at $t_{off,2}$, resulting in D_2 . In other words, dithering will occur between the boundary values D_1 and D_2 . Hence, the expected value [120] of the applied duty cycle μ_D is equal to:

$$\mu_D = \frac{1}{2} \left(D_1 + D_2 \right) = \frac{1}{2} \left(m_{on} + \frac{m_{off,1} + m_{off,2}}{2} \right)$$
(4.19)

The variance [120] of the applied duty cycle σ_D^2 is calculated as:

$$\sigma_D^2 = \frac{1}{2} \left(\left(D_1 - \mu_D \right)^2 + \left(D_2 - \mu_D \right)^2 \right) = \frac{\Delta m_{c,u}^2}{16} = \frac{\delta_{\infty,u}^2}{4}$$
(4.20)

where $\Delta m_{c,u} = m_{off,2} - m_{off,1}$. In case the jittering occurs simultaneously for both slopes of w, the

⁴As it will be shown, if the in-phase operation is present for both slopes of w, the resulting transcharacteristic will feature two jitter zones around the corresponding D_c that are either separated or merged together, depending on the size of the discontinuity of m.
value of δ_{∞} should be replaced by ΔD_{∞} from (4.18). Equation (4.20) shows that the variance of the duty cycle changes as a quadratic function of the discontinuity of the critical modulating segments. Therefore, any reduction of this discontinuity is expected to strongly reduce the negative effects on a converter's output. Note that the actual expected value μ_D in (4.20) depends on the duty cycle imposed by the controller, which can be anywhere between the boundary values D_1 and D_2 . A merit of this simplified variance expression is that it provides an intuitive quantification of the jittering impact based only on Δm_c , and not on the exact operating point.

4.5.3.3 Algorithm for jittering prevention

This section proposes an algorithm that can be used for cancelling the duty cycle jittering. The algorithm is intended for use under dc or slowly-varying operating points; however it is designed such as not to impair a response to large-signal transients. The flowchart of the algorithm is shown in Fig. 4.17.



Figure 4.17: Flowchart of the algorithm proposed for cancelling the duty cycle jittering. At the sampling instant [n], in case the jittering is detected and a large-signal transient is not present, the controller's output $m_s[n]$ is discarded and the previous value is used instead.

The idea behind the proposed method is to block the update of the second segment in a pair of critical modulating segments, in case the jittering is detected. The algorithm first detects whether the switching action is occurring near the modulating waveform update, by comparing the value of the modulating segment with the value of the carrier at the following update instant. If the absolute value of the time difference between the switching action and the modulating waveform update is below a chosen value Δt^{flag} , referred to as the *flag* zone time-span, the *flag* is raised. If the absolute value is above Δt^{flag} , the *flag* is cleared and the modulating waveform update is always allowed. When the *flag* is raised, if the detected operation is counter-phase, the modulating waveform update is always allowed. If the in-phase operation is detected, in order to start blocking the update, additional indicators are used. In Fig. 4.17, they are labelled as *up-count jittering* and *down-count jittering*, for both slopes of the carrier. These indicators prevent blocking the modulating waveform update in case the *flag* is raised, but the jittering has not yet occurred. These indicators are set when, for the first time since the *flag* is raised, for the respective carrier slope, the switching action moves from one critical modulating segment to the other.

By preventing the modulating waveform update the discontinuity effect is cancelled, however, care must be taken regarding large-signal disturbances. Namely, regardless of the position relative to the carrier, the modulating waveform should always be updated if a large disturbance is detected. This prevents delays when reacting to faults or reference changes. For this reason, adaptive limits for critical modulating segment differences are used to decide whether to update or not. In Fig. 4.17, the limit is labelled as Δ_u for the up-count and Δ_d for the down-count jittering. There are two distinct limits, as the jittering can occur for both slopes of the carrier, independently. These limits are adapted using the previous value of Δm_c , which the algorithm tracks, imposing a 50% higher margin, heuristically. This margin should be sized to ensure that the control system is robust to noise present in the experimental set-up. On the other hand, over-sizing the margin deteriorates the capability of the algorithm to respond to large-signal disturbances. Each time the counter-phase operation is detected, or the *flaq* is cleared, the adaptive limits are reset to 0. This allows the first in-phase change of the critical modulating segment pair to be always updated, which is required to set the limit based on the current operating point. After the limit is set, if a large-signal disturbance occurs, for which the difference between the critical segments is higher than the set limit, the modulating waveform update is allowed. Cases when the adaptive limit is set by the large-signal disturbance do not cause problems for the algorithm, as the operating point significantly changes, which causes the *flag* to be cleared and the limit to be reset to 0. Note that the magnitude of a large-signal disturbance that clears the flag is correlated to the choice of the flag zone time-span Δt^{flag} , which is why that value should also be sized carefully.

An initial guess for Δt^{flag} , below which the *flag* is raised, can be made using the modulating waveform discontinuities⁵. Consider again the operating point shown in Fig. 4.16; the *flag* zone time-span can be chosen as $\Delta t^{flag} = t_{off,2} - t_{off,1}$. This results in $\Delta t^{flag} \approx \frac{1}{2} \Delta m_{c,u} T_{pwm}$, considering that $t_{off,1}$ is close to the relevant update instant. Such design choice is made for the subsequent results in this thesis.

To summarize, in order for the algorithm to start preventing the update, at least 3 switching periods with the *flag* raised are needed. The first one is needed to set the *flag* if the switching action is occurring near the modulating waveform update. The second period is required for the jittering to occur for the first time. During the second period, the adaptive limit is calculated. It becomes active in the next,

 $^{{}^{5}}$ The discontinuities can be calculated off-line based on the control system parameters. The procedure for their calculation is analyzed in Section 4.5.4.

third, period when the jittering prevention starts.

4.5.3.4 Approximate in-phase conditions for current control loops with one step delay and low-pass feedback filters

For reasons described in Section 4.4.1, for inner current control loops the counter-phase operation is enabled by default. Consider again the system shown in Fig 4.1 (as noted in Section 4.2.2, the same analysis is valid for the full-bridge configuration). Blocks that may impact the shape of m, and hence also its intersection with w, are the feedback filters and the controller. This section offers a simplified analysis of how this cascade may change the operation from the counter-phase to the in-phase.

It is mentioned in Section 4.4.1 that, for current control with NS-PWM, the time delay may change the operation from the counter-phase to the in-phase. For this reason, it is expected that a similar impact on m may be brought by low-pass filters, due to the introduced phase lag at high frequencies. Therefore, the analysis is performed for the control system with a first-order low-pass filter in the feedback. The discretization is, arbitrarily, performed using the bilinear transform:

$$G_{fb}(z) = \frac{a_f}{a_f + 2} \frac{z + 1}{z + \frac{a_f - 2}{a_f + 2}} \quad , \quad a_f = \omega_{c,f} T_s, \tag{4.21}$$

where $\omega_{c,f}$ is the filter's angular cut-off frequency. Regarding the controller, the analysis is performed for the discretized version of the PI structure from (4.3), with the added one step computation delay:

$$G_c(z) = \left(k_p + k_i T_s \frac{1}{1 - z^{-1}}\right) z^{-1}.$$
(4.22)

The in-phase conditions are derived for $N = \{4, 6, 8\}$, by calculating the discontinuity of the critical modulating segments. Consider the illustration shown in Fig. 4.18. For N = 4, there is only one critical segment pair per slope of w, i.e. $\Delta m_{c,u}^{1,2} = m_{4,2} - m_{4,1}$, where 4 indexes the oversampling factor. This critical segment pair may cause vertical crossings for operating points near $D_c = \frac{1}{2}$. For N = 6, it is of interest to observe $\Delta m_{c,u}^{1,2} = m_{6,2} - m_{6,1}$ for operating points near $D_c = \frac{1}{3}$ and $\Delta m_{c,u}^{2,3} = m_{6,3} - m_{6,2}$ for operating points near $D_c = \frac{1}{4}$. $\Delta m_{c,u}^{2,3} = m_{8,3} - m_{8,2}$ for operating points near $D_c = \frac{1}{2}$, and $\Delta m_{c,u}^{3,4} = m_{8,4} - m_{8,3}$ for operating points near $D_c = \frac{1}{2}$, and $\Delta m_{c,u}^{3,4} = m_{8,4} - m_{8,3}$ for operating points near $D_c = \frac{3}{4}$.

As seen from (4.7), for buck-type converters the switching ripple is anti-symmetric around D = 0.5. For this reason, the properties derived for the positive slope of the triangular carrier w, for a certain value of D, are the same as those for the negative slope of w, for 1 - D. Hence, all derivations are given only for the positive slope of w. An illustration of this anti-symmetry is shown in Fig. 4.19 for simulated modulating waveforms of the current-controlled system from Fig. 4.1, obtained for D = 0.67 and D = 0.33. It is clear that the discontinuities of $m^{0.33}$ and $m^{0.67}$, calculated using the sign convention from (4.17), are the same for opposite slopes of w, i.e. $m_{6,4}^{0.67} - m_{6,5}^{0.67} = m_{6,1}^{0.33} - m_{6,1}^{0.33} - m_{6,6}^{0.33} = m_{6,2}^{0.67} - m_{6,2}^{0.67}$.

Ph.D. Thesis — Ivan Petrić



Figure 4.18: Illustration of the critical modulating segments for the positive slope of w and: (a) N = 4 around $D_c = \frac{1}{2}$, (b) N = 6 around $D_c = \frac{1}{3}$ and $D_c = \frac{2}{3}$, (c) N = 8 around $D_c = \frac{1}{4}$, $D_c = \frac{1}{2}$, and $D_c = \frac{3}{4}$.

For the simplified analysis in this section, the following assumptions and approximations are made. First, as before, it is assumed that the output voltage dynamics are slow, i.e. v_o in Fig. 4.1 is constant. Secondly, the converter is assumed to work with the zero reference tracking error, at the operating point equal to the analyzed critical duty cycle, i.e. $D = D_c$. Additionally, it is assumed that inside one T_{pwm} , both intersections between m and w occur exactly for $w = D_c$, i.e. $D_u = D_d = \frac{D_c}{2}$. This fixes the switching signal x to be centred around w = 0, which removes the necessity to re-iterate the calculation of the switching ripple component, once the actual intersections between sampled m and w are found. Referring to Fig. 4.4c and the sampling instants from (2.2), this approximation means that, at the peaks/valleys of w the average current is sampled (the controller error is 0), while for the other sampling instants the current error is equal to a fraction of the switching ripple.

With these assumptions, for the current ripple with its peak-to-peak value $\Delta i_L^{p-p} = 2\Delta I$, it is possible to directly compute the modulating waveform discontinuities, using the difference equations corresponding to G_c and G_{fb} . The resulting expressions for Δm_c are given in (4.23) - (4.27), where



Figure 4.19: Illustration of the anti-symmetry between modulating waveforms for D = 0.67 and D = 0.33, for the proportional current control system with N = 6. The modulating waveforms are obtained from simulations, using the proportional controller with $k_p = 2\pi f_c \frac{L}{V_{in}}$, for $f_c = 0.1 f_{pwm}$.

 $\omega_{i,rel} = \frac{\omega_i}{2\pi f_s} = \frac{\omega_i}{2\pi N f_{pwm}} = \frac{T_s}{2\pi} \frac{k_i}{k_p}$ is the relative angular frequency of the zero related to the integral action. By comparing the calculated discontinuities with zero, it is possible to find the value of the filter's cut-off frequency, for which the operation turns from the counter-phase to the in-phase. The equation is not shown for N = 8 and $D_c = \frac{3}{4}$, because already for N = 8 and $D_c = \frac{1}{2}$, the operation is always counter-phase (the first-order low-pass filter cannot provide enough phase lag to enable the in-phase operation, at least for realistic values of $\omega_{i,rel}$). The same is valid for the case of N = 6 and $D_c = \frac{2}{3}$ in (4.25). This indicates that, for the current control loops, the in-phase operation may occur only for the first pair of the critical modulating segments of the respective slope of w. For the positive slope of w this corresponds to the lowest value of D_c while for the negative slope it corresponds to the highest value of D_c . Note again that these expressions are given for the controller that features one step computation delay, which on its own has an impact on bringing the operation closer to the in-phase condition.

$$\Delta m_{c,u}^{1,2}\big|_{N=4,D_c=\frac{1}{2}} = \frac{k_p a_f \Delta I}{a_f^2 + 4} \bigg[4\pi \omega_{i,rel} - a_f + 2 \bigg]$$
(4.23)

$$\Delta m_{c,u}^{1,2}\big|_{N=6,D_c=\frac{1}{3}} = \frac{k_p a_f \Delta I}{(a_f^2 + 12) (3a_f^2 + 4)} \left[-3a_f^3 + 6a_f^2 (2\pi\omega_{i,rel} + 1) - 28a_f + 80\pi\omega_{i,rel} + 24 \right]$$
(4.24)

$$\Delta m_{c,u}^{2,3}\Big|_{N=6,D_c=\frac{2}{3}} = \frac{k_p a_f \Delta I}{2\left(a_f^2 + 12\right)\left(3a_f^2 + 4\right)} \left[-3a_f^3 \left(2\pi\omega_{i,rel} + 1\right) + 6a_f^2 \left(2\pi\omega_{i,rel} + 1\right) - 52a_f \left(2\pi\omega_{i,rel} + 1\right) + 8\left(10\pi\omega_{i,rel} - 3\right) \right]$$

$$(4.25)$$

Ph.D. Thesis — Ivan Petrić

$$\Delta m_{c,u}^{1,2}\Big|_{N=8,D_c=\frac{1}{4}} = \frac{k_p a_f \Delta I}{3(a_f^2 + 4)(a_f^4 + 24a_f^2 + 16)} \cdot \left[-3a_f^5 + 6a_f^4(2\pi\omega_{i,rel} + 1) - 72a_f^3 + 16a_f^2(14\pi\omega_{i,rel} + 5) - 176a_f + 32(14\pi\omega_{i,rel} + 3) \right]$$

$$(4.26)$$

$$\Delta m_{c,u}^{2,3}\Big|_{N=8,D_c=\frac{1}{2}} = \frac{k_p a_f \Delta I}{2\left(a_f^4 + 24a_f^2 + 16\right)} \left[-a_f^3 \left(2\pi\omega_{i,rel} + 1\right) + 2a_f^2 \left(2\pi\omega_{i,rel} + 1\right) - 28a_f \left(2\pi\omega_{i,rel} + 1\right) + 8\left(6\pi\omega_{i,rel} - 1\right) \right]$$

$$(4.27)$$

For the illustration of the filter's impact, a case of N = 4 and $D_c = \frac{1}{2}$ is analyzed in more detail. The in-phase condition is satisfied if (4.23) is greater than zero, which yields the following inequality:

$$a_f < 2\left(1 + 2\pi\omega_{i,rel}\right) \tag{4.28}$$

For the high-frequency proportional-dominant PI controllers, the zero related to the integral action is always significantly below the switching frequency, which allows the simplification of the condition above to:

$$a_f < 2 \implies \omega_{c,f} < \frac{4}{\pi} \omega_{pwm}$$
 (4.29)

Therefore, this approximated analysis tells that, for the analyzed four-time-sampled current control loop, if the angular cut-off frequency of the first-order feedback filter is placed below (4.29), the modulating waveform will be in-phase with the carrier and the jitter amplification may occur. From (4.28) it is evident that the increase of the integral gain results in a higher boundary value of $\omega_{c,f}$ that enables the in-phase regime. This further points to a correlation between the in-phase operation of the current control loops and the phase lag introduced from sampled i_L to m. For the other values of the multisampling factor, the derived expressions are not so simple; however, the trends related to the impact of the filter and the integral action remain the same⁶.

In this section, the expressions for Δm are derived only for the single first-order digital low-pass filter $G_{fb}(z)$ in (4.21). However, based on the noticed correlation between the introduced phase lag and the in-phase condition, it is of interest to see whether these simple expressions can be also used to predict the in-phase operation for some other filters, e.g. for analog low-pass filters. For this reason, the in-phase conditions are transformed to show the dependence on the phase lag at the switching frequency, introduced by the feedback filter.

The boundary lines ($\Delta m = 0$), for the lowest values of D_c and the positive slope of w, are shown

 $^{^{6}}$ In [65], the analysis is performed for the same current control system, with a PID controller. There, it is shown that practically for any non-zero derivative gains, the in-phase operation is prevented. Again, this is consistent with the reasoning that, in current control systems, the phase lag tends to change the operation from the counter-phase to the in-phase. The PID structure is not often used for current controllers; hence, for consistency with the structures used throughout this chapter, the related expressions are not shown.

in Fig. 4.20. The calculations are performed for values of ω_i from 0 ($k_i = 0$) to 0.15 ω_{pwm} . The proportional gain is calculated as $k_p = 2\pi f_c \frac{L}{V_{in}}$, for $f_c = 0.093 f_{pwm}$. Due to the approximations used in the modulating waveform calculations, it was of interest to verify these results. Therefore, simulations were implemented in MATLAB/Simulink, for the buck converter corresponding to the set-up A, described in Table 3.1. Values of m, for different filter's cut-off frequencies, were obtained while the converter was working with a fixed reference corresponding to the duty cycle $D = D_c$. The first set of verifications was performed by implementing $G_{fb}(z)$ as in (4.21); the corresponding results are shown in Fig. 4.20 using circular markers. The second set of verifications, with the results plotted using diamond markers, was performed using an analog low-pass filter:

$$G_{fb}(s) = \frac{\omega_{c,f}}{s + \omega_{c,f}} \tag{4.30}$$

From Fig. 4.20, it can be seen that the results for analog and digital filter implementations match well with the analytical traces and are very close to each other. These results confirm that the presented simplified expressions are well-suited to predict the in-phase operation independently of the specific converter parameters or filter configurations.



Figure 4.20: The in-phase condition boundary lines for the PI current control loops with $f_c = 0.093 f_{pwm}$. The results are shown for N = 4 and $D_c = \frac{1}{2}$, N = 6 and $D_c = \frac{1}{3}$, and N = 8 and $D_c = \frac{1}{4}$. The in-phase condition is satisfied below the boundary lines. The circular markers show simulated results with the digital low-pass filter $G_{fb}(z)$ in (4.21). The diamond markers show simulated results with the analog low-pass filter $G_{fb}(s)$ in (4.30).

To illustrate the impact of the feedback filter from (4.21) on m, the simulated modulating waveforms are shown in Fig. 4.21, for the same set-up as for Fig. 4.20, with N = 4 and $\omega_i = 0.03 \cdot \omega_{pwm}$. From Fig. 4.21, it is clear that without any filters, the operation is counter-phase for both slopes of w. Adding the digital low-pass filter from (4.21), with $\omega_{c,f} = 1.5 \cdot \omega_{pwm}$, almost completely flattens out both critical modulating segment pairs, by bringing the system to the border between the counter-phase and the in-phase operation. It is interesting to notice that for N = 4, due to the anti-symmetry of the triangular switching ripple, the critical modulating segments for both slopes of w are affected in the same way. Moreover, for N = 4, only one critical duty cycle is present. Therefore, this specific filter setting is very attractive, as it prevents any vertical crossings from occurring. This fact will be exploited later on in the thesis. For the case of $\omega_{c,f} = \omega_{pwm}$, the operation is in-phase for both slopes of w.



Figure 4.21: Illustration of the impact of $G_{fb}(z)$ from (4.21) on the intersections between m and w.

4.5.3.5 Examples of operation with duty cycle jittering

This section shows simulation and experimental results for operating regimes that feature the jitter amplification. The results are given for the current control loop of the dc-dc buck converter, corresponding to the set-up A described in Table 3.1. The DSP control platform, described in Section 3.2, is used for the experimental results. The PI controller from (4.22) is used, with $k_p = 0.035$ and $\omega_i = 0.03 \cdot \omega_{pwm}$.

Tests are organized in the following way. For the oversampling factors $N \in \{4, 6, 8\}$, the operation is tested for three feedback configurations. First, as a benchmark, the operation is tested without any feedback filters, which results in the counter-phase operation. Then, the digital low-pass filter (dlpf) from (4.21) is introduced, with the cut-off frequency equal to $\omega_{c,f} = \omega_{pwm}$. The dlpf introduces 52° phase lag at the switching frequency, which, based on Fig. 4.20, enables the in-phase operation for N = 4 and is close to the borderline for N = 6 and N = 8. Finally, the analog low-pass filter (alpf) from (4.30), with the cut-off frequency equal to $\omega_{c,f} = 1.5\omega_{pwm}$ is added in a cascade with the dlpf. The alpf introduces 34° phase lag at the switching frequency. Based on Fig. 4.20, this filter cascade enables the in-phase regime for all tested values of the multisampling factor.

As a first validation, the simulated modulating waveforms corresponding to the tested regimes are plotted in Fig. 4.22, with their discontinuities reported in the legend. The modulating waveforms are obtained for the operation with a fixed reference, corresponding to the lowest critical duty cycle. For the case of N = 4 with both alpf and dlpf, the jittering occurs for $D = D_c = 0.5$; hence, the corresponding modulating waveform is not in steady-state. For other cases with the in-phase operation, the jittering occurs around, but not exactly for $D = D_c$.



Figure 4.22: Simulation results of modulating waveforms for different tested filter configurations: (a) $N = 4, D = D_c = \frac{1}{2}$; (b) $N = 6, D = D_c = \frac{1}{3}$; (c) $N = 8, D = D_c = \frac{1}{4}$.

The following results show three operating conditions, for each tested multisampling factor. Current references are chosen to result in a duty cycle that causes the jitter amplification when the filters are activated. Then, for the same regimes, the results are shown when the anti-jittering algorithm from Section 4.5.3.3 is activated. Finally, when the filters are bypassed, the operating mode is counter-phase, which is used as a benchmark case without the jittering.

Simulations results in Fig. 4.23 are given to examine the time-domain effects of the jittering and the dynamic performance of the anti-jittering algorithm. Responses are tested for N = 6 and a step reference change from $i_{L,r1} = 4.43$ A (D = 0.665) to $i_{L,r1} = 4.51$ A (D = 0.677). The results are given for the cascade of dlpf and alpf, without (Fig. 4.23a) and with (Fig. 4.23b) the proposed anti-jittering algorithm. From Fig. 4.23, it can be seen that the operating point enters the jitter zone at approximately t = 1.6 ms. A short transient is noticed when the anti-jittering algorithm is used, as the blocking of the modulating waveform update begins only after the first jitter is detected. In Fig. 4.23c, the reference tracking errors are plotted in order to observe the algorithm's impact on the transient response. The error traces are plotted relative to the current reference and, for a better visualization, the switching ripple is removed using a moving average filter. It can be seen that the algorithm does not significantly impact the step response, and the error caused by blocking the modulating waveform is quickly compensated by the controller.

In Fig. 4.24, the experimental results of the inductor current spectra are shown, for N = 4 and N = 8, both with the cascade of alpf and dlpf. The inductor current data is obtained using a current probe with 300 kHz bandwidth and the oscilloscope RIGOL MS05104. The data is acquired with a rate of 500 MS/s, and the window length is equal to 50 ms. For N = 4, the current reference is set to 3.25 A and the corresponding results are shown in Fig. 4.24a. When both filters are activated, the jittering is present simultaneously for positive and negative slopes of the carrier. From the presented results, it is clear that the jittering causes a very high detrimental impact on the inductor current, which is successfully cancelled using the proposed algorithm. Similar conclusions are found for N = 8 in Fig. 4.24b, however, with a lower impact of jittering, which is expected as the discontinuities are decreased with the increase of N.

An important feature of the anti-jittering algorithm is to detect large-signal disturbances and prevent the blocking of the modulating waveform update. Fig. 4.25 is given to demonstrate that the algorithm allows detection of large-signal disturbances and offers an unimpaired response to a reference step change. The experimental verification is performed for the case of N = 4 with the cascade of dlpf and alpf, corresponding to results in Fig. 4.24a. The values of m are exported from the DSP memory. This example demonstrates the effectiveness of the proposed adaptive limits, described in Section 4.5.3.3, which are used to determine whether the modulating waveform update should be allowed or not.

Finally, the duty cycle variances, for all tested regimes, are calculated and compared with the analytical values from (4.20). For calculations, the differences between critical modulating segments are found for $D = D_c$, as shown in Fig. 4.22. For N = 4 with the dlpf, for D = 0.48, the calculated and the measured duty cycle variances are equal to $1.12 \cdot 10^{-5}$ and $1.38 \cdot 10^{-5}$, respectively. Without the jittering, the measured duty cycle variance is 10 times lower. For N = 4 with the cascade of dlpf and alpf, for



Figure 4.23: Simulation results of the reference current step responses for N = 6: (a) without the proposed anti-jittering algorithm; (b) with the proposed anti-jittering algorithm; (c) relative reference tracking errors.

D = 0.5, the calculated and the measured duty cycle variances are equal to $2 \cdot 10^{-4}$ and $2.78 \cdot 10^{-4}$, respectively. It should be noted that for this case, the jittering occurs simultaneously for both slopes of w; hence the statistical prediction is obtained using both $\Delta m_{c,d}$ and $\Delta m_{c,u}$. Without the jittering,



Figure 4.24: Experimental results of the inductor current spectra for: (a) N = 4 and $D \approx 0.5$; (b) N = 8 and $D \approx 0.25$.



Figure 4.25: Experimental demonstration of the anti-jittering algorithm operation for the reference current step change from 3.25 A to 4 A, for the case of N = 4 with alpf and dlpf. Blocking of the modulating waveform update is prevented after a large disturbance is detected.

the measured duty cycle variance is decreased more than 100 times. For N = 6 with the cascade of dlpf and alpf, for D = 0.325, the calculated and the measured duty cycle variances are equal to $1.68 \cdot 10^{-5}$ and $2.8 \cdot 10^{-5}$, respectively. Without the jittering, the measured variance is reduced approximately 17 times. For N = 8 with the cascade of dlpf and alpf, for D = 0.246, the calculated and the measured duty cycle variances are equal to $6.25 \cdot 10^{-6}$ and $1.06 \cdot 10^{-5}$, respectively. Without the jittering, the measured variance is reduced approximately 20 times. From these results, it is clear that the statistical model from (4.20) offers a decent prediction of the intensity of duty cycle jittering. A certain amount of mismatch is expected, as the resulting LCOs are a stochastic phenomenon with a pattern that is not completely predictable.

From the presented results, it is clear that the jittering may result in a very high impact on the converter's output. The results also clearly demonstrate that the proposed anti-jittering algorithm is capable of canceling the jittering phenomenon.

4.5.4 MS-PWM modulating waveform and its discontinuities

Let us now continue with the analysis of all discontinuity-related nonlinearities. In this section, the procedure for calculating m in MS-PWM control systems is explained. Additionally, the impact of N, the control loop bandwidth, and the time delay on the modulating waveform discontinuities is analyzed.

4.5.4.1 Calculating the modulator transcharacteristic for MS-PWM

As explained in Section 4.4.2, the calculation of the modulator transcharacteristic corresponds to finding the steady-state waveform m(t) that results in a specific value of D. For the case of NS-PWM, this procedure is relatively simple due to the fact that, for a given converter topology, the shape of the modulating waveform ripple $m^{ac}(t)$ is completely defined by the value of D. For a fixed value of D, any variation of the turn-on instant is equivalent to a time translation of $m^{ac}(t)$ [27].

On the contrary, for MS-PWM, the modulating waveform ripple cannot be specified within a time translation. For a specific value of D, in case the turn-on instant is changed, the output waveform ripple is simply translated, however, its sampled version features a shape change. Of course, this directly impacts the shape of m(t) as well. Therefore, to obtain the MS-PWM transcharacteristic, an iterative numeric solver is needed, with the necessary steps reported in [27].

The analysis is simplified for control systems with a proportional controller and without digital feedback filters. For those, the sampler can be moved after the controller without impacting the shape of m [27]. This allows us to find the "analog equivalent" of the modulating waveform ripple, labeled m_{NS}^{ac} , based only on the value of D, just like for NS-PWM. Based on m_{NS}^{ac} and D, the entire "equivalent analog" modulating waveform m_{NS} can be found just like in Section 4.4.2. Then, an iterative procedure from [27] is used to determine the turn on instant that corresponds to the chosen sampling instants and the value of D, resulting in the complete sampled-and-held waveform m.

For the following results, the modulating waveforms and their discontinuties are calculated assuming

the proportional controller (without the one step computation delay) and without the feedback filters, i.e. as in Fig. 4.7a, only for the digital control. This configuration is chosen in order to show results of a general value, that are scalable with the relative crossover frequency $f_{c,r}$. The following methodology can, however, be extended to an arbitrary linear control system. Moreover, the calculations are performed for the set-up B.1, described in Table 3.2; however, as explained in Section 4.2.2, the results do not change for different converter parameters as long as the system with a proportional controller is analyzed with respect to the relative crossover frequency $f_{c,r}$. As noted before, the modulating waveform analysis for a proportional controller is expected to bring good results also for other high-frequency proportionaldominant controller structures, which is verified in the following sections.

In this chapter, it was mentioned several times that, for MS-PWM with very high oversampling factors, the operation resembles the analog NS-PWM. To corroborate this claim, MS-PWM transcharacteristics are calculated for the proportional control system with N = 64, $f_{c,r} = 0.1$, and 4 values of the time delay. Results are shown in Fig. 4.26 along with the ones obtained for NS-PWM (also shown in Fig. 4.9). The transcharacteristics overlap almost perfectly, which confirms the assumed behavior of highly oversampled MS-PWM systems. Note that, for the case of MS-PWM with N = 64, there are many critical operating points where the vertical crossings may occur (see (4.16)); however, their impact is not highly evident in the transcharacteristics, as the discontinuities of m are greatly reduced for high values of N. For this reason, it is clear that the analysis of the effects caused by the discontinuities can be limited to low values of N.



Figure 4.26: Calculated MS-PWM transcharacteristics for N = 64 and 4 values of the time delay τ_D . Results are compared to those obtained for the analog NS-PWM, which are plotted sparsely using only markers for a better visualization.

4.5.4.2 Discontinuity graphs

As noted in Section 4.5.3.1, around each D_c , the in-phase discontinuity of the critical modulating segments determines the height of the jitter zone in an MS-PWM transcharacteristic. Also for the counter-phase vertical crossings, it is clear that the discontinuities are correlated with the extension of the reducedand the zero-gain zones. Therefore, it is of interest to find a way to easily determine whether the control system design enables the in-phase or the counter-phase operation, and to predict the size of the critical modulating segments' discontinuities.

Let us here propose an approximated procedure for calculating m in case of the proportional-dominant controllers, which does not require any iteration. This procedure relies on the approximation that m is obtained by simply sampling the "equivalent analog" m_{NS} at the chosen instants, e.g. those in (2.2). An illustration of the resulting waveforms is shown in Fig. 4.27. It is clear that, after m_{NS} is sampled in this way, the actual switching instants are slightly shifted, which translates the initially assumed m_{NS} as well; hence, an error is introduced. It is obvious that this error is increased for a higher switching ripple and lower values of N. Nevertheless, in this section it will be shown that this approximated method provides a good prediction of the critical modulating segments' discontinuties.



Figure 4.27: Illustration of the approximated procedure for calculating m, based on NS-PWM.

In this section, the discontinuity graphs are provided to illustrate the change of $\Delta m_{c,u,d}$, for various control loop bandwidths, as a function of the time delay. Dependency on the time delay comes from the fact that the types of intersections between m and w are determined by the shape of the m, which changes as the switching ripple component is shifted with respect to the carrier⁷.

As an illustration of the time delay dependency, Fig. 4.28 shows how shifting the "equivalent analog" modulating waveform affects the discontinuity of the sampled-and-held m (using the approximation from

⁷Changing the time delay is not the only way to impact the intersections between the modulating waveform and the carrier. This can be also achieved by using filters with high-frequency properties suitable to re-shape the waveform m with respect to w, as shown in Section 4.5.3.4. However, the time delay was found to be a useful parameter to examine the discontinuities, without needing to resort to a specific filter design.

Fig. 4.27). It can be seen that the discontinuity for the positive slope of w is completely removed for $\tau_{D,2}$. As explained above, this sampling approximation does not completely match the system's behavior; however, it is clear that time-shifting the ripple component with respect to w will impact the shape of m.



Figure 4.28: Illustration of the time delay impact on the sampled modulating waveform.

Due to the periodicity of the switching ripple, the analysis is limited to the range $\tau_D \in [0, 1)$. Results are given for $N \in \{4, 6, 8\}$ as the discontinuity-related effects are suppressed with the increase of the oversampling factor.

The following discontinuity graphs show $\Delta m_{c,u,d}$ for each examined pair of (N, D_c) . The critical modulating segments' discontinuities are calculated as in (4.17), by finding the waveform m for $D = D_c$. As previously noted, the vertical crossings appear also in the vicinity of these duty cycles, however, it is of interest to see how this simple procedure based on a single operating point can be used to predict the properties of MS-PWM transcharacteristics. As stated in Section 4.5.4.1, for the exact calculation of m, an iterative procedure is needed. In this section m is also found using the proposed approximated procedure, corresponding to Fig. 4.27. The approximated procedure is interesting as it reveals the discontinuities by simply solving a set of equations, i.e. without any iteration.

Note again that, due to the anti-symmetry of the current ripple in buck converters, the discontinuity graphs for $D = D_c$ for the positive slope of w are the same as the discontinuity graphs for $D = 1 - D_c$ for the negative slope of w. That is why, for $D_c = \frac{1}{2}$, the discontinuity graphs for the positive and negative slopes are always equal.

The results for N = 4 and $D_c = \frac{1}{2}$ are shown in Fig. 4.29. In this figure, a comparison is given for different values of $f_{c,r}$. It can be seen that the values of $\Delta m_{c,u,d}$ are indeed proportional to the relative crossover frequency. The graphs show a range of time delays for which the size of the discontinuities varies linearly, and a range for which it is constant. For the case of $f_{c,r} = \frac{1}{6}$, a comparison is made with the exact iterative calculation of m. The resulting values in the flat region are perfectly matched, while a small mismatch is seen elsewhere. For the figure clarity, other discontinuity graphs do not feature a comparison with the exact procedure; however, it is verified that the mismatch remains practically of the same entity. An important thing to note is the existence of two values of τ_D , almost independent on the relative crossover frequency, for which the operation turns from the counter-phase to the in-phase. For N = 4 and $D_c = \frac{1}{2}$, these values of τ_D are unique for both slopes of w, which is important as it indicates that the existence of such τ_D flattens out both pairs of the critical modulating segments, thus preventing any vertical crossings from occurring. This value of τ_D is tightly correlated with the low-pass filter design that brings the operation to a border between the in-phase and the counter-phase, seen in Fig. 4.21 in Section 4.5.3.4.



Figure 4.29: Discontinuity graphs for N = 4 and $D_c = 0.5$. Modulating waveforms are calculated using the approximated procedure for several values of $f_{c,r}$. For $f_{c,r} = \frac{1}{6}$, the result is compared with the one obtained using the exact procedure for the calculation of m.

For the figure clarity, results for other values of (N, D_c) are given only for $f_{c,r} = \frac{1}{10}$. It is verified that with the change of $f_{c,r}$, conclusions remain the same as for N = 4; the values of discontinuities are scaled, while the borders between the in-phase and the counter-phase operation remain located at nearly constant values of τ_D .

In Fig. 4.30, discontinuity graphs are given for N = 6, $D_c = \frac{1}{3}$. The results are not shown for N = 6, $D_c = \frac{2}{3}$ as the two discontinuity graphs are anti-symmetric (what is $\Delta m_{c,u}$ for one would be $\Delta m_{c,d}$ for the other, and vice-versa). It can be seen that the sum of the maximum discontinuities for the positive and negative slopes of w is scaled by a factor $\frac{4}{6}$ compared to N = 4, which supports the fact that the increase of N suppresses the discontinuities.



Figure 4.30: Discontinuity graphs for N = 6 and $D_c = 0.33$. The graphs are obtained using the approximated procedure for $f_{c,r} = \frac{1}{10}$.

In Fig. 4.31, the discontinuity graphs are given for N = 8, $D_c = \frac{1}{4}$, yielding similar conclusions as for N = 6, $D_c = \frac{1}{3}$. In Fig. 4.32, the discontinuity graph, valid for both slopes of the carrier, is given for N = 8, $D_c = \frac{1}{2}$. As for N = 4, $D_c = \frac{1}{2}$, there are two values of the time delay τ_D that result in a borderline operation for both slopes of w.



Figure 4.31: Discontinuity graphs for N = 8 and $D_c = 0.25$. The graphs are obtained using the approximated procedure for $f_{c,r} = \frac{1}{10}$.



Figure 4.32: Discontinuity graphs for N = 8 and $D_c = 0.5$. The graphs are obtained using the approximated procedure for $f_{c,r} = \frac{1}{10}$.

An important thing to notice is that for critical duty cycles other than $D_c = \frac{1}{2}$, there is no unique time delay that results in the borderline operation between the in-phase and the counter-phase for both slopes of w. This means that, unlike for $D_c = \frac{1}{2}$, there is no time delay that can flatten out both pairs of the critical modulating segments.

To verify the applicability of the discontinuity graphs, which are calculated only for $D = D_c$, on predicting the MS-PWM operation around the analyzed operating point, three examples of modulator transcharacteristics are given in Fig. 4.33, for the case of N = 4, $D_c = \frac{1}{2}$, and $f_{c,r} = \frac{1}{10}$. The transcharacteristics are calculated using the exact procedure described in Section 4.5.4.1. The time delays are chosen to emphasize the counter-phase regime ($\tau_D = 0.1$), the in-phase regime ($\tau_D = 0.5$), as well as the borderline between the two ($\tau_D = 0.347$).

As predicted by the discontinuity graph in Fig. 4.29, the MS-PWM transcharacteristic for N = 4 can be completely linearized by imposing an appropriate time delay. Let us here make a connection between this result and the one from Section 4.5.3.4. Namely, by subtracting the one step computation delay that is assumed in Section 4.5.3.4, the borderline for N = 4 is here obtained for $\tau_D = 0.347 - \frac{1}{N} = 0.097$. The phase lag of the remaining transport delay, at the switching frequency, is approximately equal to -35° , which is close to the simulated borderline in Fig. 4.20. This hints that the analysis based on the time delay may also be used to directly predict the impact of feedback low-pass filters, which adds to the generality of the presented results.

Regarding the in-phase regime, seen in Fig. 4.33 for $\tau_d = 0.5$, the vertical part of the transcharacteristic matches with the value obtained using (4.18) and the discontinuity graph shown in Fig. 4.29. For the case shown, the two jitter zones corresponding to the positive and negative slopes of w are merged together. Note that the match between the jitter zone heights and the values predicted by the discontinuity graphs is confirmed for other values of (N, D_c) as well.

To conclude, the discontinuity graphs represent valuable means of predicting the MS-PWM behavior



Figure 4.33: Demonstration of the impact of τ_D on the MS-PWM nonlinearities around $D_c = \frac{1}{2}$ for N = 4 and $f_{c,r} = \frac{1}{10}$. The transcharacteristics are given for the counter-phase operation (squares), which features the reduced- and zero-gain zones, the in-phase operation (stars), which features the jitter zone, and the borderline case (diamonds), which results in the linear transcharacteristic.

due to the following facts:

- they are easily calculated by solving a set of equations, without any iteration, for a single operating point;
- they can be used to find a borderline between the counter-phase and the in-phase operation, which remains nearly constant with the change of the relative crossover frequency $f_{c,r}$;
- they can be used to quantify the height of the jitter zone in case of the in-phase operation.

4.5.5 Analysis of MS-PWM transcharacteristics

This section analyzes the appearance and quantifies the extensions of the nonlinear zones in MS-PWM transcharacteristics. All transcharacteristics are obtained using the exact procedure described in Section 4.5.4.1. As for the discontinuity graphs, the following results are given as functions of the time delay.

4.5.5.1 Measures of MS-PWM nonlinearity

Let the MS-PWM transcharacteristic be labeled as $G = D(\langle m \rangle)$. The nonlinearity measures of G used in this section are:

• Root mean square (RMS) measure calculated as RMS(G - L(G)), where L(G) represents the linear regression of G, obtained using the least-square error criterion. This standard measure of nonlinearity is valuable as it tells us how close is G to its nearest possible straight line. However, it does not discriminate the different types of nonlinearities, which is important due to their unique impacts on a converter's output.

• Modified differential nonlinearity (M-DNL) measure. This measure discriminates the different nonlinearity zones in an MS-PWM transcharacteristic, and shows their relative extensions. The standard differential nonlinearity measure, i.e. $DNL = max\{|\frac{dD}{d < m >} -1|\}$, is not useful as the crucial information on the relative extension of each zone is not provided. As explained before, and seen in Fig. 4.12, there are only 3 possible types of nonlinearity zones. M-DNL shows which one is present and to what extent. For the analyzed transcharacteristic G, different zones are detected, and their total relative extensions are plotted: $\Delta < m >_{\frac{1}{2}}$ is the total span of < m > for which G features a slope close to $\frac{1}{2}$ (the reduced-gain zone); $\Delta < m >_0$ is the total span of < m > for which G is vertical (the infinite-gain zone).

In order to separately analyze each critical duty cycle D_c , the RMS and M-DNL measures are calculated for a certain ΔD around each D_c , where ΔD is chosen such that the related nonlinearity zones are fully covered.

4.5.5.2 Nonlinearity graphs

For each analyzed N, the transcharacteristics are calculated for $\tau_D \in [0, 1]$ with 1% resolution and the duty cycle resolution of 0.1%. The results are given for the proportional controller and $f_{c,r} = \frac{1}{10}$.

In Fig. 4.34, M-DNL and RMS measures are given for N = 4 around $D_c = \frac{1}{2}$. From Fig. 4.34a, it can be seen that the infinite-gain zones occur for approximately $\tau_D \in (0.35, 0.89)$. There, the values of ΔD_{∞} are in excellent match with the values obtained using (4.18) and the discontinuity graph seen in Fig. 4.29. Note that M-DNL does not discriminate whether the jitter zones for the positive and negative slopes of w are merged into a single jitter zone, but only shows their joined height. It is also visible that the zero-gain zone has the highest extent for $\tau_D \approx 0.13$. There are regions around $\tau_D = 0.3$ and $\tau_D = 0.95$ where the reduced-gain (corresponding to a single counter-phase vertical crossing) is the only nonlinearity. For dc-dc converters, these regions are favorable as there are no uncontrollable operating points caused by the zero-gain zones and no LCOs caused by the infinite-gain zones. From Fig. 4.34b, it is clear that the border between the counter-phase and the in-phase operation ($\tau_D \approx 0.35$ and $\tau_D \approx 0.89$) results in the lowest RMS measure. Due to the finite resolution of τ_D , the plot only gets to show RMS values close, but not exactly equal to 0.

In Fig. 4.35, M-DNL and RMS measures are given for N = 6 around $D_c = \frac{1}{3}$. Note that, due to the anti-symmetry of the triangular switching ripple, these results are also valid around $D_c = \frac{2}{3}$. From Fig. 4.35b, it can be seen that the local minima are found near the values of τ_D where, for one slope of the carrier, there is a transition between the counter-phase and in-phase regimes. The minimization of the modulator nonlinearity near these borderlines is intuitive, as it corresponds to the flattening of one critical modulating segment pair. These borderline values of τ_D can be also read from the discontinuity graph in Fig. 4.30.

In Fig. 4.36, M-DNL and RMS measures are given for N = 8 around $D_c = \frac{1}{4}$ that are, thanks to the



Figure 4.34: Nonlinearity measures for N = 4, $D \in [0.35, 0.65]$, and $f_{c,r} = \frac{1}{10}$: (a) M-DNL, (b) RMS measure.

anti-symmetry, also valid for $D_c = \frac{3}{4}$. The conclusions are the same as for N = 6, $D_c = \frac{1}{3}$. In Fig. 4.37, M-DNL and RMS measures are given for N = 8 around $D_c = \frac{1}{2}$. The conclusions are the same as for N = 4, $D_c = \frac{1}{2}$. There are two values of the time delay, $\tau_D \approx 0.39$ and $\tau_D \approx 0.95$, for which the transcharacteristic can be completely linearized around that value of D_c . Again, these are not precisely obtained here due to the resolution of τ_D used for calculations.

From the presented results, it is clear that the increase of N results in lower nonlinearity measures. It should be noted that the nonlinearity measures are also calculated for several other values of $f_{c,r}$, and the results follow the same trends as those reported above. The reduced-gain, zero-gain, and infinite-gain zones are present in nearly the same range of τ_D , while their extensions are scaled proportionally to $f_{c,r}$.

The dependency of the nonlinearity measures on the time delay allows for an easy prediction of the MS-PWM behavior for existing delays in the system, e.g. for a verification if the dead-bands or the jitter-zones will appear. Furthermore, the results can be used to minimize the nonlinearities by imposing



Figure 4.35: Nonlinearity measures for N = 6, $D \in [0.18, 0.48]$, and $f_{c,r} = \frac{1}{10}$: (a) M-DNL, (b) RMS measure.

an additional time delay. This is legitimate if structural delays of the system are below the desired value, also considering the resulting impact on the dynamic response. Note that it may not be optimal to add the time delay as a compensation measure. Instead, digital filters may be used, such that the desired effect on the shape of m is achieved, but with a more favorable impact on the dynamic performance. One example of such design is the borderline digital low-pass filter from Section 4.5.3.4.

To illustrate each of the distinct regions seen in the M-DNL graph in Fig. 4.35a, transcharacteristics corresponding to specific values of τ_D are plotted in Fig. 4.38.

4.5.6 Experimental verifications

The impact of the MS-PWM nonlinearities on a power converter's operation depends on the entire closed-loop control system design. As noted in Section 4.5.5, higher controller gains result in higher



Figure 4.36: Nonlinearity measures for N = 8, $D \in [0.15, 0.35]$, and $f_{c,r} = \frac{1}{10}$: (a) M-DNL, (b) RMS measure.

nonlinearity measures, however, also in a stronger disturbance rejection, which may mitigate the final impact on the converter's output. Therefore, this section offers an experimental examination of the MS-PWM nonlinearities in typical application scenarios. For dc-dc converters, the impacts on the transient response and the occurrence of LCOs are investigated. For ac-type converters, it is examined how the MS-PWM nonlinearities impact the output waveform distortion.

For the following experimental tests, the used hardware set-ups are B.1 and B.2, described in Table 3.2, with the inductance value L = 1.5 mH. The FPGA control platform is described in Section 3.2. For postprocessing, data are acquired with 25 MS/s rate, using the Tektronix MS056 oscilloscope. The inductor current is sensed using the Tektronix TCP202 current probe. For the total harmonic distortion (THD) calculations, 100 ms of data are acquired.

The FPGA platform reads the ADC output with a rate of 40 MHz. This allows to set a value of τ_D with a resolution of 0.05%, using a delay line. Structural delays in the set-up include 700 ns due to the



Figure 4.37: Nonlinearity measures for N = 8, $D \in [0.4, 0.6]$, and $f_{c,r} = \frac{1}{10}$: (a) M-DNL, (b) RMS measure.

algorithm computation and 1.5 μ s due to hardware delays from the current sensing to the transistor gate voltage. The modulating waveform is updated as soon as the control action is calculated, i.e. without the one step computation delay. The current sampling instants are re-scheduled to compensate the control execution time. In this way, the modulating waveform update instants correspond to those in (2.2). The transcharacteristics are experimentally measured by sweeping the current reference while storing the values of m and the detected D inside the control platform. For visualization, the experimentally obtained transcharacteristics are filtered by averaging results from 10 consecutive switching periods.

4.5.6.1 Impact of MS-PWM nonlinearities in dc-dc converters

To verify the impact of the MS-PWM nonlinearities on a dc-dc power converter's operation, the halfbridge set-up B.1, shown in Fig. 3.1a is used. The implemented controller is a PI from (4.3), discretized using the backward Euler method, and the crossover frequency is chosen as $f_c = \frac{1}{10} f_{pwm}$. The propor-



Figure 4.38: Examples of transcharacteristics around D = 0.33, for N = 6 and 4 values of τ_D . Each sub-figure corresponds to a different combination of the nonlinearity zones, as seen in Fig. 4.35a.

tional gain is calculated as $k_p = 2\pi f_c \frac{L}{V_{in}} = 0.048 \frac{1}{\text{A}}$. The integral gain is set to $k_i = 151 \frac{1}{\text{As}}$, to avoid an additional crossover before the *LC* resonance. Note again that the resulting MS-PWM control system is directly analyzed with respect to the previously shown nonlinearity graphs, which are calculated for the purely proportional controller. The impact of the added integral action at the frequencies of the switching harmonics is low; hence, a good match is expected.

For dc-dc converters, it is instructive to verify the jittering operation, which appears when the operating point enters the infinite-gain zone of the MS-PWM trancharacteristics. This operation is problematic because it results in LCOs, which cannot be suppressed by the feedback. The difference compared to the results shown in Section 4.5.3.5 is that, here, the time delay is used instead of the low-pass filters. The control loop configuration with N = 4 and $\tau_D = 0.5$ is chosen. From Fig. 4.34a, it can be seen that this time delay results in $\Delta D_{\infty} = 7.82\%$. As a benchmark, the operation is also tested for $\tau_D = 0.3$, which brings the presence of the reduced-gain zones only. The experimental verification of the transcharacteristics belonging to these settings are shown in Fig. 4.39a. It can be seen that an almost perfect match is obtained, which confirms the validity of using the P-controller analysis for the PI-based control loops. The experimentally obtained points of the infinite-gain zone are plotted without averaging. Note again that in this zone the converter does not operate in steady-state; hence, the transcharacteristic is not well defined. The jittering operation is tested by imposing a step reference change, such that the duty cycle moves from the linear part, $D(i_{L,r1}) = 0.42$, to the infinite-gain part, $D(i_{L,r2}) = 0.5$. The tracking errors, relative to the dc value of the current, are shown in Fig. 4.39b. For visualization, the switching ripple is removed using a MAF over the switching period. Starting from t = 2 ms, the LCOs are evident for the case of $\tau_D = 0.5$.

It is of interest to see whether the discontinuity and nonlinearity graphs can be used to predict the impact of LCOs. Given that both jitter zones in Fig. 4.39a are merged together, the variance is calculated based on (4.20) as $\sigma_D^2 = \frac{\Delta D_{\infty}^2}{4}$. For the corresponding jitter zone height, the predicted value of the duty cycle variance is $1.53 \cdot 10^{-3}$. The experimentally obtained variance is equal to $1.7 \cdot 10^{-3}$, which is in a good match with the prediction. This value can also be estimated from the discontinuity graph in Fig. 4.29.



Figure 4.39: Operation of the tested converter for N = 4 and two values of τ_D : (a) experimental verification (dashed line with markers) of the calculated transcharacteristics corresponding to $\tau_D = 0.5$ (circles) and to $\tau_D = 0.3$ (stars); (b) corresponding relative current errors for step responses from $D(i_{L,r1}) = 0.42$ to $D(i_{L,r2}) = 0.5$. The switching ripple is filtered out for a better visualization.

As mentioned in Section 4.5.2.2, impacts of the reduced- and zero-gain zones are difficult to analyze in general terms. The behavior of the converter depends on the size of the disturbance, the controller gains, the parameters of the converter, etc. Qualitatively, as long as the operation is locked in a reduced-gain

zone, the loop dynamics are halved. The zero-gain zone impact is particularly difficult to predict. For a large enough disturbance, a zero-gain zone may not have any impact at all, as the operating point may simply jump over it. If the modulating waveform is being perturbed across a dead-band, the converter practically exhibits an open-loop behavior, which may result in a significantly impaired response, far from the small-signal prediction. Nevertheless, the nonlinearity graphs can be used to find a suitable value of the time delay that minimizes the extension of the reduced- and zero-gain zones, which would consequently result in a smaller impact on the converter's operation.

4.5.6.2 Impact of MS-PWM nonlinearities in ac-type converters

For the verification of the nonlinearities' impact on ac-type power converters, the full-bridge set-up B.2, shown in Fig. 3.1b, is used. The operating parameters are summarized in Table 4.1. For the ac-dc tests, the PR controller from (4.4) is discretized using the impulse-invariant method [121]:

$$G_c(z) = k_p + k_r T_s \frac{1 - \cos(\omega_1 T_s) z^{-1}}{1 - 2\cos(\omega_1 T_s) z^{-1} + z^{-2}}.$$
(4.31)

The proportional gain is calculated based on the desired crossover frequency as $k_p = 2\pi f_c \frac{L}{2V_{in}}$, while the resonant gain k_r is calculated as $\frac{k_r}{k_p} = \frac{1}{10} 2\pi f_c$.

For ac-type converters, it is of interest to investigate the nonlinearities' impact on the output waveform distortion. As noted in Sections 4.5.4.2 and 4.5.5.2, the extension of the nonlinearity zones is proportional to the crossover frequency; however, the disturbance rejection also depends on it. Therefore, it is not possible to directly conclude that the increase of the nonlinearity measures, by increasing the bandwidth, would result in a worse quality of the output waveform.

The first set of results is obtained for N = 4. As the dead time itself is a significant source of distortion in ac-type converters [3], the impact of the MS-PWM nonlinearities cannot be observed on its own, except in simulations with the dead time set to zero. Therefore, DS-PWM (N = 2) is used as a benchmark. The designed control loops for N = 2 and N = 4 feature similar disturbance rejection properties; hence, the differences between these two values of N can be ascribed to the sources of distortion specific to MS-PWM.

The first goal of this section is to show that the analyzed nonlinearities are the dominant sources of distortion in MS-PWM controlled converters. This is tested by measuring the current THD for $f_{c,r} \in \{\frac{1}{14}, \frac{1}{10}, \frac{1}{6}\}$ and two sets of transcharacteristics related to N = 4. The first set is obtained by imposing a time delay $\tau_D = 0.1$, which results in the presence of the reduced- and zero-gain zones. The experimental verification of the corresponding transcharacteristics is shown in Fig. 4.40. Again, note that an almost perfect match is obtained between the experiments with the PR controller and the computations for the purely proportional one.

The second set is obtained by adding time delays $\tau_D = \{0.354, 0.347, 0.332\}$, respectively to the above listed values of $f_{c,r}$. These delays result in linear transcharacteristics, such as the one in Fig. 4.33. The inductor currents and their spectra are plotted in Fig. 4.41, for $f_{c,r} = \frac{1}{10}$. For time-domain

Description	label	value	unit
Output frequency	f_1	50	Hz
Output voltage - RMS	v_{a}^{RMS}	230	V
Output current - RMS	i_I^{RMS}	4.9	A
Dead time	/	500	ns
Proportional gains	k_p	$\{0.017, 0.024, 0.04\}$	$\frac{1}{A}$
Resonant gains	k_r	$\{15.4, 30.2, 83.9\}$	$\frac{1}{As}$
Relative crossover frequencies	$f_{c,r}$	$\left\{\frac{1}{14}, \frac{1}{10}, \frac{1}{6}\right\}$	/
N = 2 - benchmark case	label	value	unit
Added delays	$ au_D$	0	/
Total harmonic distortion	THD	$-\{38.1, 39.8, 42.6\}$	dB
N = 4 - nonlinear transch.	label	value	unit
Added delays	$ au_D$	$\{0.1, 0.1, 0.1\}$	/
Total harmonic distortion	THD	$-\{24.9, 24, 24.1\}$	dB
N = 4 - linearized transch.	label	value	unit
Added delays	$ au_D$	$\{0.354, 0.347, 0.332\}$	/
Total harmonic distortion	THD	$-\{38.5, 39.5, 41.2\}$	dB

Table 4.1: Summary of the dc-ac converter results



Figure 4.40: Experimental verification (dashed line with markers) of calculated transcharacteristics for $\tau_D = 0.1$ and $f_{c,r} = \{\frac{1}{14}, \frac{1}{10}, \frac{1}{6}\}$. The increase of the relative crossover frequency increases the extension of the nonlinearity zones.

visualization, the inductor current is averaged over T_{pwm} using a MAF. These results are compared to the benchmark case of the DS-PWM control, which does not feature any of the MS-PWM nonlinearities. It can be seen that for N = 4 and $\tau_D = 0.1$, the current is strongly distorted around the zero-crossings and the presence of strong odd harmonics is seen in the spectrum. When the time delay $\tau_D = 0.347$ is applied, the current distortion is significantly improved and the difference compared to the DS-PWM is practically not visible. Also for other values of $f_{c,r}$, this trend is confirmed by the THD measurements, performed up to the 40th harmonic, which are shown in Table 4.1. The THDs are reduced by 15 dB by adding the appropriate time delays.

The fact that these values are very close to those for N = 2 confirms that the remaining distortion is caused by the dead time. Differently from what was concluded in [34], the presented analysis and experiments suggest that the aliasing effect is not originating the measured distortion. Quite on the contrary, this appears to be a second order effect, negligible in size with respect to effects of the dead time and the nonlinear transcharacteristic. The aliasing-based distortion was noticed only in simulations with the linearized transcharacteristic and without any dead time. An important difference between the distortion caused by the aliasing and the modulator nonlinearities is that the aliasing-based distortion cannot be suppressed by the feedback; hence, it may be critical in some applications.



Figure 4.41: Example of the inductor current distortion for the tested full-bridge inverter. The comparison is made for $f_{c,r} = \frac{1}{10}$ between DS-PWM and MS-PWM with N = 4 for $\tau_D = 0.1$ and $\tau_D = 0.347$: (a) time-domain inductor current waveforms with the switching ripple removed for visualization; (b) spectral content in the frequency window used for THD calculations.

The second goal of this section is to show that higher nonlinearity measures do not necessarily result

in a higher distortion, if the control bandwidth is increased as well. This is tested by comparing the THD results for N = 4, $\tau_D = 0.1$, and $f_{c,r} \in \{\frac{1}{14}, \frac{1}{10}, \frac{1}{6}\}$, which corresponds to transcharacteristics in Fig. 4.40. The relative extensions of the reduced- and the zero-gain zones can be read from the nonlinearity graphs. For all values of $f_{c,r}$, the relative extensions are calculated and their values are $\Delta < m >_{\frac{1}{2}} \in \{2.13, 3.18, 5.42\}\%$ and $\Delta < m >_0 \in \{4.68, 6.47, 10.66\}\%$, respectively to the tested $f_{c,r}$. It can be seen that the extensions are proportionally scaled with $f_{c,r}$. The measured THD values are equal to $\{-24.9, -24, -24.1\}$ dB, respectively to the tested $f_{c,r}$. Although the extensions of the nonlinearity zones differ significantly, the resulting values of THD are similar, which confirms that the feedback is capable of suppressing the related distortion.

For values of N other than 4, the transcharacteristic cannot be completely linearized by flattening out the critical modulating segments; however, the resulting distortion can be minimized. As the nonlinearity graphs are given locally, around each D_c , they do not provide means for a global optimization, i.e. if the operating point moves across all values of D_c . In case the full-range optimization is required, the nonlinearity measures can be calculated for $D \in [0, 1]$. Another option is to use the presented nonlinearity graphs to find a value of the time delay that avoids a specific nonlinearity, e.g. the dead-band or the jitter-zone, in the entire operating range. Regarding the jitter zones, they were found to be highly unsuitable for ac-type converters as well because they introduce additional high-frequency distortion due to LCOs being manifested.



Figure 4.42: Inductor current THD for $f_{c,r} = \frac{1}{10}$ and control systems that feature the one step computation delay. The RMS value of the inductor current is equal to 4.9 A and the RMS value of the output voltage is equal to 230 V. The fundamental frequency is equal to 50 Hz.

As a final illustration, the impact of increasing the oversampling factor on the THD is shown in Fig. 4.42. The ac-dc operation perturbs the duty cycle in the range from 0.1 to 0.9. The results are shown for control systems with the one step computation delay, i.e. $\tau_D = \frac{1}{N}$. The crossover frequency is set to $f_{c,r} = \frac{1}{10}$ and no other actions for reducing the nonlinearities were taken. The double-update control clearly results in a lowest output distortion, setting the benchmark for MS-PWM to target. It can be seen that the distortion level is high for N = 4 and N = 8. The THD can be effectively reduced using

the methods proposed in this chapter and, at least for N = 4, even brought down to practically the same level as for N = 2. Without taking any actions, an effective suppression of the distortion caused by the MS-PWM nonlinearities starts to appear at N > 8, owing to the reduction of the discontinuities.

4.6 Summary

This chapter has analyzed the nonlinear MS-PWM properties that arise when m contains a significant switching ripple content. The main focus was placed on the set of nonlinearities caused by the jump discontinuities of m, which are specific to MS-PWM. First, the discontinuity-related nonlinearities are classified and the operating conditions that lead to them are explained in a general way. Besides the reduced- and zero-gain zones, which are already analyzed in [27], one of the original aspects of this thesis is the discovery of the third type of nonlinearities, the infinite-gain zones. These nonlinearities, which are shown to result in LCOs, are qualitatively explained, their impact on the duty cycle variance is statistically modelled, and an algorithm is proposed to cancel them out. The second part of this chapter brings the methodology for determining which nonlinearity will be manifested and to what extent, depending on the control system parameters. Moreover, provisions for minimizing their impacts are proposed and validated on the experimental prototypes.

Finally, let us note that the digital control systems that rely on multiple feedback samples per switching period are considered natural for series or parallel stacked multi-level or interleaved converters with phase-shifted PWM [55]. This is because, in balanced conditions, due to the increase of the ripple frequency, more than two instants per switching cycle appear where sampling the feedback signal yields its average value [51]. This means that the multi-sampled control can be implemented, ideally, without any ripple in m. However, in unbalanced conditions or during transients, the discontinuity-related nonlinearities may appear, a problem which is being investigated at the time of writing.

Chapter 5

Analysis of noise propagation in multi-sampled control systems

5.1 Introduction

An often mentioned advantage of adopting the single- or double-sampled PWM approach is that the sampling itself acts as a finite impulse response (FIR) filter, yielding the average feedback value. However, it is often hard to achieve a perfect synchronization, which is seen to introduce severe errors for certain operating modes [3, 63]. For example, this can cause aliasing, which is mentioned in Section 4.3. Furthermore, relying on only one or two samples per switching period can bring significant noise sensitivity [64].

To reduce the impact of acquisition errors, many applications rely on oversampling the feedback and then filtering it in the digital domain, e.g. using a moving average filter over one switching period [63, 64, 110, 111, 122, 123]. The filtered signal is then decimated to the control update rate, which is typically chosen as single- or double-update. These multi-rate strategies are introduced in Section 2.2.2 and labeled as MR-(S/D)S-PWM. For high oversampling factors, MR-(S/D)S-PWM methods approach the error-free acquisition (in terms of the actual average value) and may offer strong noise attenuation [63, 64]. These strategies can also be relatively easily implemented in commercially available DSPs, by exploiting the direct memory access (DMA) module [63]. However, strong filtering followed by the decimation to (S/D)S-PWM greatly deteriorates the dynamic capabilities of the system. These methods rely on oversampling purely for filtering purposes, with a strong penalty on the dynamic performance.

In MS-PWM control systems, the feedback is by definition oversampled, although to the purpose of improving the dynamic response. Correspondingly, the question arises of whether these systems also bring an inherent noise suppression. The motivation behind this chapter is to investigate mechanisms of noise propagation in power converters, with a specific focus on multi-sampled systems. Attention is dedicated to analyzing whether MS-PWM can reduce the impact of feedback noise, while maintaining the dynamic benefits.

This chapter starts by recalling a standard model for noise propagation, valid for linear systems, in order to examine its validity in digitally-controlled power converters. It is shows that this simple model predicts well the system behaviour only in the case of (S/D)S-PWM. For MS-PWM, it is demonstrated how the DPWM introduces a nonlinear effect, by re-sampling the modulating waveform. This results in decimation, which causes aliasing and represents the main limiting factor for white noise attenuation in multi-sampled pulsewidth modulated power converters. The decimation effect is qualitatively explained and a design procedure is proposed for digital filters used to suppress the aliasing with as little impact on the dynamic performance. With the appropriate filter design, the decimation effects are reduced and noise propagation properties approach those of linear systems.

Additionally, in this chapter, analytical models are derived for noise propagation in power electronic control systems that feature decimation. A particularly challenging case covered is the one where the decimation occurs inside a closed-loop system. The resulting model is applicable for both MS-PWM (inherent decimation due to DPWM) and MR-(S/D)S-PWM (intentionally added decimation after oversampled feedback filtering). Therefore, the results of this chapter can be used to estimate the noise sensitivity or to design filters to achieve a satisfactory performance in terms of the output noise content, for both MS-PWM and MR-(S/D)S-PWM.

Analytical derivations are followed by extensive experimental verifications that affirm the assumed noise signal properties in the experimental set-up, validate the derived noise propagation models, and confirm that, in terms of noise propagation, DPWM behavior closely resembles a re-sampler. Experimental results are given for a buck-type converter's single-stage current loop with a PI controller and a single-stage voltage loop with a PID controller. It is shown that, for PI controllers, strong noise suppression can be achieved by including digital filters that introduce a small impact on the dynamic performance. On the other hand, controllers with a derivative action bring a more emphasized impact of DPWM decimation; hence, stronger anti-aliasing filtering is needed for high noise attenuation.

In the end, this chapter features a comparison in terms of noise propagation and dynamic properties between MS-PWM with the proposed anti-aliasing digital filters and more commonly adopted strategies for oversampled filtering, i.e. MR-(S/D)S-PWM. It is shown that, for the same oversampling factor, noise suppression capabilities are practically the same (slightly better for MS-PWM), while MS-PWM results in drastically better dynamic capabilities.

It is important to note that this chapter mainly investigates and models the propagation of the uncorrelated, white noise, which is why the feedback signal should not be dominantly affected by, for example, the transistor switching noise. Most results found in this chapter are published in [67, 68].

5.2 Analyzed multi-sampled control system

An example of a dc-dc power converter, with MS-PWM control stage, is shown in Fig. 5.1. Results in this chapter are given for converters with dc operating points; however, it will be shown in Chapter 6



Figure 5.1: Multi-sampled control system of a dc-dc power converter with emphasis on its multi-rate structure. It is assumed that the converter features an output LC filter. The signals v_c and i_L correspond to the output capacitor voltage and the inductor current, respectively.



Figure 5.2: Simplified block diagram of the control system from Fig. 5.1.

that the presented analysis is directly applicable to ac-type converters as well. The analyzed converter operates using single-stage control loop configurations, where the output y(t) (e.g. the inductor current $i_L(t)$ or the output capacitor voltage $v_c(t)$, in case of an *LC* output filter) is directly controlled by modifying the duty cycle. Single-stage loops are used in order to examine different properties of noise propagation related to the specific controller design.

The presented control system is of a multi-rate nature and can be separated in 3 sections, based on their operating frequency. As explained in Section 2.1, the ADC is used to sample the continuous time output variable, transforming it into the digital domain. After the sampling process, the ADC output is summed with $n_q(k)$, which models the quantization noise due to finite ADC resolution [3]. Note again that the control and sampling frequencies are not necessarily the same, e.g. in case of MR-(S/D)S-PWM where the decimation is imposed after the applied filtering. The DPWM block transforms its digital input m_s into the analog output x. As discussed in Section 2.1, the modulating impulse train $m_s[k]$ is held constant over one sampling period, which results in the modulating waveform m(t). The triangular DPWM adjusts its output based on two values of m(t) per modulation period, which is why the DPWM domain is labeled with $2f_{pwm}$. The re-sampling nature of DPWM is hinted in Section 2.2.3 and is the focal point of this chapter. The power stage and sensing circuits belong to the continuous time domain. The converter's output y(t) is summed with the measurement noise $n_m(t)$ resulting from sensing circuits.

A simplified block-diagram of the multi-sampled dc-dc converter, with only one noise signal $n_m(t)$, is shown in Fig. 5.2. It is assumed that gains of the sensing circuits are compensated within the controller and that their filtering action is negligible in the frequency range of interest; hence, they are replaced with unity gains. As explained in Section 2.2.2, the function of the holder is inherent to the DPWM, because that is the block where the conversion from the digital to analog domain takes place [3]. The small-signal DPWM model is given in (2.3).

5.2.1 Converter under test

For the experimental results shown in this chapter, the half-bridge hardware set-up B.1, with parameters from Table 3.2, is used. The inductance value is L = 1.2 mH. Without loss of generality, the same topology is used for the following analysis. The illustration of the half-bridge buck converter, with voltage and current measurements is shown in Fig. 5.3.



Figure 5.3: Schematic of the buck converter with the direct current or voltage control, used in this chapter.

Referring to Fig. 5.2, the plant transfer function from d to y, G_p , is obtained using the averaging technique [99]. For the current loop, it is equal to:

$$G_{p,i}(s) = \frac{i_L(s)}{d(s)} = \frac{V_{in}}{R} \frac{sRC+1}{s^2 LC + s\frac{L}{R} + 1}.$$
(5.1)

For the voltage loop, it is equal to:

$$G_{p,v}(s) = \frac{v_c(s)}{d(s)} = V_{in} \frac{1}{s^2 L C + s\frac{L}{R} + 1}.$$
(5.2)

5.2.2 Controller design

For most results in this chapter, the control systems feature one step computation delay. The crossover frequency f_c of the control loops is chosen to be near $0.1 f_{pwm}$.

For the single-stage current loop, the zero in the plant transfer function (5.1) allows the crossover frequency to be set above the LC resonance using the PI controller structure [3]. The current controller
is labeled $G_{c,i}$:

$$G_{c,i}(z) = \left(k_{p,i} + k_{i,i}T_s \frac{1}{1 - z^{-1}}\right) z^{-1}.$$
(5.3)

For the single-stage voltage loop, above the LC resonance the plant transfer function (5.2) exhibits a -180° phase response. Therefore, in order to set such high crossover frequencies, a derivative action must be added to the controller; hence, the PID structure is chosen. Typically, to reduce its impact on noise, this kind of controller also features a low-pass filter in cascade with the derivative gain, which results in:

$$G_{c,v}(z) = \left(k_{p,v} + k_{i,v}T_s \frac{1}{1 - z^{-1}} + \frac{k_d}{T_s}(1 - z^{-1})G_{df}(z)\right)z^{-1}.$$
(5.4)

The derivative gain filter should provide attenuation at high frequencies, while not significantly reducing the phase margin. As an example, $G_{df}(z)$ is designed as a first order low-pass filter with a cut-off frequency equal to $\frac{f_{pwm}}{2}$, and is discretized using the bilinear transform. For simplicity, the entire structure $G_{c,v}$ will be referred to as the PID voltage controller. For a satisfactory phase margin near $0.1f_{pwm}$, the voltage loop is analysed only for $N \geq 2$.

The parameters of the controllers used in this chapter are summarized in Table 5.1.

Current loop	label	value	unit
Controller configuration	$G_{c,i}$	PI	/
Relative proportional gain	$\frac{V_{in}}{R}k_{p,i}$	0.2344	/
Relative integral gain	$\frac{V_{in}}{R}k_{i,i}$	585	$\frac{1}{s}$
Crossover frequency	$f_{c,i}$	2000	Hz
Tested oversampling factors	N	$\left[1, 2, 4, 8, 16, 32\right]$	/
Voltage loop	label	value	unit
Controller configuration	$G_{c,v}$	PID	/
Relative proportional gain	$V_{in}k_{p,v}$	1.3294	/
Relative Integral gain	$V_{in}k_{i,v}$	709	$\frac{1}{s}$
Relative Derivative gain	$V_{in}k_{d,v}$	$1.4\cdot 10^{-4}$	s
Cut-off frequency of $G_{df}(z)$	$f_{c,d}$	10	kHz
Crossover frequency	$f_{c,v}$	1850	Hz
Tested oversampling factors	N	[2, 4, 8, 16, 32]	/

 Table 5.1: Controller parameters

5.3 Noise properties and linear propagation modeling

5.3.1 Sources of noise and feedback acquisition errors

Some of the most relevant sources of noise and feedback acquisition errors, present in digitally controlled power converter systems, can be classified as:

- Synchronous sampling errors. As explained in Section 4.3, for the inductor current control with a nearly-constant output voltage and assuming no parasitic resistances, the average current value is sampled at the mid-points of the applied switched node voltage pulse (i.e. the center-pulse sampling). Often, designers rely on synchronizing the sampling instants with the peaks and valleys of the triangular carrier to obtain the average current value. However, due to many factors, such as the asymmetry of x with respect to w, parasitic resistances, actuation delays, dead-times, limited sensor bandwidths, and analog filters, the sampled value may significantly differ from the average current [3, 124]. Moreover, for voltage control loops, the average output voltage position within a switching period is not fixed for all operating points [117]. This type of an acquisition error can cause a significant impact on the system and is, therefore, one of the motivations for oversampling the feedback signal and averaging it over the switching period [63]. It should be mentioned that, at least for the current control, these errors are often reduced simply by delaying the sampling instants with respect to the ones in (2.2).
- Switching noise, resulting from high dv/dt and di/dt during commutations of the converter. Very steep edges of PWM voltage waveforms, together with the presence of various parasitic *LC* elements (due to winding parasitic capacitances, long cables in industrial drives, etc.), give rise to poorly damped oscillations in the output waveforms [63, 125, 126]. These effects again motivate the use of the oversampled averaging, for reducing the switching noise impact [63]. In multi-sampled PWM control, for certain operating points, sampling the switching noise cannot be avoided. This can cause an undesired response of the controller to noise corrupted samples of the feedback signal. The switching noise is strongly dependent on the hardware design and PCB layout, and the undesired effects depend also on the oversampling factor, the duty cycle, as well as the filter and controller design. Moreover, this type of noise is strongly correlated with the feedback signal, making it hard to treat analytically. The conclusions drawn in this chapter are valid for hardware systems that do not feature excessive propagation of switching noise.
- Quantization noise, which results from a finite resolution of ADC and DPWM modules [82–84]. As discussed in Section 2.2.1, in modern control platforms, the DPWM clock rates are in the order of hundreds of MHz and the effects of DPWM quantization have a small impact on the system if the switching frequencies are not extremely high. The ADC quantization noise depends on the full-scale feedback signal range and the number of ADC bits. In closed-loop systems, considering the quantization noise to be uncorrelated with the feedback is a strong assumption

[3,84], which prevents a simple analysis. For the following derivations, it is assumed that the ADC resolution is high enough such that the quantization noise is not dominant in the analyzed power converter system. This assumption often holds [64] and is validated in subsequent experimental measurements, using a 12-bit ADC.

• Measurement noise $n_m(t)$, which is caused by sensing and conditioning electronic circuits, as well as other sources of the electromagnetic interference (EMI) present at the converter's location [127, 128]. For the analytical considerations presented in this chapter, measurement noise is assumed to be a wide-sense stationary (WSS) Gaussian white noise [127], additive to the feedback signal. An example of noise with similar properties is the random thermal noise [127].

In closed-loop systems, such as the one in Fig. 5.1, measurement errors propagate to the output by feedback, negatively impacting the converter's operation. This motivates the analysis of noise propagation, in order to predict its overall impact on the output and, accordingly, design filters to suppress it below an acceptable level. In the subsequent analysis, it is assumed that the dominant source of noise is the *measurement noise*. This allows the use of relatively simple signal processing techniques due to its additive characteristics, being uncorrelated to the feedback signal. Extensive experimental validations support the assumed noise properties for the tested prototype.

5.3.2 Analysis of white noise propagation

For its simple and intuitive use in power electronic systems, the analysis in this chapter is performed in the frequency domain, avoiding as much as possible the strict use of the probability theory. For simplicity of the presentation, the frequency domain signals and transfer functions will be followed by parenthesis (ω) , regardless of whether they belong to discrete or continuous time domains.



Figure 5.4: Illustration of linear processing of a discrete-time stochastic signal n. In the specific example, it is assumed that n is a white noise signal with a flat PSD S_n .

Let us start the analysis by studying the effect of processing a random discrete-time signal n by a linear time-invariant (LTI) discrete-time system with a transfer function $H_n(z)$, where z is the complex variable of the Z transform corresponding to the sampling frequency f_s . The LTI system of interest is shown in Fig. 5.4 and its output is labeled as y. The signal n is assumed to be a band-limited, WSS, and Gaussian white noise [127]. It will be pointed out throughout this chapter where and why this assumption is strictly necessary. Magnitudes of frequency components of a stochastic signal n can be

estimated from its power spectral density (PSD) $S_n(\omega)$, which is assumed to be stationary. However, information on its phase is not available; hence, it cannot be completely represented in the frequency domain. For this reason, its propagation through an LTI system cannot be analyzed directly, but only using its PSD. Let us start by finding the PSD of y, $S_y(\omega)$. Its well-known relation with $S_n(\omega)$ can be found by multiplying $y(\omega)$ with its complex conjugate¹ [127]:

$$S_y(\omega) = \frac{1}{f_s} y(\omega) y^*(\omega) = \frac{1}{f_s} \left(n(\omega) H_n(\omega) \right) \left(n(\omega) H_n(\omega) \right)^* = \frac{1}{f_s} n(\omega) n^*(\omega) |H_n(\omega)|^2 = S_n(\omega) |H_n(\omega)|^2.$$
(5.5)

With the information on the output PSD, the following formula can be used to calculate the in-band noise power (variance) of the output variable y, σ_y^2 [127]:

$$\sigma_y^2 = \int_0^{2\pi f_x} S_n(\omega) |H_n(\omega)|^2 d\omega, \qquad (5.6)$$

where f_x defines the band of interest. Let us define the input noise power as $\sigma_n^2 = \int_0^{2\pi \frac{f_s}{2}} S_n(\omega) d\omega$. In case the input noise is white, S_n is constant (its PSD is flat across the Nyquist frequency range and $\sigma_n^2 = \frac{f_s}{2}S_n$); hence, it can be factored before the integral in (5.6). This can be used to define the Noise Attenuation Coefficient (NAC) for the white noise input as:

$$NAC = \frac{\sigma_y^2}{\sigma_n^2} = \frac{2}{f_s} \int_0^{2\pi f_x} |H_n(\omega)|^2 d\omega.$$
(5.7)

Going back to the analyzed power converter system in Fig. 5.2, as n_m is uncorrelated to the feedback, the sampler f_s can be applied to it independently, resulting in a discrete-time signal n entering the closedloop system. In order to find the magnitude response $|H_n(\omega)|$, the system from Fig. 5.2 is transformed into the frequency domain:

$$H_n(\omega) = \frac{y(\omega)}{n(\omega)} = -\frac{W_{ol}(\omega)}{1 + W_{ol}(\omega)} = -\frac{G_{fb}(\omega)G_c(\omega)G_{dpwm}(\omega)G_p(\omega)}{1 + G_{fb}(\omega)G_c(\omega)G_{dpwm}(\omega)G_p(\omega)},$$
(5.8)

where $W_{ol}(\omega)$ is the open-loop transfer function of the system shown in Fig. 5.2. The discrete-time blocks are transformed to the frequency domain using $z = e^{j\omega T_s}$. To consider the impact of the sampler in Fig. 5.2, the frequency response of the cascade of G_p and G_{dpwm} is obtained by first transforming it to the Z-domain, using the impulse-invariant discretization with frequency f_s [4].

PSD of the sampled, band-limited, white noise signal is spread across the Nyquist frequency window; hence, it is reduced as the sampling frequency is increased. This spreading is determined by the factor $\frac{2}{f_s}$ in (5.7). The integrand in (5.7) determines the filtering impact of the system, depending on its magnitude

¹Strictly speaking, multiplication of a frequency domain signal with its complex conjugate results in its squared frequency components' magnitudes, which are related to the energy spectral density (ESD) [127]. However, this relaxed and somewhat mathematically inaccurate definition will be used here to avoid adding complexity by introducing limit values and expectation operators. Additionally, the factor $\frac{1}{2}$ is needed to obtain squared RMS values from squared magnitudes. The considered PSD is, therefore, found by dividing the squared RMS values of frequency components by the total one-sided frequency range of interest $\frac{f_s}{2}$. In case of discrete-time measurements with a finite number of samples, PSD is obtained by dividing with spectral resolution instead.

response in the frequency range of interest. An intuitive way of decreasing the noise power, even without oversampling, is to design control loops with lower bandwidths, providing a higher filtering action by $|H_n(\omega)|$. These two mechanisms jointly impact the white noise propagation in linear multi-sampled systems.

The Nyquist frequency of the multi-sampled systems surpasses the switching frequency of the converter. As the small-signal plant model $G_p(s)$ is obtained using the averaging technique [99], it is only valid up to a fraction of the switching frequency. Additionally, at higher frequencies, the switching ripple is dominant in the output spectrum, which is why it is hard to experimentally measure the noise floor without incorporating its impact. Hence, in order to correctly estimate the output variance, the frequency window in (5.7) must be limited below f_{pwm} .

It is important to notice that the analysis based on (5.7) is strictly valid for linear systems only, which is not the case for switched power converters. However, it is of interest to examine how well, and under what conditions, can (5.7) be used to predict noise propagation in power electronic converters. In Section 5.5 the analysis is extended to cover the main nonlinearity, later to be shown as caused by the carrier in MS-PWM.

For subsequent verifications, the controller parameters in Table 5.1 are chosen according to the lowest tested N, after which they are kept constant. In this way, the crossover frequency remains the same, while the phase margin is increased for higher oversampling factors. This allows the examination of noise propagation for different multisampling factors in relative terms, without significantly altering the MS-PWM nonlinearities, which are analyzed in Chapter 4. Furthermore, by having significant differences in the phase margins for lower values of N, the results are expected to emphasize the impact of filtering brought by the closed-loop system $|H_n(\omega)|$. For higher values of N, phase margin differences are very small; hence, the only expected impact on noise propagation is, instead, the white noise spreading across the Nyquist frequency window.

Magnitude responses of the closed-loop transfer functions from the feedback noise to the output, $|H_n(\omega)|$, for the analyzed current and voltage control loops, are shown in Fig. 5.5.

5.3.3 Noise measurement organization

The organization of the system used for noise propagation measurements is illustrated in Fig. 5.6. The converter is run in closed-loop configuration, with an imposed constant reference value. After the steady state is reached, the output is measured and its PSD, determined by the feedback noise propagation, is estimated.

In simulations, the noise signal is sampled independently from the feedback to be able to directly find the frequency response from n to y. For white noise, this corresponds to obtaining the frequency response $H_n(\omega)$. In this case, both n and y are acquired for postprocessing with $f_{acq} = f_s$. For later-shown experimental tests, this approach is not feasible, as the input noise cannot be sampled independently from the feedback. Therefore, only the output y is acquired using an oscilloscope. The output is probed



Figure 5.5: Magnitude responses of the closed-loop transfer functions for: (a) direct current control $H_{n,i}$ and (b) direct voltage control $H_{n,v}$. Hardware and control parameters are shown in Table 3.2 and Table 5.1.

before being conditioned for sampling, in order to observe the impact of the feedback noise on the physical output itself, i.e. i_L or v_c . Since only the output is measured, the analytical procedure can be used to predict its variance based on knowing the input noise power or, more simply, by finding the output noise power relative to one of the measured cases. Both approaches are used in this thesis. The oscilloscope acquisition rate f_{acq} is set significantly higher than the probe bandwidth, to prevent aliasing, and the window length T_{data} is determined to obtain a satisfactory frequency resolution (equal to $\frac{1}{T_{data}}$), such that spectral leakage from dominant switching harmonics is not high. Specific oscilloscope settings are explained in Section 5.6. The resulting output noise power is calculated from the acquired signals, in the frequency domain, by summing the squared RMS component values in the frequency region of interest.



Figure 5.6: Block diagram of the noise measurement set-up. For simulations, the noise signal n_m is injected using a random number generator. It is sampled independently of (but synchronously with) the feedback, to be able to also observe the discrete signal n. For experiments, there is a unique sampler after the summing block and only the output signal y is acquired via an oscilloscope.

5.4 Impact of DPWM decimation

5.4.1 Aliasing caused by DPWM decimation

The analytical model in (5.7) is strictly valid only for LTI systems. Power converters, although being nonlinear due to their switched nature, are well-modelled below $0.5 f_{pwm}$, using small-signal averaging techniques [3,99]. Therefore, it is expected that (5.7) can offer a good prediction when the bandwidth of interest is limited below $0.5 f_{pwm}$, where the modulation process is practically linear [3].

However, let us observe the case of a control system with MS-PWM, where an interesting phenomenon takes place. Consider the system shown in Fig. 5.2, where both the controller and the feedback filter are replaced by purely proportional actions. If the feedback signal, which is summed with the band-limited white noise, is oversampled (N > 2), the noise floor is pushed down compared to DS-PWM and the PSD of the digital signal m_s is reduced. Nevertheless, the output of the triangular modulator is still determined by only two intersections between m and w. Hence, just two samples of m_s are chosen, i.e re-sampled, by the modulator in order to produce its output x. As the considered control system is purely proportional, this corresponds to using exactly two feedback samples. Therefore, resulting noise propagation cannot differ from the case of a system that would sample the feedback only twice per switching period. An illustration of this mechanism is shown in Fig. 5.7. It is clear that, in order to ensure that x is determined based on multiple feedback samples taken inside one modulation period, some filtering action needs to be included in G_{fb} or G_c , making each individual value of m_s dependent on previous feedback samples as well. As it will be shown, this comes down to providing an anti-aliasing function using digital filters, before re-sampling occurs from f_s to $2f_{pwm}$.

The mechanism discussed above points out that the impact of the triangular DPWM on noise propagation may be estimated by modeling it as a decimator of m_s with the decimation factor $M = \frac{f_s}{2f_{pwm}}$.



Figure 5.7: Illustration of the triangular DPWM re-sampling its input signal twice per switching period.

The assumed nonlinear impact of the DPWM is examined in simulations, for the current-controlled configuration of the converter shown in Fig. 5.3. Noise propagation is tested for SS-PWM, DS-PWM, and MS-PWM with N = 4. As noted in Section 5.3.3, calculating the ratio of frequency components' magnitudes of i_L and n corresponds to finding $|H_n(\omega)|$, assuming a linear system behavior. The expected magnitude responses correspond to $|H_{n,i}(\omega)|$ shown in Fig. 5.5a. Note again that this kind of test cannot be performed experimentally, as noise cannot be sampled independently of the feedback.

The results of these simulations are shown in Fig. 5.8. As can be seen, for N = 1 and N = 2, the simulated magnitude response is in perfect match with the small-signal prediction. This confirms that the linear analysis in (5.7) is well-suited for predicting noise propagation in systems with (S/D)S-PWM. However, for N = 4, when the sampled feedback contains noise components at $f > f_{pwm}$, the simulated magnitude response shows a significant mismatch from the small-signal model. This indicates that converters with MS-PWM exhibit a highly-nonlinear behavior in terms of noise propagation. At this point, based on the mechanisms described above, it is presumed that the reason behind this is that the aliasing of noise spectral components at $f > f_{pwm}$ occurs due to the decimation process introduced by the triangular DPWM. The assumed decimation behavior is supported by experimental tests in Section 5.6, where an excellent match is obtained with the extended analysis from Section 5.5.

The frequency zones of interest are summarized in Fig. 5.9, by comparing a double-sampled system with an N-sampled system. It is assumed that the control system is purely proportional and the impact of the feedback is neglected for the illustration. By oversampling the feedback, white noise spectral power is spread up to $\frac{1}{2}Nf_{pwm}$. This results in the reduction of the PSD compared to the double-sampled case, $s_N = \frac{2}{N}s_2$. The frequency range, in which noise propagation is analyzed, $f < f_x$, is marked as A. The zone B marks the region in which the spectral components are folded back due to the DPWM decimation. Without any digital filtering, aliasing due to the DPWM decimation limits the achievable PSD to the value determined by the double-sampled case. Regarding noise propagation, this can be considered equivalent to the implementation of MR-DS-PWM with the controller output being decimated to N = 2, without any prior filtering. In order to achieve the PSD equal to $\frac{s}{N}$ in zone A, spectral components in zone B must be completely filtered-out in the digital domain, prior to entering the DPWM. Hence, calculations using (5.7) are expected to provide valid predictions of noise propagation



Figure 5.8: Comparison between the simulated magnitude response from n to i_L and $|H_{n,i}|$, for the converter described in Table 3.2 and the PI controller described in Table 5.1. The converter runs at the operating point D = 0.4. The acquired data length is equal to 200 ms and the spectrum is averaged over 100 Hz for better visualization. The exact simulation settings are described in Section 5.6.1.



Figure 5.9: Illustration of the oversampling impact on the white noise PSD for the triangular PWM carrier and a purely proportional control system, neglecting the impact of the feedback. Zone A represents the frequency window of interest for noise measurements. Zone B represents the frequency window that must be filtered in digital domain to suppress the DPWM decimation effect.

for MS-PWM only if the entire spectral content in zone B is removed.

In order to reduce the spectral fold-back and make noise propagation properties closer to the ones of (S/D)S-PWM, some digital filtering is required. It is of interest to design these filters such that their impact is low in the control bandwidth, in order to retain a good dynamic performance. Filtering should ideally be limited only to $f > f_{pwm}$. In this way, the spectral aliasing would be reduced and the noise suppressed at frequencies well-below f_{pwm} , where the filter does not have a direct impact. The triangular carrier is already proven to be the most suitable for MS-PWM, with respect to reducing the impact of the discontinuity-related nonlinearities and bringing a constant modulation delay [27]. This section points to another possible advantage. Namely, in case of single-edge modulators, only one sample of m_s per modulation period is used to determine the switching signal; hence, even for N = 2 the decimation occurs. Therefore, anti-aliasing digital filtering should be performed in the region $f > \frac{f_{pwm}}{2}$, which would imply a significant deterioration of the system dynamics.

5.4.2 Qualitative approach to anti-aliasing digital filter design

In this section, digital feedback filters are designed so as to suppress the DPWM decimation effect. The design goal is to provide attenuation of m_s at $f > f_{pwm}$ (zone B in Fig. 5.9), with the least impact on degrading the dynamic performance.

Let us start by analyzing the response from the sampled noise n to the DPWM's input signal m_s , for the system in Fig. 5.2. The corresponding transfer function is:

$$H_{nm}(\omega) = \frac{m_s(\omega)}{n(\omega)} = -\frac{G_{fb}(\omega)G_c(\omega)}{1 + W_{ol}(\omega)}.$$
(5.9)

Consider the frequency window in which the DPWM-related aliasing takes place $(f > f_{pwm})$. It is clear that, at such high frequencies, the closed-loop is ineffective, i.e. $|W_{ol}(\omega)| \ll 1$. This allows to simplify the analysis by considering the open-loop structure:

$$H_{nm}(\omega) \approx -G_{fb}(\omega)G_c(\omega). \tag{5.10}$$

From (5.10), it is clear that $G_{fb}(\omega)$ or $G_c(\omega)$ should be designed to attenuate components at $f > f_{pwm}$.

For the current loop with the PI controller, without a feedback filter, the high-frequency gain of $H_{nm_i}(\omega)$ is practically determined by the proportional action. Therefore, a similar decimation impact is expected as that illustrated in Fig. 5.9, i.e. the PSD remaining constant with the increase of N above 2.

For the voltage loop with the PID controller, without a feedback filter, the impact of the decimation is more severe. Due to the derivative action, the gain of $H_{nm_v}(\omega)$ increases with the frequency, with the maximum value being determined by the Nyquist frequency. For this reason, higher values of N increase the total noise power above f_{pwm} and, consequently, also the in-band PSD after the aliasing takes place. For this reason, it is expected that a higher attenuation at $f > f_{pwm}$ is required for control loops that employ the PID, with respect to those with the high-frequency proportional-dominant controllers, e.g. P, PI, or PR.

Considerations above are summarized in Fig. 5.10, by showing results of approximated calculations for the total noise power that gets folded back due to the aliasing. Values are given for the current and voltage controllers from (5.3) and (5.4), without feedback filters. The same assumption as in (5.10) is made, i.e. that $|H_{nm}(\omega)| \approx |G_c(\omega)|$ at $f > f_{pwm}$. The traces in Fig. 5.10 show the numerically integrated squared magnitude responses of H_{nm} in the range $[f_{pwm}, \frac{f_s}{2}]$, normalized by the same frequency range as well as by the results for N = 4. This practically corresponds to calculating (5.7) except that the frequency range up to f_{pwm} is neglected so that $|H_{nm}(\omega)| \approx |G_c(\omega)|$ remains valid. From the presented results, it is clear that, for the PI controller, the total noise power that participates in the aliasing practically does not change as the sampling frequency is increased. This result comes from the nearly-



Figure 5.10: Impact of the designed current PI and voltage PID (with the filtered derivative) controllers on the total noise power above f_{pwm} , which participates in the aliasing caused by the DPWM decimation. Controller parameters are equal to those in Table 5.1. The results are given relative to N = 4.

constant high-frequency gain of $G_{c,i}$. On the other hand, for the PID controller, the noise power that gets folded back is strongly amplified as N is increased. It should be again noted that the voltage controller from (5.4) contains also a low-pass filter for the derivative action. Without including such filter, the noise power amplification is significantly higher.

In the following sections, based on the conclusions above, for the PI current controller the suppression of the DPWM decimation effect is tested by introducing a single first-order digital low-pass filter (DLPF) with the cut-off frequency equal to the switching frequency. This filter, labeled as $G_{DLPF}(z)$, is implemented as:

$$G_{DLPF}(z) = a \frac{z+1}{z+b},$$
 (5.11)

where $a = \frac{\pi}{\pi + N}$ and $b = \frac{\pi - N}{\pi + N}$. DLPF from (5.11) results in a small impact on the system dynamics as it introduces approximately 6° phase lag at the set crossover frequency $f_{c,i} = 0.1 f_{pwm}$. The frequency responses of $H_{nm,i}$, with and without the DLPF, are shown in Fig. 5.11a, given for N = 8. For the voltage loop with the PID controller, noise propagation is examined by including the $G_{DLPF}(z)$ from (5.11), and also a third-order filter, equivalent to the cascade of three $G_{DLPF}(z)$, labeled $G_{DLPF}^3(z)$. The filter $G_{DLPF}^3(z)$ introduces approximately 16° phase lag at the crossover frequency of the voltage loop, which results in a non-negligible impact on the dynamic response. The corresponding frequency responses of $H_{nm,v}(\omega)$ are shown in Fig. 5.11b, given for N = 8. In the following sections, the aliasing suppression capability of the MAF from (2.9) is tested as well.

Note that, for all following results, digital feedback filters are only used for oversampling factors N > 2 as their sole purpose is the suppression of the aliasing effect due to the DPWM decimation, and not the direct attenuation of the spectral content up to f_x .



Figure 5.11: Frequency responses of cascades of the controller $G_c(z)$ and the feedback filter $G_{fb}(z)$ for (a) the current loop with the PI controller and (b) the voltage loop with the PID controller. The controller parameters are shown in Table 5.1 and the plots are given for N = 8.

5.5 Extended models that include decimation effects

The goal of this section is to derive a better-suited model of noise propagation for systems that feature decimation. The motivating example is the DPWM decimation, which occurs inside the MS-PWM closed-loop system. Another strategy that is covered by this analysis is the case of oversampled filtering followed by the decimation to SS-PWM or DS-PWM (MR-(S/D)S-PWM) [63]. The section starts with the analysis of open-loop systems, which is a simpler case and is used as a building block for the closed-loop analysis. A relevant open-loop scenario is found in an application discussed in Chapter 6.

For all calculations, the noise signal is assumed to be a WSS and Gaussian process. For the final equivalent models, it is also assumed to be band-limited white noise.

5.5.1**Open-loop** systems

Model derivation 5.5.1.1

$$f_{s,1} \qquad f_{s,2} = \frac{f_{s,1}}{M}$$

$$\xrightarrow{\mathbf{n}} H_{eq}(\omega) \xrightarrow{\mathbf{Z}} M \xrightarrow{\mathbf{y}} S_{z}(\omega)$$

Figure 5.12: Open-loop oversampled filtering, followed by an M-time down-sampler.

Consider an open loop system shown in Fig. 5.12. The input noise signal n, sampled with frequency $f_{s,1}$, is processed by H_{eq} , after which it is decimated to $f_{s,2} = \frac{f_{s,1}}{M}$. In order to quantify noise propagation, it is of interest to find the relation between the PSD of n, $S_n(\omega)$, and the PSD of y, $S_y(\omega)$. Then, the output noise variance, in the frequency range of interest, can be easily calculated by integrating $S_y(\omega)$. Relation between S_z and S_n is found as in (5.5):

$$S_z(\omega) = S_n(\omega) |H_{eq}(\omega)|^2.$$
(5.12)

Let us now consider the impact of the decimator, which down-samples the signal z M times to obtain the signal y. From signal processing theory, it is known that the impact of an integer-rate down-sampler on its input signal can be calculated as^2 :

$$y(\omega) = \sum_{k=0}^{M-1} z(\omega - k2\pi f_{s,2}).$$
(5.13)

The expression (5.13) represents the spectral folding of the signal $z(\omega)$ and can be used to find the impact of aliasing if $z(\omega)$ contains components above the baseband of the decimator, i.e. $\frac{f_{s,2}}{2}$. However, it cannot be applied directly as complete information on the magnitude and the phase of a stochastic signal $n(\omega)$, and therefore $z(\omega)$ as well, is not available. Let us, for this reason, find the expression³ for

²Note that, when considering the Discrete Fourier Transform (DFT), an additional scaling factor $\frac{1}{M}$ is needed to address the reduced number of samples at the output. However, for all computations shown in this chapter, the scaling is incorporated in the DFT algorithm; hence, it is left out of the following expressions. ³What may confuse is that, in (5.14), the first factor should be determined by the sampling frequency of the domain where the signal y is, i.e. $\frac{1}{f_{s,2}}$. However, since the fold-backs of z and z^* both bring a factor of $\frac{1}{M}$, what is left is $\frac{1}{f_{s,2}} \frac{1}{M^2}$.

Then, the expression can be manipulated to obtain $\frac{1}{f_{s,1}} \frac{f_{s,1}}{f_{s,2}} \frac{1}{M^2} = \frac{1}{f_{s,1}} M \frac{1}{M^2} = \frac{1}{f_{s,1}} \frac{1}{M}$. The remaining scaling factor $\frac{1}{M}$ is again incorporated in the DFT calculation and is, hence, left out.

the PSD of y:

$$S_{y}(\omega) = \frac{1}{f_{s,1}} y(\omega) y^{*}(\omega) = \frac{1}{f_{s,1}} \left(\sum_{k=0}^{M-1} z(\omega - k2\pi f_{s,2}) \right) \left(\sum_{l=0}^{M-1} z(\omega - l2\pi f_{s,2}) \right)^{*} =$$

$$= \frac{1}{f_{s,1}} \sum_{\{k,l\}=0}^{M-1} z(\omega - k2\pi f_{s,2}) z^{*}(\omega - l2\pi f_{s,2}) =$$

$$= \frac{1}{f_{s,1}} \sum_{k=0}^{M-1} z(\omega - k2\pi f_{s,2}) z^{*}(\omega - k2\pi f_{s,2}) + \frac{1}{f_{s,1}} \sum_{\{k,l\}=0, k\neq l}^{M-1} z(\omega - k2\pi f_{s,2}) z^{*}(\omega - l2\pi f_{s,2}) =$$

$$= \sum_{k=0}^{M-1} S_{z}(\omega - k2\pi f_{s,2}) + \frac{1}{f_{s,1}} \sum_{\substack{\{k,l\}=0, k\neq l\\ =0, if z is a stationary stochastic signal.}}^{M-1} z(\omega - l2\pi f_{s,2}) .$$

$$(5.14)$$

It can be seen that the PSD of y can be found using the sum of two contributions. The first one represents the same spectral fold-back as in (5.13) applied directly to the PSD of z, while the second one represents a sum of products between different frequency bands of z. As noted above, it is assumed that nis a sampled WSS and Gaussian process, which also makes it strictly stationary [127]. Linear processing of WSS and Gaussian signals retains these properties, hence, signal z is also strictly stationary [127]. As shown in [129, 130], stationary stochastic signals exhibit no spectral correlation, hence, the second contribution in (5.14) is identically equal to 0, and the PSD of y can be found as:

$$S_y(\omega) = \sum_{k=0}^{M-1} S_z(\omega - k2\pi f_{s,2}) = \sum_{k=0}^{M-1} S_n(\omega - k2\pi f_{s,2}) |H_{eq}(\omega - k2\pi f_{s,2})|^2.$$
(5.15)

This result is very important, and not trivial, as it allows the fold-back expression to be applied to the squared magnitudes of stationary signal's frequency components, in the same way as it would be applied to the signal itself. In this way, only the information on magnitudes (or the PSD) of n is necessary to find the PSD of the down-sampler's output. Additionally, it should be noted that the decimated signal y remains stationary [131]. Finally, as the input signal n is assumed to be band-limited white noise, i.e. its one-sided power spectrum is constant across the bandwidth and equal to $S_n(\omega) = S_n = \frac{2}{f_{s,1}}\sigma_n^2$ [127], the PSD of n can be factored outside of the sum in (5.15), which yields the following expression:

$$S_y(\omega) = S_n \sum_{k=0}^{M-1} |H_{eq}(\omega - k2\pi f_{s,2})|^2 = \frac{2}{f_{s,1}} \sigma_n^2 |H_{eq}^{M:1}(\omega)|^2.$$
(5.16)

The steps above allow us to analyze noise propagation in the system shown in Fig. 5.12 using its equivalent representation in Fig. 5.13. It is important to note that this equivalent squared magnitude response $|H_{eq}^{M,1}(\omega)|^2$ is only valid for the analysis of the WSS Gaussian white noise and should not be misunderstood as being applicable to any input signal. For a stationary input noise with the frequency-dependent PSD, (5.15) must be used instead. Finally, if the noise is not stationary, both terms from the last row of (5.14) must be used.



Figure 5.13: Equivalent system to the one in Fig. 5.12, valid for noise propagation analysis in case of a WSS Gaussian white noise input.

5.5.1.2 Validation in simulations

To verify the derived model, a few examples of estimating S_y are given, for several different filters H_{eq} discretized with $T_{s,1} = \frac{1}{f_{s,1}} = \frac{1}{Mf_{s,2}}$. The decimation is performed with frequency $f_{s,2} = 2f_{pwm} = 40$ kHz, to correspond to the hardware set-up used in this chapter. The tested filters are:

- Derivative filter, $H_{eq,1}(z) = \frac{k_{ad}}{T_{s,1}} (1 z^{-1})$, with the gain $k_{ad} = 1.52 \cdot 10^{-5}$. The exact derivative gain is not important, however, in Section 6.6 this specific value is used in an active damping control scheme, for which the noise sensitivity is examined.
- Proportional gain with $k_{p,i}$ from Table 5.1, $H_{eq,2}(z) = k_{p,i}$.
- Cascade of the same proportional gain as above and the DLPF from (5.11), $H_{eq,3}(z) = k_{p,i}G_{DLPF}(z)$.
- PID structure used as the voltage controller in (5.4) with the parameters from Table 5.1, $H_{eq,4}(z) = G_{c,v}(z)$.
- Cascade of the same PID structure as above and 3 DLPFs from (5.11), $H_{eq,5}(z) = G_{c,v}(z)G_{DLPF}^3(z)$.

The simulation model, corresponding to Fig. 5.12, is implemented in MATLAB/Simulink, where n is obtained from a band-limited white noise generator. The noise variance is arbitrarily chosen as $\sigma_n^2 = 150 \cdot 10^{-3}$. The output signal y is acquired with $T_{s,2} = \frac{1}{f_{s,2}}$ with data length of 200 ms, which yielded a 5 Hz spectral resolution. Results are given for $M \in \{1, 2, 4, 8, 16\}$. Note that, the filters $G_{DLPF}(z)$ and $G_{DLPF}^3(z)$ are left out for M = 1 as it was only of interest to test their impact on the aliasing, which occurs for M > 1. MATLAB is used for postprocessing of simulation results, by calculating the output PSD and the total output noise power. For a better visualization, the PSD is filtered using a MAF with 100 Hz window and is plotted with a thinner line compared to analytical results. For comparison with the analytical predictions, the PSD is scaled with the input variance σ_n^2 such that the relative output noise power corresponds to the area below the given traces. The analytical traces are, correspondingly, calculated as $\frac{S_y(\omega)}{\sigma_n^2} = \frac{2}{f_{s,1}} |H_{eq}^{M:1}(\omega)|^2$ (see (5.16)). The relative output noise power is calculated as the integral of the scaled PSD from 200 Hz to $f_x = 10$ kHz. Lower frequencies are not considered due to the finite spectral resolution of the simulated PSD. Results of the output noise power are shown scaled with the result for M = 1, i.e. the case without the decimation occurring.

First, in Fig. 5.14, the results are shown for the case of the derivative filter $H_{eq,1}(z) = \frac{k_{ad}}{T_{s,1}} (1 - z^{-1})$. It can be seen that, as M is increased, the PSD flattens out and increases as well, which results in the output noise power amplification. The second result, for the purely proportional gain $H_{eq,2}(z) = k_{p,i}$, is shown in Fig. 5.15. The PSDs perfectly overlap for all values of M, meaning that there is no impact of increasing $f_{s,1}$ above $f_{s,2}$, which is consistent with the illustration from Fig. 5.9. The third result, shown in Fig. 5.16, shows the impact of introducing the DLPF from (5.11) in the cascade with the same proportional gain from above, in order to suppress the aliasing effect, $H_{eq,3}(z) = k_{p,i}G_{DLPF}(z)$. It can be seen that the output noise power is now strongly reduced by increasing the value of M. This result provides additional motivation for introducing the DLPF in the subsequent experimental tests; however, this model does not consider the impact of the closed-loop, which is addressed in the following section. The fourth result, shown in Fig. 5.17, shows the propagation for the PID structure (with the filtered derivative action) from (5.4), $H_{eq,4}(z) = G_{c,v}(z)$. It can be seen that, except for M = 2, the output noise power is amplified. Finally, for the same PID structure, results in Fig. 5.18 show that the output noise power is strongly suppressed by introducing a cascade of 3 DLPFs from (5.11), $H_{eq,5}(z) = G_{c,v}(z)G_{DLPF}^3(z)$. For all cases shown, the match between the simulations and the analytical results is practically perfect, which validates the derived model for open-loop systems.



Figure 5.14: Impact of the open-loop decimation for the derivative filter, $H_{eq,1}(z) = \frac{k_{ad}}{T_{s,1}} \left(1 - z^{-1}\right)$: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.15: Impact of the open-loop decimation for the proportional gain, $H_{eq,2}(z) = k_{p,i}$: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.16: Impact of the open-loop decimation for the proportional gain and the DLPF from (5.11), $H_{eq,3} = k_{p,i}G_{DLPF}(z)$: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.17: Impact of the open-loop decimation for the PID structure from (5.4), $H_{eq,4}(z) = G_{c,v}(z)$: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.18: Impact of the open-loop decimation for the PID structure from (5.4) and the cascade of 3 DLPFs from (5.11), $H_{eq,5}(z) = G_{c,v}(z)G_{DLPF}^3(z)$: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.

5.5.2 Closed-loop systems

In this section, the case of decimation occurring inside a closed-loop system is analyzed. This case is of interest for modeling the DPWM decimation effects explained in Section 5.4. Additionally, it is directly applicable to the widely used control strategy in power electronics, where the feedback is oversampled, filtered, and then decimated to a lower control update rate, i.e. the MR-(S/D)S-PWM control [63, 110, 111]. Although being a very common choice in industrial applications, no simple and effective analysis is present in the literature to model the impact of such system on noise propagation.

5.5.2.1 Model derivation



Figure 5.19: Closed-loop system with a multi-rate structure: (a) reference tracking system of interest; (b) simplified system focused on noise propagation; (c) equivalent system used for calculating the output PSD, where the input noise is represented as a sum of two uncorrelated WSS and Gaussian processes.

Let us consider a closed-loop reference tracking system with a band-limited white measurement noise n_m being added to the feedback, as shown in Fig. 5.19a. The feedback is oversampled with frequency $f_{s,1} = M f_{s,2}$ and processed by a filter $G_{fb}(z)$. In case of MR-DS-PWM, the decimation to the control rate $f_{s,2}$ occurs after $G_{fb}(z)$. In case of MS-PWM, which is illustrated in Fig. 5.19a, the filtered feedback is subtracted from the reference r and processed by the controller $G_c(z)$, all with frequency $f_{s,1}$. Then,

the DPWM itself down-samples its input with frequency $f_{s,2} = 2f_{pwm}$. The DPWM output is processed by the plant filter G_p . Note again that the DPWM has an intrinsic ZOH action, meaning that its output is not actually in the discrete time domain with frequency $f_{s,2}$, but rather in the continuous time domain. It is assumed that the frequency response of G_p correctly incorporates the modulation effects, as described before. Therefore, the analyzed system features 3 domains: the oversampled domain with $f_{s,1}$, the downsampled DPWM domain with $f_{s,2}$, and the continuous time domain⁴. As explained before, the bandwidth of interest for noise propagation analysis is set below f_{pwm} , to be applicable for converter models obtained using the state-space averaging and also to avoid problems related to PSD measurements at frequencies where the switching ripple dominates the spectra.

To simplify the analyzed system, the following steps are taken. First of all, considering that n_m is uncorrelated to the feedback y, the sampler $f_{s,1}$ can be applied independently to y and n_m , resulting in the sampled WSS Gaussian process n being the same as analyzed for the open-loop case. Next, it is assumed that the closed-loop system has a low-pass nature, owing to the filtering characteristic of G_p , which is determined by the output filter of the converter. Low-pass filtering nature of G_p is the main simplifying assumption, as it implies that the impact of noise on the output signal y is band-limited well-below $f_{pwm} = \frac{f_{s,2}}{2}$. Hence, the Nyquist frequency of the sampler $f_{s,1}$ is always above the bandwidth of the signal y and, therefore, the spectrum of its sampled output is equal to the one of y in the frequency range of interest. This allows for the simplification of the system, such that the sampler $f_{s,1}$ is completely neglected and the only considered rate change is the decimation between the controller and the plant. Fig. 5.19b shows the simplified system, considering only noise propagation (r = 0) and merging together the filter and the controller, such that $H_{eq}(\omega) = -G_{fb}(\omega)G_c(\omega)$.

As the discrete input noise signal is band-limited up to $\frac{f_{s,1}}{2}$, its propagation cannot be analyzed directly, without incorporating effects of the down-sampler M. Let us again use the important fact that the noise is assumed to be a stationary signal, hence it exhibits no spectral correlation. This allows us to partially process n across its bandwidth. For this purpose, n is represented as a sum of two signals, n_1 and n_2 , with the following PSDs:

$$S_{n,1} = \begin{cases} S_n = \frac{2}{f_{s,1}} \sigma_n^2, & \text{if } f < \frac{f_{s,2}}{2} \\ 0, & \text{if } \frac{f_{s,2}}{2} < f < \frac{f_{s,1}}{2} \end{cases}$$

$$S_{n,2} = \begin{cases} 0, & \text{if } f < \frac{f_{s,2}}{2} \\ S_n = \frac{2}{f_{s,1}} \sigma_n^2, & \text{if } \frac{f_{s,2}}{2} < f < \frac{f_{s,1}}{2} \end{cases}$$
(5.17)

Given that n_1 and n_2 exhibit no cross-correlation, after being processed by the closed-loop system, the resulting output signals y_1 and y_2 remain uncorrelated. Hence, the total output PSD can be found as an algebraic sum of the two PSDs, $S_{y,1}$ and $S_{y,2}$, corresponding to y_1 and y_2 , respectively. This is illustrated in Fig. 5.19c. As noted above, for all PSD calculations, the spectral region of interest⁵ is

⁴This is also illustrated in Fig. 5.1, where $f_{s,1} = f_s$ and $f_{s,2} = 2f_{pwm}$.

⁵All subsequently derived PSD expressions are valid only in the range $f \in \left(0, \frac{f_{s,2}}{2}\right)$, which, for clarity, is left out of the expressions.

below $\frac{f_{s,2}}{2}$.

Let us first consider the case of n_1 . This signal is purposely chosen to be limited exactly by the Nyquist frequency of the down-sampler M, $\frac{f_{s,2}}{2}$, meaning that it does not cause any spectral fold-back. Let us define the open-loop transfer function, neglecting the decimator, as:

$$W_{ol,n}(\omega) = H_{eq}(\omega)G_p(\omega).$$
(5.18)

The propagation of n_1 is trivial to analyze as it corresponds to the linear processing by a single-rate closed-loop system with the open-loop transfer function from (5.18):

$$S_{y,1}(\omega) = S_{n,1}(\omega) \frac{|W_{ol,n}(\omega)|^2}{|1 - W_{ol,n}(\omega)|^2} = S_n \frac{|W_{ol,n}(\omega)|^2}{|1 - W_{ol,n}(\omega)|^2}.$$
(5.19)

Note that the negative sign is incorporated in H_{eq} to correspond to the system shown in Fig. 5.19a.

The high-frequency part is more challenging for the analysis. Let us consider the same closed-loop system from Fig. 5.19b, where the signal indices will be labeled with 2, to indicate that the analysis refers to the high-frequency noise signal n_2 . It can be seen that the signal q_2 is equal to $q_2 = n_2 + y_2$. Although n_2 is found in the frequency range $\left(\frac{f_{s,2}}{2}, \frac{f_{s,1}}{2}\right)$, after being processed by H_{eq} and down-sampled M times, the PSD of the digital signal $m_{s,2,2}$ is folded back below $\frac{f_{s,2}}{2}$. Based on the above-mentioned low-pass assumption of G_p , it is concluded that the signal y_2 remains band-limited below $\frac{f_{s,2}}{2}$ even after the inherent holder action performs the conversion to the continuous time domain. This crucial assumption allows us to claim that the signal q_2 is completely determined by n_2 above $\frac{f_{s,2}}{2}$ and by y_2 below $\frac{f_{s,2}}{2}$. To propagate the signal q_2 to the output y_2 , spectral fold-back needs to be applied only to the part above $\frac{f_{s,2}}{2}$, which results in:

$$y_{2}(\omega) = G_{p}(\omega) \sum_{k=0}^{M-1} n_{2}(\omega - k2\pi f_{s,2})H_{eq}(\omega - k2\pi f_{s,2}) + W_{ol,n}(\omega)y_{2}(\omega) =$$

$$= \frac{G_{p}(\omega)}{1 - W_{ol,n}(\omega)} \sum_{k=0}^{M-1} n_{2}(\omega - k2\pi f_{s,2})H_{eq}(\omega - k2\pi f_{s,2}).$$
(5.20)

Let us now find the PSD of y_2 in the same way as for the open-loop system in (5.14), again using the fact that stationary signals exhibit no spectral correlation:

$$S_{y,2}(\omega) = \frac{1}{f_{s,1}} y_2(\omega) y_2^*(\omega) =$$

$$= \frac{1}{f_{s,1}} \frac{|G_p(\omega)|^2}{|1 - W_{ol,n}(\omega)|^2} \left(\sum_{k=0}^{M-1} n_2(\omega - k2\pi f_{s,2}) H_{eq}(\omega - k2\pi f_{s,2}) \right) \left(\sum_{l=0}^{M-1} n_2^*(\omega - l2\pi f_{s,2}) H_{eq}^*(\omega - l2\pi f_{s,2}) \right) =$$

$$= \frac{|G_p(\omega)|^2}{|1 - W_{ol,n}(\omega)|^2} \sum_{k=0}^{M-1} S_{n,2}(\omega - k2\pi f_{s,2}) |H_{eq}(\omega - k2\pi f_{s,2})|^2 =$$

$$= S_n \frac{|G_p(\omega)|^2}{|1 - W_{ol,n}(\omega)|^2} \sum_{k=1}^{M-1} |H_{eq}(\omega - k2\pi f_{s,2})|^2.$$
(5.21)

Ph.D. Thesis — Ivan Petrić

Note that in the last row of (5.21) the baseband is missing as $S_{n,2} = 0$ for k = 0 (see (5.17)). Nevertheless, the expression (5.21) is exactly valid in the range $f \in \left(0, \frac{f_{s,2}}{2}\right)$, due to the spectral fold-back.

Summing $S_{y,1}$ and $S_{y,2}$, the total output PSD is found as:

$$S_{y}(\omega) = S_{y,1}(\omega) + S_{y,2}(\omega) =$$

$$= S_{n} \frac{|W_{ol,n}(\omega)|^{2}}{|1 - W_{ol,n}(\omega)|^{2}} + S_{n} \frac{|G_{p}(\omega)|^{2}}{|1 - W_{ol,n}(\omega)|^{2}} \sum_{k=1}^{M-1} |H_{eq}(\omega - k2\pi f_{s,2})|^{2} =$$

$$= S_{n} \frac{1}{|1 - W_{ol,n}(\omega)|^{2}} \left(|H_{eq}(\omega)|^{2} |G_{p}(\omega)|^{2} + |G_{p}(\omega)|^{2} \sum_{k=1}^{M-1} |H_{eq}(\omega - k2\pi f_{s,2})|^{2} \right) =$$

$$= S_{n} \frac{|G_{p}(\omega)|^{2}}{|1 - W_{ol,n}(\omega)|^{2}} \sum_{k=0}^{M-1} |H_{eq}(\omega - k2\pi f_{s,2})|^{2} =$$

$$= \frac{2}{f_{s,1}} \sigma_{n}^{2} \frac{|G_{p}(\omega)|^{2}}{|1 - W_{ol,n}(\omega)|^{2}} |H_{eq}^{M:1}(\omega)|^{2}.$$
(5.22)

Note that this corresponds to using the same equivalent magnitude response (applied to transfer functions between the sampled noise and the decimator) as derived for the open-loop system, only in the forward path, while the feedback impact is determined by the loop without the rate change. This result is important as it allows to use just the magnitude response for the spectral fold-back, while the feedback still incorporates the phase response of the system, which impacts the filtering performance of the closed-loop structure.

A block diagram corresponding to (5.22) is shown in Fig. 5.20. The block $|CL(\omega)|^2$ represents the squared magnitude response of a loop that comprises the single-rate open-loop transfer function $W_{ol,n}$ from (5.18) in its feedback. In case of MS-PWM, H_{eq} includes the filter and the controller, while in the case of MR-(S/D)S-PWM it includes only the filter (the controller would be placed in the cascade with G_p). Another difference between MS-PWM and MR-(S/D)S-PWM is in the frequency response of G_p , which needs to include the appropriate DPWM small-signal model from (2.3). In general terms, H_{eq} consists of all oversampled blocks, i.e. those found between the samplers $f_{s,1}$ and $f_{s,2}$ in Fig. 5.19a. The system in Fig. 5.20 is only valid for a low-pass plant and a WSS Gaussian white noise input. For other input noise properties, the analysis cannot be simplified to this form.



Figure 5.20: Equivalent system to the one in Fig. 5.19a, valid for noise propagation analysis in case of a low-pass G_p and a WSS Gaussian white noise input.

5.5.2.2 Validation in simulations

Same as for the open-loop case, simulations are used to verify the derived noise propagation model. The simulated model is implemented to completely correspond to Fig. 5.19a, with the reference being set to 0. The oversampled period is set as $T_{s,1} = \frac{1}{f_{s,1}} = \frac{1}{Mf_{s,2}}$, while the decimation rate is set as $f_{s,2} = 2f_{pwm} = 40$ kHz, to correspond to the set-up used in this chapter. Relative output noise power is again found in the range from 200 Hz to $\frac{f_{pwm}}{2} = 10$ kHz. The plant transfer function is simulated in the continuous time domain, while the ZOH is included after the re-sampler $f_{s,2}$. Therefore, the simulated system features 3 domains, just like discussed before.

The tested controllers, filters, and plant transfer functions, with the parameters from Tables 3.2 and 5.1 are:

- Current loop with $G_{p,i}$ from (5.1), PI controller from (5.3), without any feedback filter.
- Current loop with $G_{p,i}$ from (5.1), PI controller from (5.3), with the DLPF from (5.11).
- Voltage loop with $G_{p,v}$ from (5.2), PID controller from (5.4), without any feedback filter.
- Voltage loop with $G_{p,v}$ from (5.2), PID controller from (5.4), with the DLPF from (5.11).
- Voltage loop with $G_{p,v}$ from (5.2), PID controller from (5.4), with 3 DLPFs from (5.11).
- Voltage loop with $G_{p,v}$ from (5.2), PID controller from (5.4), with MAF from (2.9).

The controllers are implemented without one step computation delay. As before, G_{fb} is only used for M > 1, to verify its impact on suppressing the aliasing.

The results are shown in Figs. 5.21 - 5.26. Again, the simulations show an excellent match with the analytical results, which validates the derived model. Trends related to the impact of adding the feedback filters are similar to those for the before-tested open-loop systems.

To conclude, the expression (5.22), derived in this section, can be used to replace the qualitative approach from Section 5.4.2 in designing the anti-aliasing digital low-pass filters needed to suppress the impact of the decimation on noise propagation.



Figure 5.21: Impact of the closed-loop decimation for $G_c = k_{p,i} + k_{i,i}T_s \frac{1}{1-z^{-1}}$, $G_{fb} = 1$, G_p from (5.1), with parameters from Tables 3.2 and 5.1: (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.22: Impact of the closed-loop decimation for $G_c = k_{p,i} + k_{i,i}T_s \frac{1}{1-z^{-1}}$, G_{fb} is the DLPF from (5.11), G_p from (5.1): (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.23: Impact of the closed-loop decimation for $G_c = k_{p,v} + k_{i,v}T_s \frac{1}{1-z^{-1}} + \frac{k_d}{T_s}(1-z^{-1})G_{df}(z)$, $G_{fb} = 1$, G_p from (5.2): (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.24: Impact of the closed-loop decimation for $G_c = k_{p,v} + k_{i,v}T_s \frac{1}{1-z^{-1}} + \frac{k_d}{T_s}(1-z^{-1})G_{df}(z)$, G_{fb} is the DLPF from (5.11), G_p from (5.2): (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.25: Impact of the closed-loop decimation for $G_c = k_{p,v} + k_{i,v}T_s \frac{1}{1-z^{-1}} + \frac{k_d}{T_s}(1-z^{-1})G_{df}(z)$, G_{fb} is a cascade of 3 DLPFs from (5.11), G_p from (5.2): (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.



Figure 5.26: Impact of the closed-loop decimation for $G_c = k_{p,v} + k_{i,v}T_s \frac{1}{1-z^{-1}} + \frac{k_d}{T_s}(1-z^{-1})G_{df}(z)$, G_{fb} is the MAF from (2.9), G_p from (5.2): (a) the PSD relative to the input noise variance; (b) the output variance up to $\frac{f_{pwm}}{2} = 10$ kHz, relative to the result for M = 1.

5.6 Experimental verification of noise propagation

Noise propagation properties are verified experimentally on the buck-type converter, described in Section 5.2.1. Hardware parameters are equal to those in Table 3.2 and the controller parameters are equal to those in Table 5.1.

The control system is implemented on the FPGA platform, described in Section 3.2. The 12-bit ADC quantization in the experimental set-up corresponds to $LSB_i = 17$ mA for the current loop, and $LSB_v = 75$ mV for the voltage loop.

For postprocessing, 50 ms of data is acquired with a 250 MS/s rate, using the high-resolution mode (15 bits) of Tektronix MS056 oscilloscope with channel bandwidths set to 20 MHz. The inductor current is sensed using Tektronix TCP202 current probe, and the capacitor voltage is sensed using ROHDE & SCHWARZ RT-ZD01 differential voltage probe, with the attenuation set to 100. Signal offsets and scales that result in the maximal resolution are set on the oscilloscope.

Measurements are conducted for steady-state duty cycles equal to D = [0.2, 0.3, 0.4, 0.5, 0.6, 0.7, 0.8]. To avoid overcrowding the figures, for all tested operating points, the analytical results use the DPWM model from (2.3) with only D = 0.5, as its operating point dependency was verified to have a small impact in the frequency range of interest. By covering a wide range of operating points, the sensitivity of noise propagation properties can be examined. This is important as in experimental prototypes certain amount of switching noise is always present. All presented measurements were repeated multiple times, in order to verify that they do not significantly vary from one measurement to another. The input voltage for the voltage loop tests is set to $V_{in} = 160$ V, so that the voltage probe attenuation would not have to be changed to 1000, which would increase the noise resulting from the oscilloscope itself. For the current loop tests, the input voltage is set to $V_{in} = 200$ V. The upper frequency limit for the variance calculation is arbitrarily chosen as $f_x = 0.4 f_{pwm} = 8$ kHz.

The input noise powers, corresponding to i_L and v_c measurements, are estimated in the following way. The converter is set to run in an open-loop condition and the feedback is sampled once per switching period, for the total length of 50 ms. In this way, both impacts of the closed-loop and the switching harmonics are removed and the input noise power is estimated from the spectral content of the sampled signal. These measurements were repeated and averaged across the tested operating points. The resulting values of the input noise powers are $\sigma_{n,i}^2 = 1.7 \cdot 10^{-3} \text{ A}^2$ and $\sigma_{n,v}^2 = 154 \cdot 10^{-3} \text{ V}^2$. These values are significantly higher than the quantization noise powers, determined by $\frac{\text{LSB}_{i,v}^2}{12}$ [3]. For all tests in this section, the results are given relative to the lowest tested value of N; hence, the information on the input noise power is not used. However, it is used in Section 6.6, where absolute values of the output noise power are given.

Two switching periods of i_L and v_c are shown in Fig. 5.27, for the highest possible sampling rate of the oscilloscope, without bandwidth limitations on the input channels. The presence of the switching noise is clearly seen, and its magnitude compared to the switching ripple component is much higher in the capacitor voltage than in the inductor current. In order to verify that the spikes are an actual differential voltage across the capacitor terminals, both leads of the probe were connected to the negative capacitor terminal. This ground voltage measurement is also shown in Fig. 5.27 as v_g and it can be seen that its value is negligible.



Figure 5.27: Oscilloscope screen capture showing switching components over 2 switching periods. The blue trace features 1 V/div scale and corresponds to $v_c(t)$. The orange trace features 500 mA/div scale and corresponds to $i_L(t)$. The green trace features 1 V/div scale and corresponds to the voltage measured by connecting both probe leads to the capacitor ground connection, $v_q(t)$.

Cascade of the feedback filter and the controller is labeled as $H_{eq} = -G_c G_{fb}$, for consistency with Section 5.5. For all following results, two traces are plotted for the analytical predictions based on (5.7). The first one corresponds to not including the decimation impact, i.e. $H_n(\omega)$ is determined as in (5.8). The second one corresponds to the extended model from (5.22); referring to (5.7), $|H_n(\omega)|^2$ is replaced by:

$$|H_n(\omega)|^2 = \frac{|G_{dpwm}(\omega)G_p(\omega)|^2}{|1 - H_{eq}(\omega)G_{dpwm}(\omega)G_p(\omega)|^2} |H_{eq}^{M:1}(\omega)|^2.$$
(5.23)

The decimation frequency is equal to $2f_{pwm}$. Note again that the frequency response of the cascade $G_{dpwm}G_p$ is obtained by first transforming it to the Z-domain, using the impulse-invariant discretization with frequency $f_s = Nf_{pwm}$. Both analytical models are used in order to show that introducing the appropriate filtering brings the MS-PWM noise propagation properties close to the ones of linear systems.

The controller/filter cascades validated in Section 5.5.2.2 correspond to those that are introduced in Section 5.4.2 based on the qualitative analysis, and that are experimentally tested here. Note, however, that there are differences between the model simulated in Section 5.5.2.2 and the experimental set-up, so the results are not expected to be equivalent. First of all, in MS-PWM, the DPWM is *assumed* to re-sample the modulating waveform twice per switching period. However, it is not exactly an ideal, theoretical, decimator so it is of interest to see how the derived theory directly applies to that case. Secondly, in the simulations from Section 5.5.2.2, the plant model (in the continuous time domain) follows a re-sampler and a ZOH with $T_{s,2}$, which does not match the small-signal response of the multisampled modulator.

5.6.1 Benchmark simulation settings

Simulations are performed in MATLAB/Simulink (using PLECS Blockset) by adding a random signal to the feedback, as for the validations in Section 5.5. The injected signal features high enough bandwidth, so that it can be considered as white noise for all examined sampling frequencies. In simulations, data acquisition, ADC and DPWM quantization effects match those in the experimentally tested prototype. One of the motivations for providing the simulation results here is to compare noise propagation between the simulations that feature practically ideal white noise injection and the experiments with realistic noise coming from the set-up. The switching noise is not simulated, as it is strongly dependent on hardware realisation; hence, the results would lack generality in conclusions. The simulations were performed for many operating points and different values of input noise powers. The relative noise attenuation results did not change; hence, the experimental results are here compared with the simulations for an arbitrarily chosen D = 0.4 and the input noise powers corresponding to the ones that are estimated from the experimental set-up.

5.6.2 Results for the PI current loop

Variances of i_L corresponding to the measured, simulated, and analytical results are shown in Figs. 5.28a and 5.29a. The results are given relative to the lowest tested oversampling factor (N = 1 for the current loop).

In Fig. 5.28a, the results are given for the control loop without any digital feedback filters, where $H_{eq}(z) = -G_{c,i}(z)$. It can be seen that the linear analysis using (5.8) shows a good match for (S/D)S-PWM. For N > 2, a great mismatch from the analytical values using (5.8) is obtained both in the simulations and the experimental measurements. Including the extended model from (5.23) brings a very good prediction of the MS-PWM behavior. This is the first confirmation that DPWM's behavior truly resembles a re-sampler. As observed, for N > 2, the noise power remains nearly constant, with small variations caused by the impact of the phase margin on filtering performance. Note that this result is completely coherent with the previous reasoning regarding the impact of aliasing for the high-frequency proportional-dominant controllers. Namely, if no digital filtering is used to attenuate components in zone B in Fig. 5.9, the DPWM decimation action causes aliasing and no benefit in terms of noise attenuation is obtained compared to DS-PWM. The spectral plots in Fig. 5.28b show negligible difference in the noise floor between the cases with N = 2 and N = 32. All spectral results are averaged over 60 Hz window for clarity and the dc component is removed. For this control loop realization, the phase margins are equal to $\{25.75^{\circ}, 53.3^{\circ}, 66.98^{\circ}, 73.77^{\circ}, 77.15^{\circ}, 78.84^{\circ}\}$ for $N = \{1, 2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 1 and averaged over the tested values of D, are equal to $\{-5.2, -6.45, -6.75, -6.76, -7.12\}$ dB, for N starting from 2. It can be seen that a strong noise attenuation is obtained when increasing N from 1 to 2, both due to doubling of the sampling frequency and a significant increase of the phase margin that results in a higher filtering action by the closed-loop system.



Figure 5.28: Comparison between the experiments, simulations, and analytical models for the current loop without a feedback filter: (a) variance results (b) spectral results (experimental).

In Fig. 5.29a, the variance results are shown for the case when the DLPF is introduced in the feedback, i.e. $H_{eq}(z) = -G_{c,i}(z)G_{DLPF}(z)$. It can be seen that the addition of $G_{DLPF}(z)$ brings a good match between the simulations, experiments, and both analytical models (with and without including the decimation). This confirms that appropriate high-frequency filtering brings the MS-PWM noise propagation properties close to the ones of linear systems. The trends are consistent for all operating points. For this control loop realization, the phase margins are equal to $\{25.75^{\circ}, 53.3^{\circ}, 61.2^{\circ}, 68^{\circ}, 71.38^{\circ}, 73.1^{\circ}\}$ for $N = \{1, 2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 1 and averaged over the tested values of D, are equal to $\{-5.2, -7.81, -10.77, -13.3, -16.6\}$ dB, for N starting from 2. It can be seen that, for higher values of N where the phase margins are very similar, the noise power is attenuated by approximately -3 dB for each doubling of N. These results confirm that the multi-sampled control loop with the $G_{DLPF}(z)$ significantly suppresses the feedback noise, without strongly affecting the dynamic performance. In Fig. 5.29b, the spectral results show a significant difference in the noise floor when the multisampling factor is increased. It should be noted again that the results for N = 1 and N = 2 are obtained without $G_{DLPF}(z)$ as the DPWM decimation does not occur for those cases.



Figure 5.29: Comparison between the experiments, simulations, and analytical models for the current loop with $G_{DLPF}(z)$ from (5.11): (a) variance results (b) spectral results (experimental).

For a time domain illustration, the peaks of i_L are detected and plotted in Fig. 5.30 for N = 1and N = 32 with the DLPF, being the extreme cases. The plot features relative values, obtained by subtracting the mean value of the current peaks and dividing by the dc value of i_L , for D = 0.4.

From these results, it is concluded that MS-PWM control loops with the PI controllers can offer high noise attenuation, without a significant impact on the dynamic response.

5.6.3 Results for the PID voltage loop

In Fig. 5.31, variance results are shown for the voltage loop, without and with $G_{DLPF}(z)$ added in the feedback. The noise power is shown relative to N = 2. It is interesting to note that here, unlike



Figure 5.30: Experimental results showing a time-domain impact of noise for the tested current loops. The case with N = 1 without a feedback filter is compared to the case with N = 32 and $G_{DLPF}(z)$ from (5.11). The results show the relative variation of the detected current peak values.

for the current loop, using MS-PWM control without any digital filters (Fig. 5.31a) results in noise amplification rather than attenuation, which is compliant with the conclusions made in Sections 5.4 and 5.5. For this control loop realization, the phase margins are equal to $\{20.1^{\circ}, 35.5^{\circ}, 43.1^{\circ}, 46.8^{\circ}, 48.7^{\circ}\}$ for $N = \{2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 2 and averaged over the tested values of D, are equal to $\{6.36, 6.16, 6.64, 7.3\}$ dB, for N starting from 4. With the increase of N, even though the phase margin is increased, the noise power is amplified due to the high-frequency behavior of the derivative action. Again, the extended analytical results that include the decimation impact show an excellent match with the simulations and experiments, except for N = 4. At this point, it is assumed that the mismatch for N = 4 may come from some nonlinear phenomenon that is not modeled.

When the $G_{DLPF}(z)$ is introduced (Fig. 5.31b), the aliasing is suppressed; however, the noise power is still not significantly reduced compared to DS-PWM (N = 2). The phase margins are equal to $\{20.1^{\circ}, 30.2^{\circ}, 37.8^{\circ}, 41.6^{\circ}, 43.4^{\circ}\}$ for $N = \{2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 2 and averaged over the tested values of D, are equal to $\{3.37, 1.21, -0.482, -2.23\}$ dB, for N starting from 4. The same conclusions are brought regarding the match with the analytical models.

The experimental results show consistent trends with the change of the operating point. A higher mismatch in Fig. 5.31a can be seen for D = 0.2 and N = 8, and in Fig. 5.31b for D = 0.2 and N > 4, and D = 0.8 and N = 8. It is assumed that this is caused by sampling the switching noise or by the discontinuity-related nonlinearities, which both may appear around those values of D and N [24]. The sensitivity to switching noise propagation depends on the values of N and D, and the impulse response of $H_{eq}(z)$. The switching noise differs greatly from white noise; hence, it is not covered by the models derived in this chapter. However, it should be noted that the voltage loop shows a higher sensitivity to switching noise than the current loop. This sensitivity certainly rises due to the derivative action in the



Figure 5.31: Comparison between the variances obtained from the experiments, simulations, and analytical models for the voltage loop (a) without a feedback filter and (b) with $G_{DLPF}(z)$ from (5.11).

controller, as well as the higher relative magnitude of the switching noise, seen in Fig. 5.27.

In order to further test the suppression of the aliasing effect, the third-order $G_{DLPF}^3(z)$ is added to the feedback. These results are shown in Fig. 5.32. The variance results in Fig. 5.32a show that the DPWM decimation is indeed suppressed by using filters with a higher attenuation at $f > f_{pwm}$, and the noise power keeps decreasing with the increase of N. The simulated and experimental results show a good match with the analytical predictions, both with and without including the decimation effects. For this control loop realization, the phase margins are equal to $\{20.1^{\circ}, 19.7^{\circ}, 27.3^{\circ}, 31.1^{\circ}, 33^{\circ}\}$ for $N = \{2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 2 and averaged over the tested values of D, are equal to $\{-0.97, -4.87, -7.44, -10.38\}$ dB, for N starting from 4. Again, for higher values of N, the noise power is reduced by approximately -3 dB for each doubling of N. A similar operating point dependency is seen as the one for the cases without the filter and with the $G_{DLPF}(z)$. In Fig. 5.32b, the spectral plots are given to show the impact of MS-PWM control on lowering the noise floor.



Figure 5.32: Comparison between the experiments, simulations, and analytical models for the voltage loop with a cascade of 3 DLPFs from (5.11), $G_{DLPF}^3(z)$: (a) variance results; (b) spectral results (experimental).

The filter $G_{DLPF}^3(z)$ introduces nearly 16° phase lag at $f_{c,v}$, which is close to using a MAF, where N latest feedback samples are averaged, i.e. $G_{MAF}(z)$ from (2.9). Hence, the feedback configuration with $G_{MAF}(z)$ is tested as well, and the variance results are shown in Fig. 5.33. These experimental results show an excellent match with the simulations and analytical predictions, while being very consistent for all operating points. For this scenario, the phase margins are equal to $\{20.1^\circ, 23.1^\circ, 28.6^\circ, 31.4^\circ, 32.7^\circ\}$ for $N = \{2, 4, 8, 16, 32\}$ respectively. The experimentally obtained variances, relative to N = 2 and averaged over the tested values of D, are equal to $\{-3.9, -7.4, -10.5, -13.5\}$ dB, for N starting from 4. Figure 5.34 is given for a comparison of the voltage spectra for all tested feedback configurations.

For a time-domain illustration, the peaks of v_c are detected and plotted in Fig. 5.35 for N = 2and N = 32 with the MAF, being the extreme cases. The plot features relative values, obtained by



Figure 5.33: Comparison between the variances obtained from the experiments, simulations, and analytical models for the voltage loop with $G_{MAF}(z)$ from (2.9).



Figure 5.34: Spectral results (experimental) in the frequency window used for the variance calculations for all tested feedback configurations of the voltage loop.

subtracting the mean value of the voltage peaks, and dividing by the dc value of v_c , for D = 0.6.

It is concluded that, for MS-PWM control loops with the PID controllers, high noise attenuation can only be achieved with a significant impact on the system dynamics (Fig. 5.32 and Fig. 5.33). It is recommended to use $G_{MAF}(z)$ instead of $G_{DLPF}^3(z)$ due to the higher noise suppression, better consistency across operating points, and a similar impact on the dynamic response. Furthermore, the MAF removes the switching ripple, which suppresses the problems caused by the MS-PWM nonlinearities discussed in Chapter 4. Even if strong noise attenuation is not required, it is still recommended to use anti-aliasing digital low-pass filters in the PID-based MS-PWM loops, e.g. $G_{DLPF}(z)$, as the dynamic response is not notably deteriorated and the output noise power is significantly reduced.

It should be noted that, as N is increased, even with the use of filters with a high attenuation (such as MAFs), the dynamic properties are improved compared to the DS-PWM without any filtering. A



Figure 5.35: Experimental results showing time-domain impact of noise for the tested voltage loops. The case with N = 2 without a feedback filter is compared to the case with N = 32 and $G_{MAF}(z)$ from (2.9). The results show the relative variation of the detected voltage peak values.

comparison in terms of the noise suppression and dynamic performance between DS-PWM, MS-PWM, and MR-DS-PWM methods is provided in the following section.

5.6.4 Comparison between MS-PWM and MR-DS-PWM methods



Figure 5.36: Timing illustration for MR-DS-PWM with the MAF and one step computation delay. The control period is labeled with T_c and is, for the case shown, 8 times greater than the sampling period T_s .

In this section, noise propagation is examined for the multi-rate strategies that use oversampling only for filtering purposes. These methods, readily implementable in commercially available DSPs, belong to the MR-(S/D)S-PWM group, for which the feedback is acquired with a high sampling rate, filtered, and then decimated to the single- or double-update rate [63,110]. An example of a timing diagram for MR-DS-PWM with the MAF is given in Fig. 5.36. The analysis is performed for the current control loops.
Again, the results are compared with both analytical models, with and without including the decimation impact. It is interesting to note that practically the same extended model is used for MS-PWM and MR-DS-PWM, as both cases feature the decimation inside the closed-loop system. A small, but important, difference for MR-DS-PWM is that the equivalent fold-back transfer function H_{eq} in (5.23) includes only G_{fb} , while the controller is placed in the cascade with G_p . Another difference is found in the frequency response of G_{dpwm} and one step computation delay (if added), which are determined by Nf_{pwm} for MS-PWM and $2f_{pwm}$ for MR-DS-PWM.

In this section, five cases of different control strategies are compared in terms of dynamics and noise attenuation. For all oversampled strategies, N = 16 is chosen. Case A represents the standard DS-PWM control, which is used as a benchmark. The equivalent time delay, introduced by the modulation and one step computation delay, is approximately $\tau_{d,A} = \frac{3}{2}T_{s,A} = \frac{3}{4}T_{pwm}$. Case B represents MR-DS-PWM control with $G_{MAF}(z)$ from (2.9). When implemented in DSPs, this strategy relies on a DMA module to buffer the highly-oversampled feedback signal, which is then averaged over the switching period at the double-update rate [63]. Given that the phase response of the MAF over T_{pwm} is very close to a delay of $\frac{1}{2}T_{pwm}$, the total time delay of case B is $\tau_{d,B} \approx \frac{3}{2}T_{s,B} + \frac{1}{2}T_{pwm} = \frac{5}{4}T_{pwm}$. Case B is expected to provide strong noise attenuation at the expense of a significantly reduced phase margin. Case C is the same as case B, only without the one step computation delay. An almost instantaneous control execution is possible using custom hardware platforms, e.g. FPGAs [3]. When using DSPs, this delay reduction can be achieved by re-scheduling the control interrupts just before the instants when the DPWM carrier is equal to its peak or valley (assuming a short enough computation time) [87,110]. The equivalent time delay of case C is $\tau_{d,C} \approx \frac{1}{2}T_{s,C} + \frac{1}{2}T_{pwm} = \frac{3}{4}T_{pwm}$. Case C is expected to provide strong noise attenuation while retaining similar dynamics to case A. Case D is MS-PWM control with $G_{MAF}(z)$ from (2.9). The equivalent time delay of case D is $\tau_{d,D} \approx \frac{3}{2}T_{s,D} + \frac{1}{2}T_{pwm} = \frac{\frac{3}{N}+1}{2}T_{pwm}$. As noted, all results are given for N = 16; however, it is interesting to point out that, starting from N = 6, the MS-PWM control with the MAF brings less delay than DS-PWM without any filters. Case D is expected to provide strong noise attenuation with better dynamics compared to the cases above. Finally, case E represents the MS-PWM control with $G_{DLPF}(z)$ from (5.11). This case is expected to provide strong noise attenuation with the best dynamic performance.

For all strategies, the PI current controller is used, with slightly higher gains compared to Section 5.6.2. All other parameters of the experimental set-up are the same as for the previously shown results. The summary of compared strategies and their parameters is shown in Table 5.2.

Noise power attenuation, relative to case A, is shown in Fig. 5.37. As consistency of the results for different operating points is demonstrated in Section 5.6.2, the output noise power is shown just for one duty cycle, D = 0.45.

From Fig. 5.37 it can be seen that, compared to DS-PWM, all examined strategies that use oversampling and digital filtering result in strong noise attenuation. The difference between cases B and C is due to a higher phase margin in case C, which brings a higher filtering by the closed-loop system. As

Description	label	value	unit
Controller configuration	$G_{c,i}$	PI	/
Relative proportional gain	$\frac{V_{in}}{R}k_{p,i}$	0.3064	/
Relative integral gain	$\frac{V_{in}}{R}k_{i,i}$	366	$\frac{1}{s}$
Crossover frequency	$f_{c,i}$	≈ 2400	Hz
DS-PWM group			
case A	label	value	unit
Filter configuration	/	/	/
One step computation delay	z^{-1}	yes	/
Phase margin	PM	54.2	0
MR-DS-PWM group - ${\cal N}=16$			
case B	label	value	unit
Filter configuration	$G_{MAF}(z)$	/	/
One step computation delay	z^{-1}	yes	/
Phase margin	\mathbf{PM}	33.9	0
case C	label	value	unit
Filter configuration	$G_{MAF}(z)$	/	/
One step computation delay	z^{-1}	no	/
Phase margin	\mathbf{PM}	54.9	0
MS-PWM group - ${\cal N}=16$			
case D	label	value	unit
Filter configuration	$G_{MAF}(z)$	/	/
One step computation delay	z^{-1}	yes	/
Phase margin	PM	62.8	0
case E	label	value	unit
Filter configuration	$G_{DLPF}(z)$	/	/
One step computation delay	z^{-1}	yes	/
Phase margin	\mathbf{PM}	75.7	0

Table 5.2: Control strategies tested in this section

expected, the results for cases C and D are very close as the same digital MAF is used and the phase margins are similar. Note that the phase margin for case D is affected by the one step computation delay, which is not present only for case C. Finally, case E provides nearly the same noise attenuation as cases C and D, while the dynamic properties are significantly improved. This is a confirmation that, in order to achieve high noise attenuation in current control loops, the MS-PWM control with the anti-aliasing $G_{DLPF}(z)$ is a good choice. It can be seen that both analytical models, with and without including the decimation, provide an excellent prediction of the output noise power. This is due to the fact that all tested cases feature digital filtering⁶ at $f > f_{pwm}$ that suppresses aliasing and brings noise propagation properties close to the ones of linear systems. These results further affirm that the extended model that

 $^{^{6}\}mathrm{Except}$ the DS-PWM, which does not feature decimation in any case.



Figure 5.37: Comparison of noise attenuation for the tested strategies. A: DS-PWM; B: MR-DS-PWM with $G_{MAF}(z)$ and one step computation delay; C: MR-DS-PWM with $G_{MAF}(z)$, without one step computation delay; D: MS-PWM with $G_{MAF}(z)$; E: MS-PWM with $G_{DLPF}(z)$.

includes the decimation is well suited for both MS-PWM and MR-DS-PWM.

5.7 Summary

This chapter has provided the analysis of noise propagation in digitally controlled multi-sampled power converters. First, it is concluded that the analytical model for LTI systems is able to predict-well noise propagation in case of (S/D)S-PWM. A limiting aspect for the case of MS-PWM is the nonlinear property of the DPWM, which closely resembles a decimator. Hence, for strong noise attenuation, antialiasing digital filters are required to suppress the decimation effect caused by the DPWM. Design of such filters is proposed, first using a qualitative approach, and it is shown that the single-stage loops with the high-frequency proportional-dominant controllers (P, PI, PR) can strongly suppress noise with a small impact on the dynamic response. Without any filters, these controllers result in a nearly constant output noise power, as N is increased above 2. Contrarily, when controllers include a derivative action, the oversampling on its own brings noise amplification. In such cases, for strong noise suppression, the required anti-aliasing digital filters introduce a higher impact on the system dynamics.

Additionally, this chapter brings an analogy, in terms of noise propagation, between MS-PWM and MR-(S/D)S-PWM, which both include decimation inside the closed-loop system. Models for noise propagation analysis in systems that feature decimation are derived and verified in simulations and experimentally. The proposed models can be used for the quantitative anti-aliasing digital filter design, for both cases of MS-PWM and MR-(S/D)S-PWM.

A very good match between the proposed models and the experimental measurements validates that, in terms of noise propagation, DPWM does introduce the decimation action. This brings an interesting conclusion that, in case of the oversampled US-PWM control systems, decimation cannot be avoided. Either it will be performed intentionally (MR-(S/D)S-PWM), or inherently (by the DPWM in case of MS-PWM). Hence, for noise suppression, suitable anti-aliasing filtering is always needed. In general, noise attenuation capabilities of these two control strategies are practically the same. An important difference is that, in case of MS-PWM, for similar filtering, significantly better dynamics can be achieved owing to the modulation delay reduction.

Chapter 6

Impedance shaping for robust stability of grid-connected VSCs

6.1 Introduction

Existing power systems are experiencing a great change due to the increasing presence of grid-tied power electronic converters. These devices are more and more necessary to fully exploit renewable energy resources, enable electrification of transportation, and the storage of electrical energy, among other functions [1,132]. Heavy presence of actively controlled voltage-source converters (VSCs), connected to ac distribution networks, rises some concern, as it is known that, due to the interaction between various grid-connected converters, harmonic instability may arise and compromise the system operation [133]. This may occur, for example, when control systems of grid-tied VSCs introduce negative damping at frequencies of poorly damped grid resonances. In the low frequency range, negative damping is introduced by the Phase-Locked Loop (PLL) of inverters and the constant power control of rectifiers [133,134]. In the high frequency range, negative damping is caused by the time delay and current control loop dynamics (in case of grid-following converters) [133,135]. This chapter focuses on the investigation of harmonic instability at high frequencies, caused by delays of the current control system [135].

For stability analysis of systems with interconnected power converters, the impedance based approach is found to be very effective [135–147]. It consists of representing each converter with its Norton or Thevenin equivalent circuit and assessing the system stability properties by analyzing the resulting admittance network. A sufficient, but not necessary, condition for the system stability is obtained if each admittance is passive, i.e. its real part is non-negative at the frequencies of interest [134, 135, 148–150]. For this reason, to prevent the harmonic instability, passivity-based control (PBC) has been more and more often considered [134, 135, 141, 148–164]. This concept has proved to be powerful for the analysis and the improvement of the stability of a system that consists of many interconnected power converters [133, 153, 165–167]. Initially, focus was placed on passivizing the admittance up to the Nyquist frequency (NF). However, more recently, it has been pointed out that passivity should be investigated also above the NF, where the coupling between sidebands, generated by sampling and modulation, introduces a destabilizing mechanism [141,143,144,168]. Based on these considerations, in some countries, the heavy impact of grid-tied VSCs on the traction system stability has resulted in the requirement to comply with input admittance passivity standards [152].

For US-PWM with one step computation delay, the VSC admittance exhibits a negative conductance above, approximately, one sixth of the sampling frequency f_s [135]. For enhancing passivity of VSCs operating with (S/D)S-PWM, many different strategies of active damping (AD) have been developed. Some improvement is obtained by adding damping filters to the current controller [153,169–171], however, most methods employ additional feedforward¹ or feedback actions [135,143,148–150,154–164,172]. While certain multi-loop methods are developed specifically for VSCs with *LCL* filters [157,159,161,163, 164, 172, 173], this chapter compares and refers to the most generally applicable ones, that are relevant also for VSCs with *L* filters² [135,143,148,149]. These methods rely on derivative-based feedforward of the voltage at the point of common coupling (PCC). In terms of effectiveness, for full passivity up to the NF, analog or oversampled derivative action must be applied. This kind of AD, first analyzed in [135] and generalized in [149], will be referred to as the Derivative Active Damping (D-AD). The passive region can also be extended without the analog or oversampled implementation of the derivative action, which is referred to as Discretized Derivative Active Damping (DD-AD) [143,148]. DD-AD effectively increases the passive region up to approximately $\frac{f_s}{s}$.

Given that high-frequency passivity issues arise due to digital delays, the investigation of MS-PWM performance in this application was found to be a natural choice. This has motivated the comparison between AD and MS-PWM in terms of passivizing the VSC admittance in the high-frequency range. The results found in this chapter are published in papers [70, 71, 73]. The conclusions of this research are reinforced, based on a later, independent source [43].

First contribution presented in this chapter deals with investigating the capability of MS-PWM control to render the admittance passive by inherently reducing, instead of compensating, digital delays and, in this way, approaching the NS-PWM. It is shown that the increase of the oversampling factor improves passivity, by boosting the admittance phase even above the switching frequency. In this way, passivity is achieved almost effortlessly and the design of the current control loop can be performed solely based on the required closed-loop tracking performance. Extensive experimental verification is provided to validate the small-signal analysis and to examine the impact of the MS-PWM nonlinearities investigated in Chapter 4. It is found that the impact of the MS-PWM nonlinearities on admittance passivity is not detrimental; just some mismatch from analytical modeling is seen, mainly for lower values

¹Note that the subsequently introduced active damping action based on the output voltage can only be considered as a feedforward in case the VSC is connected via an L filter to an ideal grid with zero impedance. Otherwise, the AD action should strictly be classified as a feedback. Given that analyzing interconnection stability only makes sense in case of a non-ideal grid, the feedforward active damping is an incorrect term, although widely spread in the literature. However, to avoid misinterpretation, the term feedforward will be used in this chapter.

 $^{^{2}}$ Regarding passivation effectiveness, conclusions do not differ between the two groups. Notable difference is the fact that in case of LCL filters, the capacitor current feedback can replace the output voltage derivative.

of oversampling factors. Importantly, it is found that operation in reduced-gain or zero-gain zones always brings the admittance closer to being fully passive, which is in agreement with the reasoning that for those nonlinearities, the modulator is "less active".

Second contribution is a direct comparison in effectiveness of D-AD, DD-AD, and MS-PWM in passivizing the admittance in the high-frequency range. Even though in [141] it is shown that this frequency range is important for stability, there is a lack of detailed investigation of corresponding AD effectiveness, which is addressed in this chapter. Admittance properties are analytically investigated and experimentally validated up to twice the switching frequency of the tested VSC. Results show that AD *does not* provide robust stability of grid-tied VSCs for harmonic resonances around and above the NF. This property is predictable by a linear single-frequency model of the VSC admittance. Experimental results are given to show destabilization of grid anti-resonances found slightly below the NF for DD-AD and above NF for D-AD. MS-PWM enables stable operation for all tested cases by passivizing the VSC in a wider frequency range.

Third contribution is the analytical modeling, experimental measurements, and comparison of noise sensitivity of D-AD, DD-AD, and MS-PWM. A potential practical challenge with D-AD and DD-AD concerns noise sensitivity due to implementation of the derivative feedforward action. Accordingly, this chapter provides investigation of noise propagation, which has not been analyzed in the literature yet. The proposed analytical models predict well the experimental measurements in case of both single-loop (MS-PWM) and double-loop (AD) control systems. Another step forward is taken by modeling the noise propagation for the case of oversampled (or analog) derivative being followed by a re-sampler, i.e. the case of D-AD. This analysis takes into account the impact of decimation on noise propagation, using models derived in Chapter 5. An important conclusion is that D-AD and DD-AD strategies *are not* highly sensitive to measurement noise, thanks to a relatively low required value of the derivative gain. For DD-AD, it is shown that, for all practical purposes, noise sensitivity is the same as without AD. Following the anti-aliasing digital filter design guideline from Chapter 5, MS-PWM is shown to be able to offer strong additional noise attenuation. It is verified that corresponding digital filters do not pose a significant impact on admittance properties in the frequency range of interest.

6.2 Control of grid-following VSCs

Grid-following VSCs are employed in power systems as current-controlled converters. As the most general case, this chapter considers a single-phase VSC with an inductive output filter, as shown in Fig. 6.1a. However, the analysis is easily extendible to three-phase VSCs in stationary coordinates as well as to any output filter structure. The current reference $i_{L,r}$ is provided by outer loops that control the required power flow by changing the reference current magnitude $|i_{L,r}|$ and phase angle Φ , with respect to the voltage at PCC, v_{pcc} . A PLL is used to track the fundamental component of v_{pcc} .

The closed-loop block diagram of the current control system is shown in Fig. 6.1b, where v_{pcc} is shown as a disturbance. The current controller (CC) $G_c(z)$ processes the error between $i_{L,r}$ and the



Figure 6.1: Illustration of a grid-following VSC: (a) schematic representing connection to a grid; (b) multirate current control system with a feedforward action. Black color depicts blocks executed at the current sampling rate. Red color depicts the feedforward execution rate. Continuous-time domain is shown in blue. Signals n_i and n_v represent measurement noise for current and voltage sensing, respectively.

filtered inductor current feedback $i_{L,f}$. In this chapter, the control execution and the feedback sampling are always performed with the same frequency, f_s . The complete control signal, i.e. the reference average switched node voltage $v_{x,r}$, may also include an action based on v_{pcc} , which will be referred to as AD, $H_{ad}(z_1)$. The execution rate of the AD block is independently chosen as $f_{s,1} = N_{ad}f_{pwm}$, where N_{ad} will be referred to as the AD oversampling factor. The Z-domain variable z_1 corresponds to $T_{s,1} = \frac{1}{f_{s,1}}$. In case $f_{s,1} \neq f_s$, a rate conversion needs to be included after H_{ad} . The reference voltage signal $v_{x,r}$ is scaled with V_{in} to obtain the digital modulating signal m_s , which is not illustrated in Fig. 6.1. It is assumed that, due to finite algorithm computation time, the controller output is delayed by one control period. The switching signal x, together with V_{in} , determines the VSC output switched node voltage v_x , as explained in Section 4.2.1. The difference between the VSC switched node voltage v_x and v_{pcc} is applied to the output inductive filter, which determines the plant transfer function as:

$$G_p(s) = \frac{i_L(s)}{v_x(s) - v_{pcc}(s)} = \frac{1}{sL}.$$
(6.1)

The analyzed system is modeled without any resistances, which represents the worst case regarding

stability, as any resistance positively impacts the system resonance damping [150].



6.2.1 Equivalent time delay and controller design

Figure 6.2: S-domain representation of a grid-following VSC: (a) current control system and grid connection; (b) Norton equivalent circuit used for impedance-based stability assessment.

In this chapter, modeling is performed in the s-domain to allow investigation of the VSC admittance properties, which determine the continuous-time physical response of the converter, even above the NF [141]. Impact of the sampling and modulation sidebands [73,141,143,144,167,168] is not considered, as it introduces additional complexity, without being crucial for the conclusions drawn in this chapter. The s-domain representation of the system is shown in Fig 6.2a. All modeled time delays are represented by a unique block $G_d(s)$.

As mentioned before, and seen from (2.3), digitally-controlled VSCs exhibit operating point dependency, which is often overlooked when using delay or ZOH models for the DPWM, due to a limited impact at low and medium frequencies [73]. In [73], it is shown that this operating point dependency may have a significant impact in the high frequency range. However, for the purpose of bringing some general conclusions based on simple and intuitive expressions, for most parts of this chapter, the simplified delay model from (2.5) is considered³. It will be pointed out when the delay model brings a significant mismatch from the actual VSC behavior and some considerations regarding the impact of the operating point dependency on admittance properties are placed in Section 6.4. Together with the added one step computation delay, the total time delay in the digital current control system is found as:

$$G_d(s) = \frac{v_x(s)}{v_{x,r}(s)} \approx e^{-s\tau_D} = e^{-s\frac{3}{2}T_s} = e^{-s\frac{3}{2}\frac{T_{pwm}}{N}}$$
(6.2)

As a standard approach for sinusoidal reference tracking [3], the proportional-resonant (PR) current

³For the tested buck-type VSC, scaling used to transform $v_{x,r}$ to m_s is inversely applied to obtain v_x from d (the continuous time duty cycle of x); hence, it can be omitted from the small-signal analysis, using the plant described in (6.1).

controller structure from (4.4) is chosen. As a design example used here, the proportional gain is chosen as $k_p = \omega_c L$, where $\omega_c = 2\pi f_c$ is the sought angular crossover frequency of the current loop. The resonant gain is chosen as $k_r = \frac{1}{10}\omega_c k_p$, which limits its impact to lower frequencies. In this chapter, the relative crossover frequency (bandwidth) is labeled $\alpha = f_{c,r} = \frac{\omega_c}{\omega_{pwm}}$. As discussed in Section 2.3, the value of α is limited by the small-signal stability margins and the PWM carrier frequency.

6.3 Impedance-based stability analysis of grid-tied VSCs

6.3.1 Norton equivalent circuit

The impedance-based stability assessment requires representing the VSC with its Norton equivalent circuit, as in Fig 6.2b. The Norton current source determines the short circuit current of the VSC, i.e. the value of i_L when $v_{pcc} = 0$. It is defined by the inner closed-loop transfer function, W_{cl} , of the system shown in Fig 6.2a:

$$W_{cl}(s) = \frac{i_L(s)}{i_{L,r}(s)} \bigg|_{v_{pcc}=0} = \frac{W_{ol}(s)}{1 + W_{ol}(s)},$$
(6.3)

where $W_{ol}(s) = G_c(s)G_d(s)G_p(s)G_{fb}(s)$ is the open-loop gain of the corresponding system. The input admittance Y_i is found as the inductor current response to the voltage v_{pcc} , when the current reference is set to 0:

$$Y_i(s) = -\frac{i_L(s)}{v_{pcc}(s)}\Big|_{i_{L,r}=0} = \frac{G_p(s)\left(1 - H_{ad}(s)G_d(s)\right)}{1 + W_{ol}(s)}.$$
(6.4)

From (6.4), it can be seen that the VSC input admittance is shaped by the output filter and the current control system, which includes digital delays and, if implemented, AD. Without the AD filter, at frequencies above the control bandwidth, where $|W_{ol}(s)| \approx 0$, Y_i converges to the physical filter admittance. The impact of outer loops on the admittance is not modeled, as this chapter focuses on properties at frequencies significantly above their bandwidth [134].

Considering the VSC connection to a non-ideal grid, i.e. such that $Z_g \neq 0$, the inductor current can be calculated using the equivalent circuit parameters as:

$$i_L(s) = \frac{W_{cl}(s)}{1 + Y_i(s)Z_g(s)} i_{L,r}(s) - \frac{Y_i(s)}{1 + Y_i(s)Z_g(s)} v_g(s).$$
(6.5)

Assuming that W_{cl} is stable, stability of the grid-connected VSC depends on the product $Y_i(s)Z_g(s)$, which is referred to as the minor-loop gain [137]. Stability analysis can be performed by applying the Nyquist stability criterion to the minor-loop gain⁴.

⁴The admittance Y_i in (6.4) is the so-called single-frequency model [73,141,143]. Impact of sampling and modulation sideband components can be taken into account using multi-frequency modeling [73,141,143]. It is important to note that the multi-frequency model is by itself non-linear; hence, using control tools derived for linear systems, such as the Nyquist stability criterion, is not strictly accurate. It will be shown that single-frequency Y_i models well the VSC behaviour in the frequency range of interest. Consequently, linear analysis will be used to predict stability of the grid-connected operation.

6.3.2 Passivity criterion

A linear single-input single-output (SISO) system, such as $Y_i(s)$, is said to be passive if the real part of its frequency response is non-negative for all frequencies [149]. In other words, $Y_i(s)$ is passive if its phase response is limited to the range $\left[-\frac{\pi}{2}, \frac{\pi}{2}\right]$. A stronger condition of "strict" passivity is obtained by limiting the phase to the open interval $\left(-\frac{\pi}{2}, \frac{\pi}{2}\right)$. In case both $Y_i(s)$ and $Z_g(s)$ are passive, the minor-loop gain always satisfies the Nyquist stability criterion as its phase is limited to the range $\left[-\pi, \pi\right]$. Imposing the requirement⁵ for passive behavior of $Y_i(s)$, as in [152], comes from the fact that a sufficient condition for the VSC stability is satisfied if $Y_i(s)$ is passive and the VSC is connected to another passive network.

In this chapter, the input feedforward passivity (IFP) index is used as a measure of the VSC passivity [149,150]. It is defined as the minimum value of the real part of the input admittance, i.e. the minimum value of the input conductance:

$$IFP(Y_i) = min_{\omega \in \mathbb{R}} Re\{Y_i(j\omega)\}.$$
(6.6)

The IFP is useful to quantify the impact a VSC may have on the grid stability, as it points to the minimal damping introduced by it. The IFP is negative in case of a non-passive input admittance.

Considering a system without the AD and the feedback filters, for a purely proportional current controller, the real part of $Y_i(j\omega)$ is calculated using (6.4):

$$Re\{Y_i(j\omega)\}|_{H_{ad}=0,G_{fb}=1,k_r=0} = \frac{\omega_c}{L} \frac{\cos\left(\omega\tau_D\right)}{\left(\omega_c - \omega\sin\left(\omega\tau_D\right)\right)^2 + \left(\omega\cos\left(\omega\tau_D\right)\right)^2}.$$
(6.7)

The expression is not given for the case of a PR controller, as it is more complicated and, hence, lacks a clear qualitative insight. Nonetheless, the resonant gain is designed such that its impact at high frequencies is very small, meaning that the IFP for a PR controller will be very similar to the one obtained with the proportional gain only [149]. It can be seen from (6.7) that, for the proportional controller, the sign of the conductance depends only on the cosine term, while its magnitude depends also on the bandwidth and is inversely proportional to the value of the inductance. The conductance first becomes negative when the argument of the cosine term goes over $\frac{\pi}{2}$. Considering the total time delay in (6.2), this occurs at the frequency $\frac{Nf_{pwm}}{6}$, which is often referred to as the critical frequency [149]. For the case of SS-PWM, the non-passive region is found above $\frac{f_{pwm}}{6}$, for DS-PWM above $\frac{f_{pwm}}{3}$, and so on. It can be concluded that, for the case of NS-PWM, without any other control delays, the admittance is inherently passive at high frequencies.

To illustrate the impact of system parameters on passivity, the IFP is found for the conductance from (6.7) and the results are shown in Fig. 6.3. The values are given for different multisampling factors and three values of α , corresponding to $\frac{\omega_c}{\omega_{pwm}} = \{5, 10, 20\}\%$. For N = 1, results are not shown for $\alpha = 0.2$ as

⁵In [152] the passivity requirement is set above 5th harmonic of the fundamental. Low frequency passivity issues, e.g. caused by dynamics of the PLL, are not investigated in this chapter. Since the PLL bandwidth is always set considerably lower than the one of the current loop, its impact on Y_i in medium- and high-frequency range is negligible [134]; hence it is not included in the admittance model.

such gain results in the instability of $W_{cl}(s)$ in (6.3). For better generality, the values on the left axis show the IFP relative to the VSC filter susceptance at the angular frequency ω_m , for which the IFP is found. In this way, the results become independent on the specific value of L. On the right axis, the values of ω_m are shown. Firstly, it can be seen that control loops with high bandwidths bring a significantly stronger negative damping, while the frequency at which the IFP is found remains practically constant. This yields the conclusion that, for a constant delay, the passivity issues are more likely to be found in high bandwidth systems. Secondly, as the value of N is increased, the relative IFP rapidly converges towards 0. Additionally, ω_m moves to higher frequencies meaning that the absolute IFP, i.e. negative conductance, reduces even more. Finally, it can be concluded that lower values of L result in a stronger negative damping. As mentioned, repeating the analysis for realistic values of k_r practically does not change the results shown in Fig. 6.3.



Figure 6.3: Left axis shows the relative passivity index of the VSC input admittance for a proportional controller and three different values of the relative crossover frequency α . Right axis shows frequency for which the IFP is found, relative to the switching frequency.

There is some ambiguity regarding the interpretation of standards like [152] to determine the upper frequency limit for passivity measurements. Several publications have shown that, for single- and doubleupdate PWM, investigation is necessary also above the Nyquist frequency due to the sampling and modulation sideband components [141,143]. In [148], authors conclude that if an LCL filter is used, even in the case the inner current loop is non-passive, high-frequency components on the grid side flow mainly through the capacitor; hence the non-passive high-frequency region of the inner current loop gets compensated and passivity should only be addressed around the bandwidth. However, it should be noted that the existence of an LCL filter can only reduce, and not completely remove, the non-passive behavior. This is due to the fact that a parallel connection of a non-passive element and a reactive element remains non-passive [149]. A generalized conclusion is hard to be brought, as it depends on the control bandwidth, passive component sizing, and losses in the circuit.

As for the MS-PWM, the results shown in Fig. 6.3 predict the existence of very small negative conductances at frequencies significantly above the switching frequency. However, this comes from the

small-signal modeling that does not take into account the PWM large-signal behavior. Namely, a twolevel converter still physically reacts to disturbances only twice per switching period, by modifying rising and falling edges of x(t). This brings doubts that a VSC can be active at, for instance, $f = 4f_{pwm}$. Additionally, for high values of N, the small-signal analysis predicts the existence of negative conductances with values very close to 0; hence, even very small resistances would be enough to compensate and make the experimental measurements of non-passive zones impossible. In the end, given that, at frequencies significantly above the bandwidth, the admittance magnitude is determined by the physical VSC output filter, its low-pass nature would require extremely high values of perturbation voltages to cause the resulting current to even be detectable by the ADC. Therefore, the subsequent admittance measurements are performed up to $2f_{pwm}$, which corresponds to the sampling frequency for DS-PWM.

6.4 Analysis and experimental verification of admittance properties

In this section, the VSC admittance is measured for all compared control strategies. First, the admittance measurements are given for the case of DS-PWM without AD, to serve as a benchmark. Then, it is shown that MS-PWM is capable of rendering the admittance passive in a wide frequency range, without any active damping. The results are given for values of $N \in \{4, 8, 16, 32\}$. Finally, the impact of AD on admittance properties is examined, and a comparison between all tested strategies is presented.

For the experimental verification, the set-up B.3 is used. It is illustrated in Fig. 3.1c and its parameters are shown in Table 3.3. Two inductive filters with different values of L are used to demonstrate the impact of the inductance on the IFP, as seen in Fig. 6.3. The control system is implemented on the FPGA platform described in Section 3.2. For all tested cases, the control system features one step computation delay. The feedback filter G_{fb} is not used, except for two tests where this is clearly indicated. For the digital implementation of the PR controller, the impulse-invariant structure from (4.31) is used.

6.4.1 Measurement set-up

Direct admittance measurements of current-controlled converters are typically performed by setting the current reference to 0, which disconnects the Norton current source shown in Fig. 6.2b [143,148]. The perturbation voltage is applied to the converter's output filter and the measured current determines the admittance response at the same frequency. As mentioned in Section 6.3.1, even if a PLL is used during the admittance measurements, its bandwidth must be set well-below the current loop bandwidth [139]; hence, its impact can always be considered negligible for the examined high-frequency passivity properties [134].

The circuit used for the admittance measurements, shown in Fig 6.4, consists of the tested VSC with its inductive output filter, the perturbation branch, and the bias branch. For the first set of results, the perturbation source v_p was formed using the power operational amplifier PA107DP, from APEX.



Figure 6.4: Illustration of the set-up for admittance measurements.

Due to the current limit of PA107DP, the corresponding admittance measurements were conducted for a reduced value of $V_{in} = 250$ V and for the higher-value inductance L = 2.5 mH, so that the switching ripple in i_L is decreased. The value of V_{in} does not impact the small-signal admittance, as seen from (6.4). Subsequently, PA107DP was replaced by MP118, also from APEX, which allowed measurements with L = 1.5 mH as well. The perturbation source is connected in series with a capacitor $C_p = 10$ μ F and a resistor $R_p = 1 \Omega$ that is used to smooth any transients. The capacitor is used to block dc currents as well as to decouple the bias branch from the VSC, so that the switching ripple harmonics of i_L would remain practically the same as if the VSC was connected to an ideal grid. In this way, the ripple-induced MS-PWM nonlinearites, analyzed in Chapter 4, are not altered by the measurement setup. The bias branch features an inductance $L_b = 2.4$ mH, which attenuates the perturbation components so that the perturbation source is not heavily loaded, and a resistor R_b that can be bypassed with a switch s_b . The bias branch has two functions. First of all, it allows the circulation of any dc current. All admittance measurements are performed for an imposed dc current reference, such that no current crossover occurs. This scenario is more critical for passivity as it is reported that dead-times may provide a certain damping when the current crosses 0 [174]. Note that a dc component of i_L does not impact the admittance measurements in the high frequency range. Secondly, if the switch SW is in its left position, when the switch s_b is opened, a dc component of i_L provides a voltage drop across R_b . This creates a dc component of v_{pcc} , referred to as the bias voltage v_b , without needing to use an additional dc power supply. By imposing an adequate bias voltage, the admittance can be measured around specific operating

points. Alternatively, in case the switch SW is in its right position, the admittance measurements can be obtained while the VSC is directly connected to an ac power supply, e.g. the power grid. As the DPWM exhibits several nonlinearities and, as seen from (2.3), operating point dependency, investigation of small-signal properties of the system makes sense to be performed only around stable and constant operating points. When the converter operates connected to an ac grid, its operating point moves in a certain range and the measured admittance is expected to be some sort of an average of the results obtained for steady operating point in the same range. Detailed investigation and modeling of the impact of the DPWM's operating point dependency on the admittance measurements can be read in [73]. Measurements around specific, critical, operating points also allow investigation of the impact of the discontinuity-related MS-PWM nonlinearities on admittance properties. The admittances are measured both without and for three positive values of the voltage bias. For the bias values of 65 V and 75 V, the resistance is set to $R_b = 24 \ \Omega$. For a 150 V bias, the resistance is set to $R_b = 48 \ \Omega$. These bias values determine the steady-state operating point as $D = 0.5(\frac{v_b}{V_{in}} + 1)$, if series voltage drops are neglected.

The first set of measurements, using PA107DP and for L = 2.5 mH, was conducted for 17 frequencies, one at the time, in the range from 5 kHz to 31.5 kHz $(0.25 f_{pwm})$ to $1.575 f_{pwm}$), avoiding the switching frequency. When PA107DP was replaced with MP118, the corresponding measurements for L = 1.5mH were conducted in an extended frequency range from 5 kHz to 41 kHz, which is just above f_s for N = 2. Qualitatively, the results did not change, however, to discriminate between these two sets of measurements, data points obtained using PA107DP are plotted using dashed line with markers, while the ones obtained using MP118 are plotted using only circular markers. For each point, a sinusoidal v_p , with the perturbation frequency f_p , is imposed. Its magnitude is calculated to obtain at least 100 mA magnitude of the perturbation current component flowing through the VSC. This allowed precise measurements using the oscilloscope, in the presence of the switching ripple and noise. The inductor current i_L and the PCC voltage v_{pcc} are acquired using the Tektronix 5 series oscilloscope with the data length equal to 40 ms and the acquisition rate equal to 250 MS/s. This yields a spectral resolution, relative to the perturbation frequency, of 0.5 % for $f_p = 5$ kHz, which further improves as f_p is increased. The data is postprocessed in MATLAB, by calculating the FFT of $-i_L$ and v_{pcc} . Based on the magnitude ratio and the phase difference of the perturbation components, the admittance at f_p is calculated. The measurements are performed for the relative crossover frequency $\alpha = 0.1$ and $\alpha = 0.2$. The results are compared with the frequency response of the analytical model from (6.4). For MS-PWM, in the frequency range of interest, it was verified that the analytical model of Y_i does not vary significantly in case the complete DPWM model from (2.3) is included; hence, the delay model from (2.5) is used for the analytical traces. Note that, for DS-PWM and D = 0.5, the DPWM model is exactly equal to the delay from (2.5).

6.4.2 Benchmark results for DS-PWM without AD

Let us begin by showing the impact of delays on admittance properties in case no passivizing strategy is used, i.e. for DS-PWM without AD. The results obtained without applying the output bias voltage (D = 0.5) are shown in Fig. 6.5, for different values of α and L, to emphasize their impact. First of all, it can be seen that a very good match with the analytical prediction is obtained, even above the NF. Approximately at frequencies $f > 1.5 f_{pwm}$, the single-frequency model (6.4) starts to show a more significant mismatch, which is addressed in [73]. Note the existence of two non-passive zones, where the phase drops below -90° . The first one starts at approximately 6.5 kHz, which is consistent with the often mentioned critical frequency $\frac{f_s}{6}$ [135]. Another non-passive zone appears above the NF. The goal of the passivity-based design, which is analyzed in the following sections, is to suppress these non-passive zones.



Figure 6.5: Admittance measurements for DS-PWM without AD: N = 2, $H_{ad} = 0$, and: (a) L = 2.5 mH and $\alpha = 0.1$; (b) L = 1.5 mH and $\alpha = 0.2$. The vertical line marks the switching (and the Nyquist) frequency. The results are measured without the output bias voltage.

An alternative representation, useful for bringing conclusions regarding passivity, is to show only the real part of Y_i , i.e. its conductance. For admittances in Fig. 6.5, conductances are shown in Fig. 6.6, with values being scaled by the nominal VSC admittance $Y_n = \frac{S_n}{v_{pcc,RMS}^2} = 56.7$ mS.



Figure 6.6: Relative conductance for DS-PWM without AD: N = 2, $H_{ad} = 0$, and: (a) L = 2.5 mH and $\alpha = 0.1$; (b)L = 1.5 mH and $\alpha = 0.2$. The vertical line marks the switching (and the Nyquist) frequency. The results are measured without the output bias voltage.

6.4.3 Results for MS-PWM without AD

In this section, the impact of MS-PWM on admittance properties is investigated. Specifically, attention is dedicated to examine the impact of the MS-PWM nonlinearities on the passivation effectiveness. All results are given for L = 2.5 mH and $\alpha = 0.1$ as, at the time of these measurements, only PA107DP was available. The results were later repeated for the other values of L and α and the conclusions did not qualitatively change.

The results for N = 4 are given in Fig. 6.7, without the voltage bias and for the bias levels of 75 V



Figure 6.7: Admittance measurements for MS-PWM without AD: N = 4, $H_{ad} = 0$, L = 2.5 mH, $\alpha = 0.1$, and 3 dc values of the output bias voltage. Without the dc bias, vertical crossings between m and w occur. The vertical line marks the switching frequency.



Figure 6.8: Admittance measurements for MS-PWM without AD: N = 4, $H_{ad} = 0$, L = 2.5 mH, $\alpha = 0.1$, without the output bias voltage. MS-PWM is linearized such that no vertical crossings between m and w occur. The vertical line marks the switching frequency.

and 150 V. Without the bias voltage, duty cycle is perturbed around a steady-state value of D = 0.5. It can be seen that, for that case, the measurements show a slight mismatch from the analytical results, particularly in the range below 10 kHz. It is verified that, for this operating point, one edge of x is determined by a vertical crossing between m and w, which results in the reduced-gain operation, as discussed in Chapter 4. It is interesting to note that, at higher frequencies, this nonlinearity does not cause the admittance to go deeper in the non-passive zone; on the contrary, it seems that this operation adds some damping to the system. The same effect is noticed for other values of N under reduced-gain operation. Moreover, for operation inside the dead-bands (double vertical crossings, see Chapter 4), it is verified that the VSC is completely passive as its admittance phase is locked at -90° . In Fig. 6.7, when a dc bias is added, the operating point is such that no vertical crossings occur, and a good match with the analytical results is seen. Around 12 kHz, the admittance enters the non-passive zone, however, this zone is much less emphasized compared to N = 2.

It is of interest to verify whether the mismatch for N = 4 without the bias results from the discontinuity-related MS-PWM nonlinearity. For this, the MS-PWM behavior is linearized, based on the approach from Chapter 4. As shown in Section 4.5.3.4, for N = 4, the vertical crossings caused by the modulating waveform discontinuities can be completely prevented by adding a feedback low-pass filter with the suitable cut-off frequency. For the specific case, G_{fb} from (4.21) is used, with the cut-off frequency $\omega_{c,f} = 39$ kHz. The role of G_{fb} is solely on shifting the switching ripple component of m with respect to the carrier, in order to disable vertical crossings, and its impact on dynamics is negligible. The corresponding results are shown in Fig. 6.8, and it can be seen that a good match with the analytical predictions is once again obtained. This confirms that the effect observed in Fig. 6.7 is due to the MS-PWM nonlinearities and not the small-signal modeling errors.



Figure 6.9: Admittance measurements for MS-PWM without AD: N = 8, $H_{ad} = 0$, L = 2.5 mH, $\alpha = 0.1$, and 3 dc values of the output bias voltage. Without dc bias, vertical crossings between m and w occur. The vertical line marks the switching frequency.

The results for N = 8 are given in Fig. 6.9 and similar conclusions as for N = 4 are brought. Analysis predicts that the non-passive zone occurs around 26 kHz, with the phase of Y_i being just slightly under $-\frac{\pi}{2}$. As for N = 4, without the output bias, the reduced-gain operation is present and a mismatch from the analytical predictions is seen. Again, this mismatch is more evident at lower frequencies, and no detrimental impact on passivity occurs. Note that some drop in the phase is seen with 150 V bias at frequencies very close to the switching frequency. This effect is also measured for other oversampling factors, as well as for DS-PWM, and is most-likely caused by some other nonlinearity in the system (e.g. related to the sampling and modulation sidebands).

The results for N = 16 are given in Fig. 6.10. For the 75 V bias, the reduced-gain operation is present. The results are also shown for the 65 V bias, which slightly moves the operating point, enough

to prevent the vertical crossings between m and w, yielding again a very good match with the analytical results. The measurements show that, for all tested operating points, the admittance does not enter the non-passive zone. Moreover, the admittance is strictly passive, which indicates additional robustness.



Figure 6.10: Admittance measurements for MS-PWM without AD: N = 16, $H_{ad} = 0$, L = 2.5 mH, $\alpha = 0.1$, and 4 dc values of the output bias voltage. For 75 V bias, vertical crossings between m and w occur. Word "experimental" is omitted from the legend to fit the figure. The vertical line marks the switching frequency.

The results for N = 32 are given in Fig. 6.11. For all operating points, the match with the analytical results is excellent and the admittance is strictly passive.



Figure 6.11: Admittance measurements for MS-PWM without AD: N = 32, $H_{ad} = 0$, L = 2.5 mH, $\alpha = 0.1$, and 3 dc values of the output bias voltage. The vertical line marks the switching frequency.

For a comparison, Fig. 6.12 shows real parts of the measured admittances for all tested values of N. A clear improvement of passivity is obtained as N is increased. The results are given for the operating points that do not feature vertical crossings between m and w.

The results of admittance measurements, performed while the VSC was connected to a sinusoidal



Figure 6.12: Comparison of experimentally measured conductances for all tested values of N, L = 2.5 mH, and $\alpha = 0.1$. The results are given for operating points that do not feature vertical crossings between m and w. The vertical line marks the switching frequency.

ac source with the frequency f_1 , are not shown. This kind of set-up perturbs the operating point with the frequency f_1 , which is not consistent with the small-signal analysis of an inherently nonlinear system. However, it should be noted that both simulations and measurements were performed for such case, as required by the standard [152], and the impact of the MS-PWM nonlinearities was not seen; the measurements still matched well with the analytical modeling and all passivity-related conclusions remained valid. The results were somewhere in between the results for the tested dc operating points that did not feature vertical crossings between m and w. This leads to a conclusion that, regarding passivity, effects of the MS-PWM nonlinearities, which arise at specific operating points, may be masked by the ac operation. However, a general conclusion for any fundamental frequency and perturbation magnitude cannot be brought. Detailed investigation of impact of such measurement scenario is analyzed in [73], but is not further discussed in this thesis.

It should be stated that the experimental results show a very good match with the analytical ones, with errors in the range of a few degrees. This affirms the validity of the used single-frequency modeling. In [143], where a more complex multi-frequency modeling procedure is presented, some mismatch is present both in phase and magnitude. Experimental measurements in [143] are performed up to the switching frequency, for the case of SS-PWM. In [148], measurements are performed up to slightly above the Nyquist frequency, for SS-PWM, and a higher mismatch from the analytical modeling is seen compared to the results presented above. This can be partly due to the use of Z-domain modeling in [148], which fails around the Nyquist frequency.

To conclude, the results of this section have shown that MS-PWM is capable of rendering the admittance strictly passive in a wide frequency range, which, incidentally, provides compliance with the standard [152] without any active damping. Some impact of the MS-PWM nonlinearities is present; however, its nature is not found to be detrimental for passivity properties.

6.4.4 Results for DS-PWM with AD

In this section, active damping strategies are introduced, and the corresponding conductance measurements are given for L = 1.5 mH and $\alpha = 0.2$.

The first method considered for comparison is D-AD [135], which extends passivity up to the NF by adding a derivative feedforward of v_{pcc} to the controller output:

$$H_{ad}(s) = sk_{ad}.\tag{6.8}$$

The AD gain design is based on the procedure from [135]. Calculating the real part of (6.4) with H_{ad} from (6.8), for a proportional controller and without the feedback filter, yields the same expression as (6.7) multiplied by $\left(1 - k_{ad} \frac{\omega^2}{\omega_c}\right)$. The active damping gain k_{ad} is chosen such that this added term changes its sign at the same frequency as the cosine term in (6.7), i.e.:

$$k_{ad} = \frac{4\tau_D^2}{\pi^2}\omega_c.$$
(6.9)



Figure 6.13: Relative conductance for DS-PWM with D-AD: N = 2, $N_{ad} = 16$, $H_{ad} = \frac{k_{ad}}{T_{s,1}}(1 - z_1^{-1})$, L = 1.5 mH, and $\alpha = 0.2$. Orange dashed line shows the impact of including the additional filter H_{bq} from (6.10) in cascade with the derivative gain [135]. The vertical line marks the switching (and the Nyquist) frequency. The results are measured without the output bias voltage.

For all subsequent experimental tests employing AD, the gain k_{ad} is calculated using (6.9). In [135], it is shown that, for passivity up to the NF, the derivative action must be implemented in the analog domain, or using oversampling. The latter approach is chosen here, in order to keep the implementation fully digital. Backward-Euler discretization is chosen, as advised in [135]. Referring to Fig. 6.1b, D-AD is implemented with N = 2, $N_{ad} = 16$, $H_{ad}(z_1) = k_{ad}f_{s,1}(1 - z_1^{-1})$. The AD oversampling factor was chosen to be as high as allowed by the control platform.

The relative value of the measured conductance for D-AD is shown in Fig. 6.13. A very good match with the s-domain analytical results is obtained, apart for frequencies very close to f_s . It is supposed

that this mismatch comes from not including the modulation and sampling sideband effects [73]. The results in Fig. 6.13 confirm that D-AD is capable of improving passivity up to the NF⁶. It can be seen that, above the NF, a non-passive zone is introduced with a peak of negative conductance being slightly higher than for the case of DS-PWM without AD (see Fig. 6.6b). In [155], this effect is mentioned, however it is not thoroughly investigated in the literature, even if, from [141], it is clear that admittance properties above the NF need to be examined.

D-AD (as well as M-D-AD) was also tested for other values of the derivative gain (by multiplying k_{ad} from (6.9) with a factor K) and the corresponding analytical results are shown in Fig. 6.14. For reference, the M-D-AD filter H_{bq} is designed in s-domain based on the approach from [135]:

$$H_{ad}^{\rm M-D-AD}(s) = sKk_{ad} \cdot H_{bq}(s) = sKk_{ad} \cdot \frac{s^2 + 2\xi_z \omega_z s + \omega_z^2}{s^2 + 2\xi_p \omega_p s + \omega_p^2},$$
(6.10)

where $\omega_p = 0.1 \cdot 2\pi f_s$, $\omega_z = 0.2 \cdot 2\pi f_s$, $\xi_p = 2$, $\xi_z = \frac{\omega_p ((2\pi f_s)^2 - 4\omega_z^2)}{\omega_z ((2\pi f_s)^2 - 4\omega_p^2)}$. For both D-AD and M-D-AD, the trend is that higher values of k_{ad} add more positive damping in the frequency region from $\frac{f_s}{6}$ to the NF and more negative damping in the region above the NF, where the peak negative conductance is found. Lower values of k_{ad} decrease the peak negative conductance above the NF; however, passivity is not achieved around $\frac{f_s}{6}$. From the presented results, it can be concluded that for some applications it may be favorable to choose a lower value of k_{ad} , in order to suppress the emphasized negative peak above the NF, which, in the following section, will be shown to bring stability issues. Additional considerations regarding AD design are out-of-scope of this thesis and are left for future research.



Figure 6.14: Impact of the AD derivative gain on the conductance for DS-PWM with M-D-AD, for L = 1.5 mH and $\alpha = 0.2$. The vertical line marks the switching (and the Nyquist) frequency.

⁶In [135], an additional biquad filter H_{bq} is added in cascade with the derivative action, which is referred to as the modified active damping (M-D-AD). This filter adds damping to the narrow non-passive zone around $\frac{f_s}{6}$ that remains when a purely derivative H_{ad} is used. It is not included for the experimental comparison in this chapter as it lacks a standardized design procedure [149] and its impact above the NF, where robustness of D-AD is tested, is negligible. However, it is important to note that M-D-AD can fully passivize the admittance below the NF. The analytical trace corresponding to M-D-AD is shown in Fig. 6.13 using the orange dashed line.

Second AD method compared, DD-AD, uses the same derivative action as D-AD, however, without the oversampling: N = 2, $N_{ad} = 2$, $H_{ad}(z) = k_{ad}f_s(1 - z^{-1})$. DD-AD is analyzed in [148], with a specific parameter design procedure. However, it was verified that implementing the derivative gain as (6.9) brings practically the same results, without needing to use one fixed value of bandwidth as in [148]. In [148], Z-domain modeling was used to predict the impact of DD-AD on the admittance, which yielded a significant mismatch as frequencies approached the NF. Hence, here, H_{ad} is directly transformed to s-domain, using $z = e^{sT_s}$. The results are reported in Fig. 6.15. It can be seen that a good match between the experiments and the analytical results is obtained, in the same frequency range as in Fig. 6.6 and Fig. 6.13. From Fig. 6.15, it can be seen that DD-AD improves passivity at medium frequencies; however, above approximately $\frac{f_s}{3}$, the passivity is lost. The same frequency limit is reported in papers that use single-loop AD, by adding a derivative gain to the controller structure [153]. Note that the negative conductance peak for DD-AD is slightly reduced compared to the case of DS-PWM without AD.



Figure 6.15: Relative conductance for DS-PWM with DD-AD: N = 2, $N_{ad} = 2$, $H_{ad} = \frac{k_{ad}}{T_{s,1}}(1 - z_1^{-1})$, L = 1.5 mH, and $\alpha = 0.2$. The vertical line marks the switching (and the Nyquist) frequency. The results are measured without the output bias voltage.

In [73], admittance properties for DS-PWM without AD are examined for many steady-state values of D. It is found that, as D approaches extreme values, the negative conductance peaks become less emphasized, which can be predicted by including the full DPWM model from (2.3). To illustrate this, Fig. 6.16 is given to show the impact of D for DS-PWM with and without AD. It can be seen that the negative conductance peaks are highest for D = 0.5. The duty cycle dependence is most evident for the case of D-AD. Based on these results it can be concluded that, for the sinusoidal ac operation, the highest introduced negative damping is expected around the zero crossings of v_g , where D = 0.5. This is validated in time-domain experimental tests shown in Section 6.5.2.2 (see Fig. 6.30i).



Figure 6.16: Comparison of the operating point impact on the conductance for DS-PWM with and without AD, for L = 1.5 mH and $\alpha = 0.2$. The vertical line marks the switching (and the Nyquist) frequency.



Figure 6.17: Comparison of conductances for all tested control strategies: L = 1.5 mH, $\alpha = 0.2$. The vertical line marks the switching frequency. The results are measured without the output bias voltage.

6.4.5 Comparison of effectiveness

In Fig. 6.17, conductances for all tested cases are shown in a wider frequency range. It can be seen that both D-AD and DD-AD are able to extend the passive zone; however, for both methods, another emphasized non-passive zone is introduced. This points to the fact that the addition of AD only shifts the non-passive zone to higher frequencies and can, therefore, compromise stability, instead of guaranteeing it. It is important to note that these non-passive zones are found in the frequency range where the single-frequency admittance models predict well the experimentally-measured VSC responses. This is important as it can be assumed that assessing stability using the Nyquist criterion most-likely yields correct conclusions, which is validated in the following section. MS-PWM shows a natural drop towards zero conductance, where Y_i starts to behave as an inductor. A summary of the measured conductances



Figure 6.18: Impedance network at the frequency of the grid anti-resonance. The VSC admittance is represented as $Y_i(j\omega_{res}) = G_i(j\omega_{res}) + jB_i(j\omega_{res})$. Grid impedance features passive damping, represented by G_g .

is shown in Table 6.1, by providing peaks of negative values and frequencies for which they are found. The results are given for $\alpha = 0.2$ and both tested inductances. For MS-PWM and N = 16, the analytical model predicts the non-passive zone to fall outside of the measurement range.

Table 6.1: Comparison of conductance minimums and frequencies for which they are found: $min\{R_e\{Y_i\}\}$ [%] @ f_{min} [kHz]. The results are given for the relative bandwidth $\alpha = 0.2$.

PWM		DS-PWM		MS-PWM
AD	w/o AD	D-AD	DD-AD	w/o AD
L = 1.5 mH measured analytical	-12.2 @ 8 -13.4 @ 7.8	-14.1 @ 27 -16.2 @ 27.9	-8.4 @ 18.5 -9.3 @ 18.9	-0.2 @ 36 -0.1 @ 81
L = 2.5 mH measured analytical	-8.2 @ 8 -8 @ 7.8	-7.9 @ 27 -9.7 @ 27.9	-5 @ 18.5 -5.6 @ 18.9	-0.1 @ 36 -0.05 @ 81

6.5 Passivity implications in grid-connected scenarios

Impedance-based analysis from Section 6.3 predicts that a VSC can be destabilized if connected to a grid with a certain Z_g . For grid-following VSCs with inductive filters, a parallel anti-resonance that coincides with the non-passive zone of Y_i can cause instability [135]. In this section, operation stability is tested for all strategies compared in Section 6.4, by connecting the same VSC to a grid with a specific Z_g . Results presented in this section hold, in relative terms, also for VSC designs with different switching frequencies. Therefore, irrespective of the specific design parameters, reliable conclusions can be drawn on the robustness ensured by each controller organization.



Figure 6.19: Nyquist plots of the minor-loop gain $Y_i(j\omega)Z_g(j\omega)$ for three different grid anti-resonant networks and all compared control strategies. All results are given for L = 1.5 mH and $\alpha = 0.2$. The parameters of Z_g correspond to the ones experimentally tested.

6.5.1 Analysis of the equivalent impedance network

Let us suppose that the analyzed VSC is connected to a grid with an anti-resonance formed by L_g and $C_g^{\ 7}$. Grid passive damping can be modeled as a parallel conductance G_g . Such grid impedance can, therefore, be described using its anti-resonant frequency $f_{res} = \frac{\omega_{res}}{2\pi} = \frac{1}{2\pi\sqrt{L_g}C_g}$, characteristic impedance $Z_c = \sqrt{\frac{L_g}{C_g}}$, and quality factor $Q = \frac{1}{Z_c G_g}$. Considering passive damping is important, as a non-passive Y_i does not necessarily destabilize a grid anti-resonance with a realistic value of Q [70].

In this section, VSC connection to three different grid anti-resonant networks is tested. The antiresonances that coincide with the non-passive admittance zones are formed using an inductor $L_g = 72$ μ H and different parallel capacitors C_g . Passive damping is estimated using the procedure described in the following section. The first anti-resonance, corresponding to $Z_{g,1}$, is obtained with $C_g = 5.6 \ \mu$ F. It is very close to the negative conductance peak for N = 2 without AD and its parameters are $f_{res} = 7.9$ kHz, $Q = 28, Z_c = 3.6 \ \Omega$, and $G_g^{\%} = 17.6 \ \%$. The second anti-resonance, corresponding to $Z_{g,2}$, is obtained with $C_g = 1 \ \mu$ F. It is very close to the negative conductance peak for DD-AD and its parameters are $f_{res} = 18.4 \ \text{kHz}, Q = 38, Z_c = 8.3 \ \Omega$, and $G_g^{\%} = 5.6 \ \%$. The third anti-resonance, corresponding to $Z_{g,3}$, is obtained with $C_g = 440 \ n$ F. It is very close to the negative conductance peak for D-AD and its parameters are $f_{res} = 28.2 \ \text{kHz}, Q = 30, Z_c = 12.7 \ \Omega, G_g^{\%} = 4.6 \ \%$.

The equivalent impedance network at the frequency of the grid anti-resonance is illustrated in Fig. 6.18. At ω_{res} , the VSC input admittance is represented by its conductance $G_i(j\omega_{res})$ and susceptance $B_i(j\omega_{res})$. Intuitively, from Fig. 6.18, it can be concluded that the overall system damping at ω_{res} is impacted by the parallel connection of $G_i(j\omega_{res})$ and G_g . Hence, the value of $G_i(j\omega_{res}) + G_g$ indicates the damping of the anti-resonant network. This intuitive approach is valuable as it allows to predict the stability properties based solely on the admittance measurements, without knowing the analytical expression for Y_i . The formal approach, in case Y_i is analytically known, is to analyze stability using the Nyquist plots of the minor-loop gain. Examples of Nyquist plots, with the analytical and measured data points, are shown in Fig. 6.19, for all tested control strategies and grid impedances, for L = 1.5 mH and $\alpha = 0.2$. From Fig. 6.19a, it is clear that, although the anti-resonance $Z_{g,1}$ coincides with the non-passive zone of Y_i for DS-PWM without AD, passive damping of $Z_{g,1}$ is high enough to prevent instability. This can be also concluded from the fact that, for that case, $G_i(j\omega_{res}) + G_g > 0$. The Nyquist plots in Figs. 6.19b and 6.19c predict instability for DD-AD and D-AD, respectively.

6.5.1.1 Experimental measurement of anti-resonant circuit parameters

The grid voltage is formed using Chroma 6460 programmable ac source. Due to the unknown highfrequency damping properties of the ac source, it was of interest to decouple it from the anti-resonant circuit. This was achieved by connecting Chroma via an LC filter with the inductance $L_{ch} = 324 \ \mu\text{H}$ and the capacitance $C_{ch} = 150 \ \mu\text{F}$. In this way, an almost ideal ac source, at the frequencies of interest,

⁷Note that the specific values of L_g and C_g should not be understood as parts of a properly designed LCL filter, but as anti-resonance forming components that may be found due to, for example, the presence of grid-forming VSCs [162].

was formed. With such a configuration of the reactive elements, two anti-resonant frequencies appear in the circuit. The first one, due to decoupling of the ac source, is found close to 650 Hz and it has no direct implications on the subsequent stability tests.

Passive damping of each of the three tested anti-resonant networks was measured in two ways. First, an impedance analyzer was used to directly measure the network parameters, without connecting the VSC and the ac voltage source. For the anti-resonant networks $Z_{g,1}$, $Z_{g,2}$, and $Z_{g,3}$, the measured values of Q were found equal to {44, 60, 72}, respectively. The resulting values of Q factors seemed unrealistically high, and their implementation in the simulation model did not bring a good match with the subsequently shown experimental results. It is assumed that this was due to the low magnitude of the imposed perturbation, which was not able to stimulate certain loss mechanisms in the circuit.

For this reason, Q factors were estimated by triggering an open-loop transient, while the VSC was disconnected. This kind of approach was taken in order to consider also the impact of the ac voltage power supply, due to its unknown damping properties (i.e. its internal impedance Z_{ch}). The schematic of the relevant set-up is shown in Fig. 6.20a. The Chroma voltage source was set to output a sinusoidal voltage with 20 V magnitude and 50 Hz. The resistor $R = 20 \ \Omega$ was placed in series with the switch sw. Near the peak of the voltage v_{pcc} , the switch was turned off, which resulted in a transient due to the current i_R becoming equal to 0. Correspondingly, v_{pcc} was acquired using the oscilloscope and plotted in MATLAB, in order to estimate damping of the triggered oscillations. Passive damping, i.e. the Q factor, of the anti-resonant networks was calculated based on the measured time constant of the exponential decay as:

$$Q = \frac{\pi f_{res} \tau}{2},\tag{6.11}$$

where τ is the time needed for the oscillation magnitude to decay by the factor of e. An example of the measurement procedure for $Z_{g,2}$ is shown in Fig. 6.20b. In Fig. 6.20b, a lower-frequency oscillation is visible, which appears to be due to the above-mentioned Chroma decoupling LC filter. The 50 Hz component is removed from the voltage waveform, for a better visualization of the decayed oscillatory response. For the tested anti-resonant networks $Z_{g,1}$, $Z_{g,2}$, and $Z_{g,3}$, the estimated values of Q are equal to {28, 38, 30}, respectively. These values were verified in simulations, which closely matched the subsequently shown experimental responses.

6.5.2 Tests for various grid anti-resonances

The objective of this section is to verify the implications of connecting a VSC with a non-passive input admittance to a grid with an anti-resonance that coincides with the non-passive region. The VSC works as a rectifier with a unity power factor, as the used Chroma power supply does not support regenerative operation. The output frequency is $f_1 = 50$ Hz, and the PLL is implemented with 5 Hz bandwidth. The schematic of the grid-connected scenario is shown in Fig. 6.21.



Figure 6.20: (a) Circuit used for measuring passive damping of the tested anti-resonant networks. (b) Measured response for $Z_{g,2}$; the fundamental component of v_{pcc} is filtered out for a better visualization.

6.5.2.1 Comparison between MS-PWM and DS-PWM without AD

In this section, the operation is tested for $N = \{2, 4, 32\}$, two values of the relative crossover frequency $\alpha = 0.1$ and $\alpha = 0.2$, and two values of the filter inductance, L = 1.5 mH and L = 2.5 mH. The two inductors feature the same core properties and similar winding resistances, yielding comparable losses.



Figure 6.21: Illustration of the grid-connected set-up. The grid anti-resonance appears due to the presence of C_g and L_g . In blue part, OCP and OVP stand for over-current and over-voltage protection, respectively.

Tests in this section are performed for the grid impedance $Z_{g,1}$. The corresponding anti-resonance appears at $f_{res} = 7.9$ kHz, which is very close to the frequency where the IFP is found for N = 2. For the other tested values of N, MS-PWM renders the admittance passive around f_{res} . To illustrate the impact of the tested values of L and α on the negative damping introduced by the VSC, for DS-PWM without AD, Fig. 6.22 shows the real part of Y_i . The small-signal modeling predicts that the configuration with the higher bandwidth and the lower inductance will have the worst impact on stability. Based on Fig. 6.19a, even for the case with the highest negative damping (L = 1.5 mH and $\alpha = 0.2$), instability is not expected.

The experimental results show oscilloscope measurements of one fundamental period of i_L (blue trace) and v_{pcc} (red trace). A transient is triggered by imposing the step change of the current magnitude reference from 4 A to 12 A. The reference step is synchronized with the instant where the current is at its peak value. To test the impact of various system parameters on damping, results are also given for two settings of the input/output voltages: $V_{in} = 400$ V and $v_{pcc,RMS} = 230$ V, and $V_{in} = 250$ V and $v_{pcc,RMS} = 110$ V.

In Fig. 6.23, results are shown for the case of N = 2, $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 400$ V, and



Figure 6.22: Real part of the VSC input admittance Y_i for DS-PWM, the two tested inductances L, and the two tested relative bandwidths α .

 $v_{pcc,rms} = 230$ V. For all tested cases with N = 2 and $\alpha = 0.2$, the phase margin is around 30°, which results in a step response overshoot. This bandwidth setting, corresponding to 10 % of the sampling frequency, is used for experimental results in [135] and a slightly higher one is used in [148]. When the reference step change is imposed, the oscillations at the anti-resonant frequency are evident, especially in the v_{pcc} trace. After nearly half of the fundamental period, the oscillations are fully damped due to the non-negligible passive damping of the grid anti-resonant circuit. Note also the additional presence of a lower frequency oscillation in v_{pcc} , corresponding to the before-mentioned ac source decoupling.



Figure 6.23: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. System parameters are: N = 2, $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 400$ V, and $v_{pcc,rms} = 230$ V.

For the subsequent test, shown in Fig. 6.24, DS-PWM is again used with $\alpha = 0.2$ and L = 1.5 mH. However, the voltages are changed to $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V. It can be seen that much stronger oscillations are triggered, but again, parasitic resistances of the experimental anti-resonant

impedance $Z_{g,1}$ prevent the full system instability to occur. The reason for stronger oscillations is consistent with the previously mentioned impact of D on admittance properties. Namely, for the configuration with $V_{in} = 400$ V, and $v_{pcc,rms} = 230$ V, the maximal value of D is equal to approximately 0.91, while for $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V, it is equal to 0.81. Given that the first negative conductance peak is decreased as values of D approach extreme values (see Fig. 6.16 and [73]), the oscillations are more damped. For all subsequent tests, the configuration with $V_{in} = 250$ V and $v_{pcc,rms} = 110$ V is chosen to enable a more critical scenario regarding stability.



Figure 6.24: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. System parameters are: N = 2, $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V.

In Fig. 6.25, results are shown for the case of the higher inductance: N = 2, $\alpha = 0.2$, L = 2.5 mH, $V_{in} = 250$ V and $v_{pcc,rms} = 110$ V. It can be seen that much stronger resonance damping is present than in Fig. 6.24. Since the two cases feature the same crossover frequency and phase margin, the only difference in damping comes from the more favourable passivity measures of VSCs with higher values of L, as predicted by Fig. 6.22.

To show the impact of reducing the bandwidth, the results in Fig. 6.26 are shown for the case of N = 2, $\alpha = 0.1$, L = 1.5 mH, $V_{in} = 250$ V and $v_{pcc,rms} = 110$ V. It can be seen that oscillations are strongly damped compared to the case with the same parameters except for a higher α , seen in Fig. 6.24.

To verify the positive impact of MS-PWM, the control systems with N = 4 and N = 32 were tested for the most critical system configuration: $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V. The results are directly comparable to the one seen in Fig. 6.24. For N = 4, the results are shown in Fig. 6.27. It can be seen that, compared to the double-update case, the triggered oscillations are fully damped in a significantly shorter time. For N = 32, the results are shown in Fig. 6.28 and it is clear that the damping is even higher.



Figure 6.25: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. The control system parameters are: N = 2, $\alpha = 0.2$, L = 2.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V.



Figure 6.26: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. System parameters are: N = 2, $\alpha = 0.1$, L = 1.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V.

The experimental results are consistent with the analytical prediction of the impact of N, L, and α on the introduced negative damping. For the tested impedance $Z_{g,1}$, transients trigger oscillations at the anti-resonant frequency; however, the system instability does not occur. This confirms that passivity is not a necessary condition for the system stability, as damped grid resonances can compensate the non-passive input admittance of the VSC.

Two positive aspects of MS-PWM are seen from the presented results. Firstly, due to delay reduction, for the same crossover frequency, the phase margin is improved compared to N = 2, which can be seen from the step response. Secondly, MS-PWM renders the admittance passive, which adds a high damping to oscillations at the anti-resonant frequency. Therefore, MS-PWM offers the capability of implementing very high bandwidth loops, while the passivity extension is inherently provided, without any additional active damping strategy.



Figure 6.27: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. System parameters are: N = 4, $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V.



Figure 6.28: Experimental tests of the grid-connected scenario. Reference current magnitude is changed from 4 A to 12 A. System parameters are: N = 32, $\alpha = 0.2$, L = 1.5 mH, $V_{in} = 250$ V, and $v_{pcc,rms} = 110$ V.

As a final verification, hardware-in-the-loop (HIL) results are given to demonstrate the instability condition in case the VSC is connected to a grid with a high quality factor. These results were not obtainable in the experimental set-up; however, in a general case, the grid resonance can feature very low passive damping. The simulations were organized using the Typhoon HIL 402 HIL platform to emulate the VSC and the grid, i.e. red part in Fig. 6.21. The inductor current and the PCC voltage are obtained via HIL analog outputs. Sensing, ADC, PLL, current control, and PWM were implemented in the same control platform that was used for the experimental tests, i.e. the blue part in Fig. 6.21 remains unchanged, except for the input signal scaling. Resulting transistor switching signals are connected to HIL digital inputs. In order to verify the consistency of the simulation, all admittance measurements were repeated and results matched those obtained experimentally; hence, admittance properties remain unchanged. For the instability test, the operation is compared between N = 2 and N = 16 and the chosen system parameters are: $\alpha = 0.2, L = 2.5$ mH, $V_{in} = 400$ V, $v_{pcc,rms} = 230$ V, $P = P_n = 3$ kW. The values of L_g and C_g correspond to $Z_{g,1}$. An 1 Ω resistor is placed in series with the grid inductance L_q . At t = 20 ms, the resistor is paralleled with another 50 m Ω resistor, in order to reduce the passive damping and observe the resulting transient. The voltage protection is set to turn off the VSC when $|v_{pcc}|$ exceeds 380 V or $|i_L|$ exceeds 25 A. The results are shown in Fig. 6.29. It can be seen that, for N = 2, a stable operation cannot be maintained as the grid anti-resonance, which falls into the non-passive admittance region, is no longer damped by the 1 Ω resistor. The protection is triggered at approximately t = 39 ms and the VSC is turned off. On the other hand, for N = 16 and the same exact test, the operation remains stable and the anti-resonant oscillations are quickly damped.

6.5.2.2 Impact of AD strategies and comparison of effectiveness with MS-PWM

Let us now compare the effectiveness of DS-PWM with the previously analyzed AD strategies and MS-PWM. The operation is tested for L = 1.5 mH and $\alpha = 0.2$, and the three anti-resonant networks described in Section 6.5.1. The tested scenarios correspond to the predictions by the Nyquist plots in Fig. 6.19. Voltage levels are set to $V_{in} = 250$ V and $v_{pcc,RMS} = 110$ V.

Results of grid-connected tests are shown in Fig. 6.30. The oscilloscope screen shows i_L (orange) and v_{pcc} (red). Each column in Fig. 6.30 shows results for one of the tested grid anti-resonant networks. Note that the switching ripple in v_{pcc} changes for different Z_g configurations. In the first two rows, the results are shown for the case of MS-PWM with N = 4 and N = 16, which enable a stable operation for all the tested anti-resonances. A transient is imposed by stepping the current reference from 4 A to 15 A, at the peak of the fundamental component. The results for N = 4 are given to show that it is not always necessary to raise N to very high values. Note that, based on the measurements from Section 6.4.3, MS-PWM with N = 4 results in a small non-passive zone around f_{pwm} , which coincides with the anti-resonance $Z_{g,2}$. However, the negative conductance is very small and the passive grid damping is enough to compensate it.

In the third row of Fig. 6.30, results are shown for DS-PWM strategies that feature non-passive zones, which coincide with the correspondingly tested anti-resonances. In Fig. 6.30g, strong antiresonant oscillations are triggered; however, as predicted by Fig. 6.19a, due to a sufficiently high passive damping of $Z_{g,1}$, instability is prevented. In Figs. 6.30h and 6.30i, instability occurs for AD strategies, as predicted by Figs. 6.19b and 6.19c. For these anti-resonances, DS-PWM without AD is stable, which


Figure 6.29: HIL tests of the grid-connected scenario. The 1 Ω resistance, placed in series with L_g , is paralleled with a 50 m Ω resistor at t = 20 ms. System parameters are: $\alpha = 0.2$, L = 2.5 mH, $V_{in} = 400$ V, $v_{pcc,rms} = 230$ V, and: (a) N = 2, (b) N = 16.

is used to impose transients by activating the respective AD strategy. In Fig. 6.30h, the overvoltage protection is triggered when DD-AD is activated, and the VSC is turned off. In Fig. 6.30i, when D-AD is activated, the protection is not triggered; however, instability clearly occurs and the controller output bounces from positive to negative saturation limits. From Fig. 6.30i, it is clear that oscillations are much more emphasized around the zero crossings of v_{pcc} , where D = 0.5. This is in agreement with the comments made in Section 6.4.4 regarding the impact of the operating point on the negative conductance peak. Note that this case corresponds to instability caused by an anti-resonance above the NF. Its enabling mechanism is the coupling between the sampling and modulation sidebands, which allows the controller to react to a disturbance even above the NF [141]. In the fourth row of Fig. 6.30, zoomed-in waveforms corresponding to the third row are given to show that the oscillations are triggered at the anti-resonant frequency.

It is concluded that the implementation of DS-PWM with AD may destabilize the operation for



 $Z_{g,1}:f_{res} = 7.9 \text{ kHz}, \ Z_c^{\%} = 20 \ \%, \ Q = 28 \ Z_{g,2}:f_{res} = 18.4 \text{ kHz}, \ Z_c^{\%} = 47 \ \%, \ Q = 38 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Q = 30 \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_c^{\%} = 72 \ \%, \ Z_{g,3}:f_{res} = 28.2 \text{ kHz}, \ Z_{g,3}:f_{res} = 28.2 \text{ k$

Figure 6.30: Experimental demonstration of passivity implications in grid-connected scenarios. Orange line shows i_L and red line shows v_{pcc} that has a variable ripple depending on C_g . Each column features results for one of the tested grid anti-resonant networks, described in Section 6.5.1. The results are shown as transient responses to: (a), (b), (c) the step current reference change for MS-PWM with N = 4; (d), (e), (f) MS-PWM with N = 16; (g) DS-PWM without AD with non-passive zone of Y_i coinciding with $Z_{g,1}$ anti-resonance; the transient causes strong oscillations at f_{res} but the passive grid damping prevents instability; (h) the operation change from DS-PWM without AD to DD-AD; due to a weakly-damped $Z_{g,2}$ anti-resonance, coinciding with the non-passive zone of Y_i , instability occurs and the protection is triggered; (i) the operation change from DS-PWM without AD to D-AD; due to a weakly-damped $Z_{g,3}$ anti-resonance, coinciding with the non-passive zone of Y_i , instability occurs; (j), (k), (l) zoomed-in waveforms from the row above, showing the oscillation frequencies.

certain grid anti-resonances. Hence, AD *does not guarantee stability* of a grid-connected VSC for any passive grid impedance. On the other hand, MS-PWM retains stability for all tested cases.

6.5.2.3 Some considerations on validity of single-frequency models

As previously mentioned, the modeling approach used in this chapter does not include the impact of the sidebands generated by the sampling and the modulation. Some of the reasoning behind this is given

below.

First of all, the single-frequency model offers a good prediction of VSC behavior in the frequency range of interest (up to at least 1.5 f_{pwm}). This is validated based on the match with the admittance measurements as well as the grid-connected stability tests. Secondly, multi-frequency models are successfully derived only for single-loop controllers without AD. Their extension to derivative-based AD is not straightforward as two signals of different harmonic properties are sampled, at possibly different rates, and the application of superposition is questionable due to the nonlinear nature of the multi-frequency system. In [18], the modeling approach offers inclusion of the AD, however, the resulting MIMO model is quite complex and its implementation has resulted in a significant mismatch with the behavior of the analyzed system (whether in simulations or experiments). There is definitely a motivation for deriving a multi-frequency model for the case of AD and correlating it with the grid-connected stability; however, this is out-of-scope of the conducted research on MS-PWM and is, hence, left for future work.

Here, it is investigated how the Nyquist plots change for different models, for DS-PWM without AD. Besides the single-frequency model from (6.4) and the multi-frequency model from [73], it was of interest to see how credible it is to make conclusions regarding stability based only on the equivalent network parameters at the anti-resonant frequency, as in Fig. 6.18. Namely, by considering the VSC admittance only at the anti-resonant frequency, the values of the equivalent resistor and the inductor (if inductive) can be calculated and used to extrapolate Y_i for other frequencies as well. This approach may be of interest as it is based only on one measurement point. In Fig. 6.31, the Nyquist plots are shown for the case of DS-PWM without AD, L = 1.5 mH, $\alpha = 0.2$, the first tested grid impedance $Z_{g,1}$, and:

- Experimentally measured points of Y_i , corresponding to Fig. 6.5b.
- Extrapolated trace considering Y_i to be a parallel connection of the equivalent resistor and inductor, obtained at the anti-resonant frequency.
- Multi-frequency model of Y_i from [73].
- Single-frequency model of Y_i , used in the previous sections.

As it can be seen from the Fig. 6.31, all 4 modeling approaches yield very similar Nyquist contours. Although one might argue that the single-frequency model is not mathematically accurate, it is linear and sufficient to give a good insight on a connection stability. Since the focus of this chapter is not on the modeling itself, it was concluded that the fact that all outcomes of the grid-connected tests are predictable by the linear single-frequency models is good enough for the purpose of comparison.

6.6 Noise sensitivity of passivizing current control strategies

A question that arises for AD strategies is their noise sensitivity [159], as the derivative action is applied to v_{pcc} . The goal of this section is to quantify the noise propagation from sensing to the output current i_L , for passivizing control strategies tested beforehand. As seen in Fig 6.1b, noise due to the current



Figure 6.31: Comparison of Nyquist plots for DS-PWM without AD, obtained with various modeling strategies. The results are given for $Z_{g,1}$, L = 1.5 mH, and $\alpha = 0.2$.

measurement, n_i , is present for all strategies, while AD brings an additional noise path to the system, from n_v to i_L . Results of this section are based on the analysis from Chapter 5. Again, it is assumed that n_i and n_v are uncorrelated WSS Gaussian white noise signals [127], with corresponding variances labeled as σ_i^2 and σ_v^2 .

For all compared strategies, noise propagation due to n_i is determined by:

$$\sigma_{i_L}^2\Big|_{n_v=0} = \int_0^{f_x} \sigma_i^2 \frac{2}{f_s} |H_{ni}(f)|^2 df = \int_0^{f_x} S_{ni}(f) df,$$
(6.12)

where f_x is the band of interest, $|H_{ni}(f)|^2 = \frac{|W_{ol}(f)|^2}{|1+W_{ol}(f)|^2}$ is the squared magnitude response from n_i to i_L in Fig. 6.1b, and $S_{ni}(f)$ is the PSD of the inductor current, caused by n_i . The PSD is decreased by oversampling the feedback, as determined by f_s . As shown in Section 5.6, the analytical prediction using (6.12) provides good results for SS-PWM and DS-PWM single-loop controllers. For the case of MS-PWM, it is shown that the limiting aspect is the fact that the triangular DPWM behaves as a decimator with frequency $2f_{pwm}$, thus causing aliasing of noise components above f_{pwm} . For this reason, digital anti-aliasing filters are needed to suppress the resulting noise floor below the level determined by DS-PWM.

For AD, noise propagation due to n_v is determined by:

$$\sigma_{i_L}^2\Big|_{n_i=0} = \int_0^{f_x} \sigma_v^2 \frac{2}{f_{s,1}} |H_{nv}(f)|^2 df = \int_0^{f_x} S_{nv}(f) df,$$
(6.13)

where $|H_{nv}(f)|^2$ is the squared magnitude response from n_v to i_L and $S_{nv}(f)$ is the PSD of the inductor current, caused by n_v . Assuming that the two noise signals are uncorrelated, the total in-band variance of i_L is obtained as an algebraic sum of (6.12) and (6.13) [127].

An interesting fact is that, for the case of D-AD, the white noise n_v is shaped by the active damping filter before being sampled with f_s . In case of the oversampled digital implementation, this corresponds to the decimation from $f_{s,1}$ to f_s . Let us, for this reason, define $|H_{nv}(f)|^2$ as a product of squared magnitude responses from n_v to the decimator output and from the decimator output to i_L :

$$|H_{nv}(f)|^2 = |H_{ad}^{M:1}(f)|^2 \frac{|G_d(f)G_p(f)|^2}{|1 + W_{ol}(f)|^2},$$
(6.14)

where $M = \frac{f_{s,1}}{f_s}$ is the decimation factor and $|H_{ad}^{M:1}(f)|^2$ is the modified squared magnitude response of the AD filter that includes the decimation impact. Given that n_v is a stationary white stochastic signal, it exhibits no spectral correlation and its PSD is constant [130, 175]. As explained in Section 5.5, assuming integer values of M, this allows us to form the equivalent model for the PSD analysis, as in (5.16):

$$|H_{ad}^{M:1}(f)|^2 = \sum_{k=0}^{M-1} |H_{ad}(f - kf_s)|^2.$$
(6.15)

For the case of DD-AD, where M = 1, $|H_{ad}^{M:1}(f)|^2 = |H_{ad}(f)|^2$. The equivalent squared magnitude response from (6.15) is verified in simulations in Section 5.5.1.2 (see Fig. 5.14).

Individual impacts of (6.12) and (6.13) on the total variance of i_L , for DS-PWM, are illustrated in Fig. 6.32 by showing S_{ni} and S_{nv} , calculated for the input noise powers found in the experimental set-up (see Section 5.6). It can be seen that, for DD-AD, the PSD due to the voltage noise sensing is well below the one due to the current noise sensing; hence, the voltage noise propagation is completely masked and $\sigma_{i_L}^2$ is expected to be the same as for DS-PWM without AD. This outcome is the result of a relatively low required derivative gain k_{ad} in (6.9). For D-AD, however, the derivative in H_{ad} first causes the amplification of the high-frequency noise content, which then gets decimated and folded back to frequencies below the NF. This causes the in-band noise power to significantly increase. From Fig. 6.32 it can be seen that for $N_{ad} = 16$, which is used in this section, the PSDs due to n_i and n_v almost overlap. For $N_{ad} = 32$, n_v has a higher impact than n_i . Note that, for the analog implementation of D-AD, $M \to \infty$ and the impact of n_v is even higher, with the exact amount being dependent on the actual bandwidth of the input noise signal. These conclusions come from the fact that for D-AD, H_{ad} needed to render Y_i fully passive up to the NF, does not include an anti-aliasing filter [135]. In cases where the full passivity up to the NF is not crucial, an anti-aliasing filter can be placed between H_{ad} and the decimator, which would result in a strong noise suppression compared to DD-AD, while offering a similar passivizing effectiveness. This requires additional investigation and is a potential topic of future research. It should be noted that results in Fig. 6.32 are directly proportional to given σ_i^2 and σ_v^2 . Hence, for a higher ratio $\frac{\sigma_v^2}{\sigma_i^2}$, D-AD can show an increased sensitivity even for lower values of N_{ad} . However, regarding DD-AD, the PSD line for n_v stays more than 20 dB $\frac{A^2}{Hz}$ below the one for n_i , meaning that σ_v^2 should increase more than 100 times to result in a comparable impact, which is not a realistic case. Hence, it can be concluded that DD-AD is not likely to be prone to noise sensitivity.

For the experimental verification, the reference current is set to 5 A RMS and L = 2.5 mH is used in order to capture the current signal using the oscilloscope with 2 A/div scale. Larger scale was resulting in the noise floor of the oscilloscope compromising i_L noise measurements. The output variance, up to



Figure 6.32: Output power spectral densities $(S_n \in \{S_{ni}, S_{nv}\})$ of i_L for DS-PWM due to current noise sensing (blue trace) and voltage noise sensing (red traces). Presented results are calculated for L = 2.5 mH, $\alpha = 0.1$, $\sigma_i^2 = 1.7 \cdot 10^{-3} \text{ A}^2$, and $\sigma_v^2 = 154 \cdot 10^{-3} \text{ V}^2$.

 $\frac{f_{pwm}}{2}$, is calculated in MATLAB using 200 ms of the oscilloscope data, sampled with 250 MS/s. For post-processing, the fundamental component of i_L and its harmonics are removed from the spectra, so that distortion does not impact the variance calculations. As a benchmark, simulations are implemented in Simulink using PLECS Blockset, as in Chapter 5. White noise signals are injected, with variances corresponding to the measured σ_i^2 and σ_v^2 .

Results that show the output variance $\sigma_{i_L}^2$, for all tested strategies, are presented in Fig. 6.33. The experimental measurements are compared with the simulations and the analytical results. Results in Fig. 6.33a are given for the relative bandwidth $\alpha = 0.1$. First, it can be seen that an excellent analytical prediction is obtained for DS-PWM without AD and with DD-AD. The results show almost the same output noise power for these two strategies. As both cases feature the same magnitude response $|H_{ni}|$, it is concluded that the voltage noise propagation for DD-AD is completely masked by the stronger current noise propagation, which is consistent with Fig. 6.32. For D-AD, the analytical model predicts 2.6 dBA² noise power increase, while the simulations and experiments show approximately 2.4 dBA² and 2.5 dBA² increase, respectively, which supports the validity of the model (6.15). As discussed above, for set-ups that feature a higher ratio $\frac{\sigma_v^2}{\sigma^2}$, higher noise sensitivity of D-AD is expected.

Equation (6.12) predicts that, compared to DS-PWM, implementing MS-PWM with N = 16 would decrease the output noise power by, at least, a factor 8. The exact amount depends on the impact of delay reduction on $|H_{ni}|$. However, as analyzed in Chapter 5, for N > 2 and the high-frequency proportionaldominant controllers, the PSD remains almost unaffected due to the aliasing caused by the DPWM decimation; hence, additional high-frequency digital filtering is needed to enable a strong in-band noise suppression. Here, three DLPFs $G_{fb}(z)$ are tested to observe their impact on noise attenuation. The filter structure corresponds to the one in (4.21), and the tested cut-off frequencies are $\omega_{c,f} = \{2, 1.5, 1\} \cdot \omega_{pwm}$. It is important to notice that these DLPFs do not directly attenuate noise power in the observed frequency window. The design goal is to suppress the noise power above f_{pwm} with as little impact on the dynamic



Figure 6.33: Variance of i_L up to $\frac{f_{pwm}}{2}$ for L = 2.5 mH and: (a) $\alpha = 0.1$; (b) $\alpha = 0.2$.

performance⁸. DLPFs with lower cut-off frequencies bring more filtering at $f > f_{pwm}$; hence, the aliasing is additionally suppressed and the in-band noise power is expected to further decrease. The analytical prediction for MS-PWM is obtained using the equivalent transfer function as in (5.23). Note that, even without the DLPFs, the output noise power is reduced compared to DS-PWM without AD by 2.2 dBA², due to a higher phase margin, which brings additional filtering by the closed-loop system.

Same conclusions are brought based on results shown in Fig. 6.33b, given for $\alpha = 0.2$, except that all variances are increased due to the higher bandwidth setting. It can be also seen that a higher noise suppression of MS-PWM without DLPF relative to DS-PWM without AD is obtained, equal to

⁸In order to verify that the implemented DLPFs do not strongly deteriorate passivity properties, the admittances are measured for the tested DLPFs and only a small impact is seen. Negative conductance peaks, measured for $\alpha = 0.2$ for compatibility with results in Table 6.1, are $\{-0.25, -0.17\}$ % for $L = \{1.5, 2.5\}$ mH at 28 kHz.

approximately 5 dBA², due to a higher phase margin difference between the two. Compared to $\alpha = 0.1$, a slightly higher error between the analytical data and measurements is obtained for MS-PWM. Some mismatch is assumed to come from the ac operation tested in this chapter. Still, the trend is wellpredicted and the error is not much higher than in Chapter 5.

The main conclusion of this section is that implementing AD does not necessarily result in excessive noise sensitivity. Still, MS-PWM is able to offer additional noise suppression without compromising the passivation effectiveness.

6.7 Summary

Firstly, this chapter has shown that MS-PWM is capable of rendering the VSC admittance passive in a wide frequency range. This is achieved without any passive or active damping technique, which significantly reduces the system complexity and cost. Some impact of the MS-PWM nonlinearities is present; however, it is not found to be detrimental for passivity properties. Even for lower values of N, where the measured admittance still features a small non-passive zone, the resulting negative damping is strongly reduced compared to double-update.

Secondly, this chapter has provided a comparison in terms of the effectiveness and noise sensitivity of different control strategies used to render the admittance of grid-connected VSCs passive. MS-PWM is compared against typically used derivative-based AD methods. It is shown that MS-PWM outperforms AD in the passivizing effectiveness, as AD brings a negative impact around and above the Nyquist frequency. Grid-connected operation is tested for different anti-resonant networks and the results demonstrate that only MS-PWM enables robust stability. Analytical modeling of noise propagation is proposed and verified, revealing that AD strategies are not highly sensitive. Still, MS-PWM outperforms AD in this respect as well.

MS-PWM is shown to be a promising candidate for grid-connected VSCs as it enables very high bandwidths, while inherently suppressing the non-passive input admittance behavior, therefore guaranteeing robustness to high-frequency harmonic instability of current-controlled grid-connected VSCs. To enable highest damping and lowest impact of the MS-PWM nonlinearities, it is advisable to choose as high an oversampling factor as allowed by the control platform.

Chapter 7

Conclusions

This thesis analyzes the multi-sampled pulsewidth modulation and focuses on its fundamental properties, important implementation aspects, and applicability for solving a practical problem found in grid-tied converters. The dissertation structure is organized such that it first brings the analysis of negative aspects of MS-PWM and provides a methodology for their suppression. Then, it investigates a way to emphasize inherently positive aspects of MS-PWM. Finally, it builds upon these findings to examine the potential of MS-PWM in solving a practical problem, while having a full awareness of its capacities and limitations.

The first three chapters serve as an introduction and a review of the existing knowledge.

Chapter 4 analyzes the MS-PWM nonlinearities caused by the switching ripple, particularly those related to the jump discontinuities of the modulating waveform. The discontinuity-related nonlinearities are classified, relations between the modulating waveform and the carrier that enable them are explained, and the resulting impacts on the converter's operation are described. Then, the chapter focuses on the MS-PWM nonlinearity that is discovered during this work. It is demonstrated how this nonlinearity, which is manifested as the infinite-gain zone of the modulator, brings a mechanism for limit-cycle oscillations. The consequential jittering of the duty cycle is statistically modelled and it is shown to be directly dependent on the size of the critical modulating waveform discontinuities. An algorithm is proposed to remove the jittering in steady-state conditions. Then, the chapter continues with exploring how different control system settings, feedback filters, and controller structures, impact the sign and size of the critical discontinuities, in this way determining which nonlinearity will arise and to what extent. The end result is a framework for the system design that minimizes the impact of the MS-PWM nonlinearities, without having to completely filter out the switching ripple. The main conclusions of this chapter are:

• Sign of the critical modulating waveform discontinuities determines which type of nonlinearity will be brought to the system. The negative sign, i.e. the counter-phase operation, brings the reduced-gain and the zero-gain zones. The positive sign, i.e. the in-phase operation, brings the infinite-gain zones that are, in general, highly unfavourable.

• Size of the discontinuities determines the nonlinearities' extension and, hence, the overall impact on the converter. The discontinuities can be reduced by an appropriate filtering or simply by increasing the oversampling factor. This brings a motivation for choosing as high an oversampling factor as possible.

Chapter 5 analyzes the noise propagation in MS-PWM systems. First, it is shown that the DPWM down-samples the modulating waveform, which causes aliasing and limits the noise suppression when the oversampling factor is increased above 2. Correspondingly, design of the anti-aliasing digital filter is proposed, with a goal to reduce the decimation impact while not strongly affecting the dynamic performance. Then, new analytical models are derived for noise propagation in open-loop and closedloop systems that feature decimation. Extensive experimental measurements validate the derived models and show that the MS-PWM control systems with the high-frequency proportional-dominant controllers (P, PI, PR) and the proposed anti-aliasing filters can strongly suppress noise with a small impact on the dynamic response. Without any filters, increase of the oversampling factor above 2 practically does not have any impact on the output noise power. Contrarily, for MS-PWM systems with derivative-based controllers, increase of the oversampling factor on its own brings noise amplification. In such cases, for strong noise suppression, the required anti-aliasing digital filters introduce a higher impact on the system dynamics. This chapter is concluded by comparing the noise suppression capacities of MS-PWM and a more common MR-DS-PWM strategy, which applies the oversampled filtering and subsequently decimates the control rate to double-update. The newly derived models for noise propagation are shown to be valid also for MR-DS-PWM. It is revealed that the noise attenuation capabilities of these two strategies are practically the same. However, unlike MS-PWM, MR-DS-PWM strongly penalizes the dynamic performance. The main conclusions of this chapter are:

- In terms of noise propagation, digital pulsewidth modulators introduce a re-sampling action with the rate determined by the number of modulated edges of the switching signal. Correspondingly, for MS-PWM with the triangular carrier, the decimation occurs when the oversampling factor is increased above 2.
- To enable strong noise suppression with MS-PWM, some anti-aliasing digital filtering is needed. These filters can be designed so as to impose a small impact on the dynamic performance.
- For strong noise suppression and a high dynamic performance, choosing as high an oversampling factor as possible is advised. However, care should be taken if derivative-based controllers are used. For those, without including the anti-aliasing digital filters, the increase of the oversampling factor amplifies noise rather than suppressing it.

Chapter 6 analyzes the effectiveness of MS-PWM in providing robust stability of grid-tied VSCs. The chapter begins by analyzing the impact of the control system delays on introducing negative damping in the high-frequency range. It is explained how the system stability can be analyzed using the VSC admittance and how a sufficient condition for stability can be achieved by rendering the admittance passive. The chapter continues with the experimental admittance measurements and comparison between the control system with MS-PWM and the ones with DS-PWM and typically-used active damping methods, based on the derivative feedforward actions. It is demonstrated that MS-PWM is capable of rendering the admittance passive in a wide frequency range, without having to use any additional active damping technique. Furthermore, impact of the MS-PWM nonlinearities is examined and it is shown not to be detrimental for the passivity properties. Suppression of the nonlinearities using a digital filter designed in Chapter 4 is successfully tested as well. As for the AD strategies, it is shown that their effectiveness is limited to the medium frequency range, while bringing a detrimental impact at very high frequencies. The implications of the examined admittance properties on stability is experimentally tested by connecting the VSC to a grid with specific impedances. It is verified that MS-PWM enables robust stability, while AD strategies cause instability for some of the tested scenarios. In the end, the chapter examines the noise sensitivity of the analyzed passivizing control strategies by deriving the corresponding noise propagation models and comparing them with the experimental measurements. It is shown that the derivative-based AD strategies are not highly sensitive; however, MS-PWM again offers a superior performance. For MS-PWM, anti-aliasing filters designed based on the procedure from Chapter 5 are tested and it is demonstrated that a strong noise suppression is possible, with a negligible impact on the admittance properties. In summary, it is verified that MS-PWM is a very suitable highperformance control strategy for grid-tied VSCs. Together with enabling a high dynamic performance, it inherently suppresses the non-passive admittance behavior and therefore offers a robustness to highfrequency harmonic stability problems found in grid-following VSCs. The main conclusions of this chapter are:

- MS-PWM is suitable for the high-frequency passivity-oriented control design. It inherently extends the passive zone by reducing, instead of compensating, control delays.
- For highest damping and lowest impact of MS-PWM nonlinearities, it is advisable to choose as high an oversampling factor as allowed by the control platform.

Some practical remarks on MS-PWM, based on the research conducted in this thesis, are listed here.

- The oversampling factor should always be chosen as high as allowed by the control platform. Besides the positive impact on delay reduction and noise suppression, higher oversampling factors significantly reduce the impact of the MS-PWM nonlinearities caused by the modulating waveform discontinuities, which are the main limitation of this modulation strategy.
- With MS-PWM, applying some low-pass filtering to the feedback signal is generally advisable. First, a suitable anti-aliasing filter design can enable a strong noise suppression. Second, particularly for lower oversampling factors, the discontinuity-related nonlinearities can be effectively mitigated with an appropriate design. In this case, care must be taken because low-pass filters do not always bring a positive impact, as they may cause the in-phase operation and, consequently, introduce limit-cycle oscillations. Third, for high oversampling factors, although the discontinuities

have a low impact, low-pass filtering reduces the switching ripple magnitude, which suppresses the effects of ripple modulation. The simplest solution that removes practically all of the nonlinearities is the moving average filter, however, its impact on the dynamic performance is very high.

To conclude, there are several interesting topics for further research:

- Analysis of sensitivity to the switching noise caused by high dv/dt and di/dt during commutations of the converter. In principle, sampling of the switching noise is not only present in MS-PWM and its suppression is a topic of high importance in many applications. As noted in Chapter 5, this type of noise is correlated to the feedback signal and is strongly dependent on the hardware design; hence, it is hard to treat analytically. With a proper hardware and sensing circuit design, switching noise is not highly manifested and MS-PWM can be fully exploited in relevant industrial applications. On the other hand, in systems where the feedback signal is strongly corrupted by the switching noise, having a very high bandwidth control is a hardly attainable goal. However, with many samples per switching period, it may be possible to implement some smart filtering strategy. For example, ripple predictors may be used to replace a noise-corrupted sample by tracking its difference from the estimated value. Another possibility is the use of non-linear rank filters to discard the corrupted sample with a low impact on the dynamic response. Finally, it should be mentioned that using moving average filters may help reduce the switching noise impact if the number of samples if sufficiently high. For example, if one out of 32 samples contains the switching noise, the averaging will strongly reduce its impact. However, for a lower number of samples, the averaging may have a detrimental impact on the system performance. For example, if 4 samples are averaged, out of which 1 is corrupted by the switching noise, its impact is not highly suppressed. Moreover, the averaging action maintains the information on the corrupted sample throughout the averaging period. On the other hand, without any filters, if the switching noise is acquired several sampling periods before the switching action occurs, it may not have any impact at all on the modulating segment that intersects with the carrier. In general, for MS-PWM, it is not clear whether simple low-pass filtering reduces or amplifies problems related to the switching noise. It is, however, evident that, once again, related effects are reduced for higher oversampling factors.
- Further optimization in designing the feedback filters. Supposing that the switching noise is not highly manifested, the objective is to effectively suppress the MS-PWM nonlinearities caused by the switching ripple and, if needed, to provide an anti-aliasing function for a strong white noise suppression. The design should be such that it brings as low impact on the dynamic performance as possible.
- Investigation of MS-PWM properties in multi-level and interleaved converters with PS-PWM. In most reported applications, to avoid sampling the switching ripple, the oversampling factor is determined by the number of phase-shifted carriers, i.e. by the ripple frequency multiplication. However, in unbalanced conditions, the ripple is unavoidably sampled and, therefore, the MS-

PWM nonlinearites are expected to arise. Moreover, with a powerful-enough control platform, nothing prevents an additional increase of the oversampling factor. It is expected that this would further push the dynamic capabilities, even though at the expense of the switching ripple being introduced in the feedback.

Bibliography

- D. Boroyevich, I. Cvetkovic, R. Burgos, and D. Dong, "Intergrid: A Future Electronic Energy Network?" *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 1, no. 3, pp. 127–138, 2013.
- [2] J. M. Guerrero, J. C. Vasquez, J. Matas, L. G. de Vicuna, and M. Castilla, "Hierarchical Control of Droop-Controlled AC and DC Microgrids—A General Approach Toward Standardization," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 1, pp. 158–172, 2011.
- [3] S. Buso and P. Mattavelli, "Digital Control in Power Electronics, 2nd Edition," Synthesis Lectures on Power Electronics, Morgan & Claypool Publishers, USA, 2015.
- [4] L. Corradini, D. Maksimovic, P. Mattavelli, and R. Zane, *Digital control of high-frequency switched-mode power converters*. John Wiley & Sons, 2015.
- [5] S. Vazquez, J. Rodriguez, M. Rivera, L. G. Franquelo, and M. Norambuena, "Model Predictive Control for Power Converters and Drives: Advances and Trends," *IEEE Transactions on Industrial Electronics*, vol. 64, no. 2, pp. 935–947, 2017.
- [6] P. Cortes, M. P. Kazmierkowski, R. M. Kennel, D. E. Quevedo, and J. Rodriguez, "Predictive Control in Power Electronics and Drives," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 12, pp. 4312–4324, 2008.
- [7] J. Y. Tsao et al., "Ultrawide-Bandgap Semiconductors: Research Opportunities and Challenges," Advanced Electronic Materials, vol. 4, no. 1, p. 1600501, 2018. [Online]. Available: https://onlinelibrary.wiley.com/doi/abs/10.1002/aelm.201600501
- [8] M. A. Knight and R. A. Torkildsen, "One-kVA, Three-Phase dc-ac Inverter with Digital Control," *IEEE Transactions on Aerospace and Electronic Systems*, vol. AES-5, no. 6, pp. 989–995, 1969.
- [9] S. Bowes and M. Mount, "Microprocessor control of PWM inverters," *IEE Proceedings B* (*Electric Power Applications*), vol. 128, pp. 293–305(12), November 1981. [Online]. Available: https://digital-library.theiet.org/content/journals/10.1049/ip-b.1981.0053

- [10] S. Buso, P. Mattavelli, L. Rossetto, and G. Spiazzi, "Simple digital control improving dynamic performance of power factor preregulators," *IEEE Transactions on Power Electronics*, vol. 13, no. 5, pp. 814–823, 1998.
- [11] A. Wu, J. Xiao, D. Markovic, and S. Sanders, "Digital PWM control: application in voltage regulation modules," in 30th Annual IEEE Power Electronics Specialists Conference. Record. (Cat. No.99CH36321), vol. 1, 1999, pp. 77–83 vol.1.
- [12] B. Patella, A. Prodic, A. Zirger, and D. Maksimovic, "High-frequency digital controller IC for DC/DC converters," in APEC. Seventeenth Annual IEEE Applied Power Electronics Conference and Exposition (Cat. No.02CH37335), vol. 1, 2002, pp. 374–380 vol.1.
- [13] Maksimovic, Zane, and Erickson, "Impact of digital control in power electronics," in 2004 Proceedings of the 16th International Symposium on Power Semiconductor Devices and ICs, 2004, pp. 13–22.
- [14] A. Syed, E. Ahmed, D. Maksimovic, and E. Alarcon, "Digital pulse width modulator architectures," in 2004 IEEE 35th Annual Power Electronics Specialists Conference (IEEE Cat. No.04CH37551), vol. 6, 2004, pp. 4689–4695 Vol.6.
- [15] S. Bowes, "New sinusoidal pulsewidth-modulated invertor," Proceedings of the Institution of Electrical Engineers, vol. 122, pp. 1279–1285(6), November 1975. [Online]. Available: https://digital-library.theiet.org/content/journals/10.1049/piee.1975.0312
- [16] D. M. Mitchell, "Pulsewidth modulator phase shift," *IEEE Transactions on Aerospace and Electronic Systems*, vol. AES-16, no. 3, pp. 272–278, 1980.
- [17] R. Middlebrook, "Predicting modulator phase lag in PWM converter feedback loops," in *Powercon*, vol. 8, 1981, p. H4.
- [18] G. Walker, "Digitally-implemented naturally sampled PWM suitable for multilevel converter control," *IEEE Transactions on Power Electronics*, vol. 18, no. 6, pp. 1322–1329, 2003.
- [19] L. Corradini and P. Mattavelli, "Analysis of multiple sampling technique for digitally controlled dc-dc converters," in 2006 37th IEEE Power Electronics Specialists Conference, 2006, pp. 1–6.
- [20] E. Tedeschi, P. Mattavelli, D. Trevisan, and L. Corradini, "Repetitive Ripple Estimation in Multisampling Digitally Controlled dc-dc Converters," in *IECON 2006 - 32nd Annual Conference on IEEE Industrial Electronics*, 2006, pp. 1685–1690.
- [21] L. Corradini, E. Tedeschi, and P. Mattavelli, "Advantages of the Symmetric-on Time Modulator in Multiple-Sampled Digitally Controlled DC-DC Converters," in 2007 IEEE Power Electronics Specialists Conference, 2007, pp. 1974–1980.

- [22] L. Corradini, P. Mattavelli, and S. Saggini, "Elimination of Sampling-Induced Dead Bands in Multiple-Sampled Pulse Width Modulators for DC-DC Converters," in *IECON 2007 - 33rd Annual Conference of the IEEE Industrial Electronics Society*, 2007, pp. 1495–1500.
- [23] L. Corradini, P. Mattavelli, and S. Saggini, "Elimination of Sampling-Induced Dead Bands in Multiple-Sampled Pulsewidth Modulators for DC–DC Converters," *IEEE Transactions on Power Electronics*, vol. 24, no. 11, pp. 2661–2665, 2009.
- [24] L. Corradini and P. Mattavelli, "Modeling of Multisampled Pulse Width Modulators for Digitally Controlled DC–DC Converters," *IEEE Transactions on Power Electronics*, vol. 23, no. 4, pp. 1839–1847, 2008.
- [25] L. Corradini, W. Stefanutti, and P. Mattavelli, "Analysis of Multisampled Current Control for Active Filters," *IEEE Transactions on Industry Applications*, vol. 44, no. 6, pp. 1785–1794, 2008.
- [26] L. Corradini, P. Mattavelli, E. Tedeschi, and D. Trevisan, "High-Bandwidth Multisampled Digitally Controlled DC–DC Converters Using Ripple Compensation," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 4, pp. 1501–1508, 2008.
- [27] L. Corradini, "Analysis and Implementation of Digital Control Architectures for DC-DC Switching Converters," *PhD Thesis*, 2008.
- [28] X. Zhang and J. W. Spencer, "Study of Multisampled Multilevel Inverters to Improve Control Performance," *IEEE Transactions on Power Electronics*, vol. 27, no. 11, pp. 4409–4416, 2012.
- [29] S. He, D. Zhou, X. Wang, Z. Zhao, and F. Blaabjerg, "A Review of Multisampling Techniques in Power Electronics Applications," *IEEE Transactions on Power Electronics*, vol. 37, no. 9, pp. 10514–10533, 2022.
- [30] J. Böcker and O. Buchholz, "Can oversampling improve the dynamics of PWM controls?" in 2013 IEEE International Conference on Industrial Technology (ICIT), 2013, pp. 1818–1824.
- [31] T. Mouton, A. de Beer, B. Putzeys, and B. McGrath, "Modelling and design of single-edge oversampled PWM current regulators using z-domain methods," in 2013 IEEE ECCE Asia Downunder, 2013, pp. 31–37.
- [32] H. d. T. Mouton, S. M. Cox, B. McGrath, L. Risbo, and B. Putzeys, "Small-Signal Analysis of Naturally-Sampled Single-Edge PWM Control Loops," *IEEE Transactions on Power Electronics*, vol. 33, no. 1, pp. 51–64, 2018.
- [33] R. Gupta, A. Ghosh, and A. Joshi, "Characteristic Analysis for Multisampled Digital Implementation of Fixed-Switching-Frequency Closed-Loop Modulation of Voltage-Source Inverter," *IEEE Transactions on Industrial Electronics*, vol. 56, no. 7, pp. 2382–2392, 2009.

- [34] S. He, D. Zhou, X. Wang, and F. Blaabjerg, "Aliasing Suppression of Multi-sampled Current Controlled LCL-Filtered Inverters," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, pp. 1–1, 2021.
- [35] T. Mouton and B. Putzeys, "Digital control of a PWM switching amplifier with global feedback," in Audio Engineering Society Conference: 37th International Conference: Class D Audio Amplification. Audio Engineering Society, 2009.
- [36] E. J. Burstinghaus, G. F. Ledwich, G. R. Walker, H. Pezeshki, and M. A. H. Broadmeadow, "Advanced resampling techniques for PWM amplifiers in real-time applications," in 2016 IEEE 2nd Annual Southern Power Electronics Conference (SPEC), 2016, pp. 1–6.
- [37] M. A. Broadmeadow, E. J. Burstinghaus, G. R. Walker, and G. F. Ledwich, "FPGA implementation of an arbitrary resample rate, FOH, pulse width modulator," *The Journal of Engineering*, vol. 2019, no. 17, pp. 3730–3735, 2019.
- [38] J. Yang, J. Liu, Y. Shi, N. Zhao, J. Zhang, L. Fu, and T. Q. Zheng, "Carrier-Based Digital PWM and Multirate Technique of a Cascaded H-Bridge Converter for Power Electronic Traction Transformers," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 7, no. 2, pp. 1207–1223, 2019.
- [39] H. Tian, Y. W. Li, and Q. Zhao, "Multirate Harmonic Compensation Control for Low Switching Frequency Converters: Scheme, Modeling, and Analysis," *IEEE Transactions on Power Electronics*, vol. 35, no. 4, pp. 4143–4156, 2020.
- [40] Y. He and Y. Wu, "A Universal Multi-Carrier Multisampling Method for Digitally-Controlled Inverters," *IEEE Transactions on Power Electronics*, pp. 1–11, 2022.
- [41] X. Zhang, P. Chen, C. Yu, F. Li, H. T. Do, and R. Cao, "Study of a Current Control Strategy Based on Multisampling for High-Power Grid-Connected Inverters With an LCL filter," *IEEE Transactions on Power Electronics*, vol. 32, no. 7, pp. 5023–5034, July 2017.
- [42] R. Zhang, C. Zhang, X. Xing, Z. Chen, and X. Liu, "Modeling and Control Method to Suppress Common Mode Resonance Circulating Current for High Power Parallel Three-Level Inverters System with Improved LCL Filter," *IEEE Transactions on Industrial Electronics*, pp. 1–1, 2022.
- [43] S. He, D. Zhou, X. Wang, and F. Blaabjerg, "Passivity-Based Multisampled Converter-Side Current Control of LCL-Filtered VSCs," *IEEE Transactions on Power Electronics*, vol. 37, no. 11, pp. 13848–13860, 2022.
- [44] M. Tomlinson, T. Mouton, and R. Kennel, "Finite-control-set model predictive control with a fixed switching frequency vs. linear control for current control of a single-leg inverter," in 2015 IEEE International Symposium on Predictive Control of Electrical Drives and Power Electronics (PRECEDE), 2015, pp. 109–114.

- [45] S. Buso, T. Caldognetto, and D. I. Brandao, "Comparison of oversampled current controllers for microgrid utility interface converters," in 2015 IEEE Energy Conversion Congress and Exposition (ECCE), 2015, pp. 6888–6895.
- [46] K. Miyata, K. Tsuchiya, and T. Yokoyama, "A study of 1MHz multi-sampling deadbeat control with disturbance compensation for PMSM drive system using FPGA," in 2016 IEEE 8th International Power Electronics and Motion Control Conference (IPEMC-ECCE Asia), 2016, pp. 654–659.
- [47] S. Buso, T. Caldognetto, and D. I. Brandao, "Dead-Beat Current Controller for Voltage-Source Converters With Improved Large-Signal Response," *IEEE Transactions on Industry Applications*, vol. 52, no. 2, pp. 1588–1596, 2016.
- [48] L. Rovere, A. Formentini, and P. Zanchetta, "FPGA Implementation of a Novel Oversampling Deadbeat Controller for PMSM Drives," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 5, pp. 3731–3741, 2019.
- [49] D. Hiroe, Z. Xiaohan, R. Suzuki, K. Nakamura, K. Sato, K. Yoshimoto, and T. Yokoyama, "A Study of 10MHz Multi-Sampling Deadbeat Control for PMSM Drive System using USPM Controller," in 2022 International Power Electronics Conference (IPEC-Himeji 2022- ECCE Asia), 2022, pp. 51–56.
- [50] S. Buso and T. Caldognetto, "A Nonlinear Wide-Bandwidth Digital Current Controller for DC-DC and DC-AC Converters," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 12, pp. 7687– 7695, 2015.
- [51] D. Holmes and B. McGrath, "Opportunities for harmonic cancellation with carrier-based PWM for a two-level and multilevel cascaded inverters," *IEEE Transactions on Industry Applications*, vol. 37, no. 2, pp. 574–582, 2001.
- [52] H. Fujita, "A Single-Phase Active Filter Using an H-Bridge PWM Converter With a Sampling Frequency Quadruple of the Switching Frequency," *IEEE Transactions on Power Electronics*, vol. 24, no. 4, pp. 934–941, 2009.
- [53] M. Odavic, V. Biagini, M. Sumner, P. Zanchetta, and M. Degano, "Multi-sampled carrier-based PWM for multilevel active shunt power filters for aerospace applications," in 2011 IEEE Energy Conversion Congress and Exposition, 2011, pp. 1483–1488.
- [54] S. Huang, L. Mathe, and R. Teodorescu, "A new method to implement resampled uniform PWM suitable for distributed control of modular multilevel converters," in *IECON 2013 - 39th Annual Conference of the IEEE Industrial Electronics Society*, 2013, pp. 228–233.
- [55] J. Ma, X. Wang, F. Blaabjerg, W. Song, S. Wang, and T. Liu, "Multisampling Method for Single-Phase Grid-Connected Cascaded H-Bridge Inverters," *IEEE Transactions on Industrial Electronics*, vol. 67, no. 10, pp. 8322–8334, 2020.

- [56] S. He, D. Zhou, X. Wang, and F. Blaabjerg, "Multisampling Control of Two-Cell Interleaved Three-phase Grid-connected Converters," in 2021 IEEE Applied Power Electronics Conference and Exposition (APEC), 2021, pp. 1432–1437.
- [57] G. Bonanno and L. Corradini, "Digital Predictive Current-Mode Control of Three-Level Flying Capacitor Buck Converters," *IEEE Transactions on Power Electronics*, vol. 36, no. 4, pp. 4697– 4710, 2021.
- [58] I. Z. Petric, R. K. Iyer, N. C. Brooks, and R. C. N. Pilawa-Podgurski, "A Real-Time Estimator for Capacitor Voltages in the Flying Capacitor Multilevel Converter," in 2022 IEEE 23rd Workshop on Control and Modeling for Power Electronics (COMPEL), 2022, pp. 1–8.
- [59] J. Dai, Y. Lang, B. Wu, D. Xu, and N. R. Zargari, "A Multisampling SVM Scheme for Current Source Converters With Superior Harmonic Performance," *IEEE Transactions on Power Electronics*, vol. 24, no. 11, pp. 2436–2445, 2009.
- [60] G. Oriti and A. L. Julian, "Three-Phase VSI with FPGA-Based Multisampled Space Vector Modulation," *IEEE Transactions on Industry Applications*, vol. 47, no. 4, pp. 1813–1820, 2011.
- [61] M. Bradley, E. Alarcón, and O. Feely, "Design-Oriented Analysis of Quantization-Induced Limit Cycles in a Multiple-Sampled Digitally Controlled Buck Converter," *IEEE Transactions on Circuits* and Systems I: Regular Papers, vol. 61, no. 4, pp. 1192–1205, 2014.
- [62] Z. Zhou, J. Wang, Z. Liu, and J. Liu, "Accurate Prediction of Vertical Crossings for Multi-Sampled Digital-Controlled Buck Converters," in 2020 IEEE Applied Power Electronics Conference and Exposition (APEC), 2020, pp. 292–298.
- [63] S. N. Vukosavić, L. S. Perić, and E. Levi, "AC Current Controller with Error-Free Feedback Acquisition System," *IEEE Transactions on Energy Conversion*, vol. 31, no. 1, pp. 381–391, 2016.
- [64] J. Yang, J. Liu, J. Zhang, N. Zhao, Y. Wang, and T. Q. Zheng, "Multirate Digital Signal Processing and Noise Suppression for Dual Active Bridge DC–DC Converters in a Power Electronic Traction Transformer," *IEEE Transactions on Power Electronics*, vol. 33, no. 12, pp. 10885–10902, 2018.
- [65] I. Z. Petric, P. Mattavelli, and S. Buso, "A Jitter Amplification Phenomenon in Multisampled Digital Control of Power Converters," *IEEE Transactions on Power Electronics*, vol. 36, no. 8, pp. 8685–8695, 2021.
- [66] —, "Investigation of Nonlinearities Introduced by Multi-sampled Pulsewidth Modulators," IEEE Transactions on Power Electronics, vol. 37, no. 3, pp. 2538–2550, 2022.
- [67] —, "Noise Attenuation Properties of Multisampled Control in Power Electronics," in 2020 IEEE 21st Workshop on Control and Modeling for Power Electronics (COMPEL), 2020, pp. 1–6.

- [68] —, "Feedback Noise Propagation in Multisampled DC–DC Power Electronic Converters," IEEE Transactions on Power Electronics, vol. 37, no. 1, pp. 150–161, 2022.
- [69] I. Z. Petric, R. Cvetanovic, P. Mattavelli, and S. Buso, "Models for Noise Propagation in Multi-Sampled Power Electronic Control Systems with Decimation," *IEEE Transactions on Power Electronics (under review)*, 2022.
- [70] I. Z. Petric, P. Mattavelli, and S. Buso, "Multi-Sampled Grid-Connected VSCs: A Path Toward Inherent Admittance Passivity," *IEEE Transactions on Power Electronics*, vol. 37, no. 7, pp. 7675– 7687, 2022.
- [71] —, "Passivation of Grid-Following VSCs: A Comparison Between Active Damping and Multisampled PWM," *IEEE Transactions on Power Electronics*, pp. 1–12, 2022.
- [72] I. Petric, R. Cvetanovic, P. Mattavelli, S. Buso, and S. Vukosavic, "Analysis and DSP Implementation of Multi-sampled Three-Phase Current Controllers," in 2021 21st International Symposium on Power Electronics (Ee), 2021, pp. 1–7.
- [73] R. Cvetanovic, I. Z. Petric, P. Mattavelli, and S. Buso, "Accurate High-Frequency Modeling of the Input Admittance of PWM Grid-Connected VSCs," *IEEE Transactions on Power Electronics*, vol. 37, no. 9, pp. 10534–10545, 2022.
- [74] —, "Small-Signal Modeling of Phase-Shifted Digital PWM in Interleaved and Multilevel Converters," *IEEE Transactions on Power Electronics*, pp. 1–12, 2022.
- [75] R. W. Erickson and D. Maksimovic, Fundamentals of power electronics. Springer Science & Business Media, 2007.
- [76] J. Holtz, W. Lotzkat, and K.-H. Werner, "A high-power multi transistor-inverter uninterruptable power supply system," in 1986 17th Annual IEEE Power Electronics Specialists Conference, 1986, pp. 311–320.
- [77] G. Carrara, S. Gardella, M. Marchesoni, R. Salutari, and G. Sciutto, "A new multilevel PWM method: a theoretical analysis," *IEEE Transactions on Power Electronics*, vol. 7, no. 3, pp. 497– 505, 1992.
- [78] B. McGrath and D. Holmes, "Multicarrier PWM strategies for multilevel inverters," *IEEE Trans*actions on Industrial Electronics, vol. 49, no. 4, pp. 858–867, 2002.
- [79] H. S. Black, Modulation theory. van Nostrand, 1953.
- [80] S. Bowes and B. Bird, "Novel approach to the analysis and synthesis of modulation processes in power convertors," *Proceedings of the Institution of Electrical Engineers*, vol. 122, pp. 507–513(6), May 1975. [Online]. Available: https://digital-library.theiet.org/content/journals/10.1049/piee. 1975.0141

- [81] D. Holmes, "A general analytical method for determining the theoretical harmonic components of carrier based PWM strategies," in *Conference Record of 1998 IEEE Industry Applications Conference. Thirty-Third IAS Annual Meeting (Cat. No.98CH36242)*, vol. 2, 1998, pp. 1207–1214 vol.2.
- [82] H. Peng, A. Prodic, E. Alarcon, and D. Maksimovic, "Modeling of Quantization Effects in Digitally Controlled DC–DC Converters," *IEEE Transactions on Power Electronics*, vol. 22, no. 1, pp. 208– 215, 2007.
- [83] A. V. Peterchev and S. R. Sanders, "Quantization resolution and limit cycling in digitally controlled PWM converters," *IEEE Transactions on Power Electronics*, vol. 18, no. 1, pp. 301–308, 2003.
- [84] S. Saggini, W. Stefanutti, D. Trevisan, P. Mattavelli, and G. Garcea, "Prediction of limit-cycles oscillations in digitally controlled DC-DC converters using statistical approach," in 31st Annual Conference of IEEE Industrial Electronics Society, 2005. IECON 2005., 2005, pp. 6 pp.–.
- [85] B. Patella, A. Prodic, A. Zirger, and D. Maksimovic, "High-frequency digital PWM controller IC for DC-DC converters," *IEEE Transactions on Power Electronics*, vol. 18, no. 1, pp. 438–446, 2003.
- [86] A. Peterchev, J. Xiao, and S. Sanders, "Architecture and IC implementation of a digital VRM controller," *IEEE Transactions on Power Electronics*, vol. 18, no. 1, pp. 356–364, 2003.
- [87] P. Mattavelli, F. Polo, F. Dal Lago, and S. Saggini, "Analysis of Control-Delay Reduction for the Improvement of UPS Voltage-Loop Bandwidth," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 8, pp. 2903–2911, 2008.
- [88] A. R. Brown and R. Middlebrook, "Sampled-data modeling of switching regulators," in 1981 IEEE Power Electronics Specialists Conference, 1981, pp. 349–369.
- [89] D. Van de Sype, K. De Gusseme, A. Van den Bossche, and J. Melkebeek, "Small-signal Laplacedomain analysis of uniformly-sampled pulse-width modulators," in 2004 IEEE 35th Annual Power Electronics Specialists Conference (IEEE Cat. No.04CH37551), vol. 6, 2004, pp. 4292–4298 Vol.6.
- [90] —, "Small-signal z-domain analysis of digitally controlled converters," in 2004 IEEE 35th Annual Power Electronics Specialists Conference (IEEE Cat. No.04CH37551), vol. 6, 2004, pp. 4299–4305 Vol.6.
- [91] W. Qiu, G. Miller, and Z. Liang, "Dual-Edge Pulse Width Modulation Scheme for Fast Transient Response of Multiple-Phase Voltage Regulators," in 2007 IEEE Power Electronics Specialists Conference, 2007, pp. 1563–1569.
- [92] A. Comacchio, G. Bonanno, H. Abedini, P. Mattavelli, and M. Corradin, "Asymmetric Digital Dual-Edge Modulator For Dynamic Performance Improvement Of Multi-Loop Controlled VSI," *IEEE Transactions on Industrial Electronics*, pp. 1–10, 2022.

- [93] H. Kim, M. W. Degner, J. M. Guerrero, F. Briz, and R. D. Lorenz, "Discrete-Time Current Regulator Design for AC Machine Drives," *IEEE Transactions on Industry Applications*, vol. 46, no. 4, pp. 1425–1435, 2010.
- [94] I. Z. Petric, S. N. Vukosavic, M. Degano, and A. Galassini, "A Digital Internal Model Current Controller for Salient Machines," *IEEE Transactions on Industrial Electronics*, vol. 68, no. 6, pp. 4703–4717, 2021.
- [95] S. Buso, L. Malesani, and P. Mattavelli, "Comparison of current control techniques for active filter applications," *IEEE Transactions on Industrial Electronics*, vol. 45, no. 5, pp. 722–729, 1998.
- [96] P. Mattavelli, "An improved deadbeat control for UPS using disturbance observers," *IEEE Trans*actions on Industrial Electronics, vol. 52, no. 1, pp. 206–212, 2005.
- [97] K. Yao, Y. Meng, P. Xu, and F. Lee, "Design considerations for VRM transient response based on the output impedance," in APEC. Seventeenth Annual IEEE Applied Power Electronics Conference and Exposition (Cat. No.02CH37335), vol. 1, 2002, pp. 14–20 vol.1.
- [98] G. Liu, P. Mattavelli, and S. Saggini, "Resistive–Capacitive Output Impedance Shaping for Droop-Controlled Converters in DC Microgrids With Reduced Output Capacitance," *IEEE Transactions* on Power Electronics, vol. 35, no. 6, pp. 6501–6511, 2020.
- [99] R. D. Middlebrook and S. Cuk, "A general unified approach to modelling switching-converter power stages," in 1976 IEEE Power Electronics Specialists Conference, 1976, pp. 18–34.
- [100] T. Nussbaumer, M. L. Heldwein, G. Gong, S. D. Round, and J. W. Kolar, "Comparison of Prediction Techniques to Compensate Time Delays Caused by Digital Control of a Three-Phase Buck-Type PWM Rectifier System," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 2, pp. 791–799, 2008.
- [101] J. Holtz, "Pulsewidth modulation for electronic power conversion," *Proceedings of the IEEE*, vol. 82, no. 8, pp. 1194–1214, 1994.
- [102] M. Kazmierkowski and L. Malesani, "Current control techniques for three-phase voltage-source PWM converters: a survey," *IEEE Transactions on Industrial Electronics*, vol. 45, no. 5, pp. 691–703, 1998.
- [103] G. Walker and G. Ledwich, "Bandwidth considerations for multilevel converters," *IEEE Transac*tions on Power Electronics, vol. 14, no. 1, pp. 74–81, 1999.
- [104] I. Deslauriers, N. Avdiu, and B. T. Ooi, "Naturally sampled triangle carrier PWM bandwidth limit and output spectrum," *IEEE Transactions on Power Electronics*, vol. 20, no. 1, pp. 100–106, 2005.

- [105] L. Marco, A. Poveda, E. Alarcon, and D. Maksimovic, "Bandwidth limits in PWM switching amplifiers," in 2006 IEEE International Symposium on Circuits and Systems, 2006, pp. 4 pp.– 5326.
- [106] R. Gupta, A. Ghosh, and A. Joshi, "Switching Characterization of Cascaded Multilevel-Inverter-Controlled Systems," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 3, pp. 1047–1058, 2008.
- [107] J. Carson, "Notes on the Theory of Modulation," Proceedings of the Institute of Radio Engineers, vol. 10, no. 1, pp. 57–64, 1922.
- [108] P. Enjeti, P. Ziogas, and J. Lindsay, "Programmed PWM techniques to eliminate harmonics: a critical evaluation," *IEEE Transactions on Industry Applications*, vol. 26, no. 2, pp. 302–316, 1990.
- [109] C. Restrepo, T. Konjedic, F. Flores-Bahamonde, E. Vidal-Idiarte, J. Calvente, and R. Giral, "Multisampled Digital Average Current Controls of the Versatile Buck–Boost Converter," *IEEE Journal* of Emerging and Selected Topics in Power Electronics, vol. 7, no. 2, pp. 879–890, 2019.
- [110] S. N. Vukosavić, L. S. Perić, and E. Levi, "A Three-Phase Digital Current Controller With Improved Performance Indices," *IEEE Transactions on Energy Conversion*, vol. 32, no. 1, pp. 184–193, 2017.
- [111] S. N. Vukosavic, L. S. Peric, and E. Levi, "Digital Current Controller With Error-Free Feedback Acquisition and Active Resistance," *IEEE Transactions on Industrial Electronics*, vol. 65, no. 3, pp. 1980–1990, 2018.
- [112] T. Caldognetto, "Control of Electronic Power Converters for Low-Voltage Microgrids," *PhD Thesis*, 2016.
- [113] T. Sakharuk, B. Lehman, A. Stankovic, and G. Tadmor, "Effects of finite switching frequency and delay on PWM controlled systems," *IEEE Transactions on Circuits and Systems I: Fundamental Theory and Applications*, vol. 47, no. 4, pp. 555–567, 2000.
- [114] X. Li, X. Ruan, Q. Jin, M. Sha, and C. K. Tse, "Small-Signal Models With Extended Frequency Range for DC–DC Converters With Large Modulation Ripple Amplitude," *IEEE Transactions on Power Electronics*, vol. 33, no. 9, pp. 8151–8163, 2018.
- [115] D. Zmood and D. Holmes, "Stationary frame current regulation of PWM inverters with zero steady-state error," *IEEE Transactions on Power Electronics*, vol. 18, no. 3, pp. 814–822, 2003.
- [116] D. G. Holmes, T. A. Lipo, B. P. McGrath, and W. Y. Kong, "Optimized Design of Stationary Frame Three Phase AC Current Regulators," *IEEE Transactions on Power Electronics*, vol. 24, no. 11, pp. 2417–2426, 2009.

- [117] S. Buso, T. Caldognetto, and Q. Liu, "Analysis and Experimental Characterization of a Large-Bandwidth Triple-Loop Controller for Grid-Tied Inverters," *IEEE Transactions on Power Electronics*, vol. 34, no. 2, pp. 1936–1949, 2019.
- [118] J. Houston, W. Qiu, and C. Cheung, "A new large signal average model for variable frequency pulsewidth modulators," in 2013 Twenty-Eighth Annual IEEE Applied Power Electronics Conference and Exposition (APEC), 2013, pp. 1738–1745.
- [119] G. Ripamonti, S. Saggini, L. Corradini, R. Rizzolatti, F. Faccio, S. Michelis, A. Koukab, and M. Kayal, "A Dual-Edge Pulsewidth Modulator for Fast Dynamic Response DC–DC Converters," *IEEE Transactions on Power Electronics*, vol. 34, no. 1, pp. 28–32, 2019.
- [120] S. V. Vaseghi, Advanced digital signal processing and noise reduction. John Wiley & Sons, 2008.
- [121] A. G. Yepes, F. D. Freijedo, J. Doval-Gandoy, s. López, J. Malvar, and P. Fernandez-Comesaña, "Effects of Discretization Methods on the Performance of Resonant Controllers," *IEEE Transactions on Power Electronics*, vol. 25, no. 7, pp. 1692–1712, 2010.
- [122] B. Weber, K. Wiedmann, and A. Mertens, "Increased signal-to-noise ratio of sensorless control using current oversampling," in 2015 9th International Conference on Power Electronics and ECCE Asia (ICPE-ECCE Asia), 2015, pp. 1129–1134.
- [123] B. Weber, T. Brandt, and A. Mertens, "Compensation of switching dead-time effects in voltage-fed PWM inverters using FPGA-based current oversampling," in 2016 IEEE Applied Power Electronics Conference and Exposition (APEC), 2016, pp. 3172–3179.
- [124] S.-H. Song, J.-W. Choi, and S.-K. Sul, "Current measurements in digitally controlled AC drives," *IEEE Industry Applications Magazine*, vol. 6, no. 4, pp. 51–62, 2000.
- [125] L. Saunders, G. Skibinski, S. Evon, and D. Kempkes, "Riding the reflected wave-IGBT drive technology demands new motor and cable considerations," in *Proceedings of 1996 IAS Petroleum* and Chemical Industry Technical Conference, 1996, pp. 75–84.
- [126] S. Amarir and K. Al-Haddad, "A Modeling Technique to Analyze the Impact of Inverter Supply Voltage and Cable Length on Industrial Motor-Drives," *IEEE Transactions on Power Electronics*, vol. 23, no. 2, pp. 753–762, 2008.
- [127] W. A. Gardner, "Introduction to random processes with applications to signals and systems," New York, MacMillan Co., 1986, 447, 1986.
- [128] J. Palmer, S. Ji, X. Huang, L. Zhang, W. Giewont, F. F. Wang, and L. M. Tolbert, "Improving Voltage Sensor Noise Immunity in a High Voltage and High dv/dt Environment," in 2020 IEEE Applied Power Electronics Conference and Exposition (APEC), 2020, pp. 107–113.

- [129] W. A. Gardner, "The spectral correlation theory of cyclostationary time-series," Signal Processing, vol. 11, no. 1, pp. 13–36, 1986. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/0165168486900927
- [130] W. Gardner, "Common Pitfalls in the Application of Stationary Process Theory to Time-Sampled and Modulated Signals," *IEEE Transactions on Communications*, vol. 35, no. 5, pp. 529–534, 1987.
- [131] V. P. Sathe and P. P. Vaidyanathan, "Effects of Multirate Systems on the Statistical Properties of Random Signals," *IEEE Transactions on Signal Processing*, vol. 41, pp. 131–146, 1993.
- [132] F. Blaabjerg, R. Teodorescu, M. Liserre, and A. Timbus, "Overview of Control and Grid Synchronization for Distributed Power Generation Systems," *IEEE Transactions on Industrial Electronics*, vol. 53, no. 5, pp. 1398–1409, 2006.
- [133] X. Wang and F. Blaabjerg, "Harmonic Stability in Power Electronic-Based Power Systems: Concept, Modeling, and Analysis," *IEEE Transactions on Smart Grid*, vol. 10, no. 3, pp. 2858–2870, 2019.
- [134] L. Harnefors, X. Wang, A. G. Yepes, and F. Blaabjerg, "Passivity-Based Stability Assessment of Grid-Connected VSCs—An Overview," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 4, no. 1, pp. 116–125, 2016.
- [135] L. Harnefors, A. G. Yepes, A. Vidal, and J. Doval-Gandoy, "Passivity-Based Controller Design of Grid-Connected VSCs for Prevention of Electrical Resonance Instability," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 2, pp. 702–710, 2015.
- [136] R. Middlebrook, "Input filter considerations in design and application of switching regulators." 1976.
- [137] J. Sun, "Impedance-Based Stability Criterion for Grid-Connected Inverters," *IEEE Transactions on Power Electronics*, vol. 26, no. 11, pp. 3075–3078, 2011.
- [138] A. Riccobono and E. Santi, "Comprehensive Review of Stability Criteria for DC Power Distribution Systems," *IEEE Transactions on Industry Applications*, vol. 50, no. 5, pp. 3525–3535, 2014.
- [139] B. Wen, D. Dong, D. Boroyevich, R. Burgos, P. Mattavelli, and Z. Shen, "Impedance-Based Analysis of Grid-Synchronization Stability for Three-Phase Paralleled Converters," *IEEE Transactions* on Power Electronics, vol. 31, no. 1, pp. 26–38, 2016.
- [140] B. Wen, D. Boroyevich, R. Burgos, P. Mattavelli, and Z. Shen, "Analysis of D-Q Small-Signal Impedance of Grid-Tied Inverters," *IEEE Transactions on Power Electronics*, vol. 31, no. 1, pp. 675–687, 2016.

- [141] L. Harnefors, R. Finger, X. Wang, H. Bai, and F. Blaabjerg, "VSC Input-Admittance Modeling and Analysis Above the Nyquist Frequency for Passivity-Based Stability Assessment," *IEEE Transactions on Industrial Electronics*, vol. 64, no. 8, pp. 6362–6370, 2017.
- [142] X. Wang, L. Harnefors, and F. Blaabjerg, "Unified Impedance Model of Grid-Connected Voltage-Source Converters," *IEEE Transactions on Power Electronics*, vol. 33, no. 2, pp. 1775–1787, 2018.
- [143] F. D. Freijedo, M. Ferrer, and D. Dujic, "Multivariable High-Frequency Input-Admittance of Grid-Connected Converters: Modeling, Validation, and Implications on Stability," *IEEE Transactions* on Industrial Electronics, vol. 66, no. 8, pp. 6505–6515, 2019.
- [144] V. Pirsto, J. Kukkola, M. Hinkkanen, and L. Harnefors, "Intersample Modeling of the Converter Output Admittance," *IEEE Transactions on Industrial Electronics*, vol. 68, no. 11, pp. 11348– 11358, 2021.
- [145] J. Sun, "Two-Port Characterization and Transfer Immittances of AC-DC Converters-Part I: Modeling," *IEEE Open Journal of Power Electronics*, vol. 2, pp. 440–462, 2021.
- [146] —, "Two-Port Characterization and Transfer Immittances of AC-DC Converters-Part II: Applications," *IEEE Open Journal of Power Electronics*, vol. 2, pp. 483–510, 2021.
- [147] —, "Frequency-Domain Stability Criteria for Converter-Based Power Systems," IEEE Open Journal of Power Electronics, vol. 3, pp. 222–254, 2022.
- [148] E. Rodriguez-Diaz, F. D. Freijedo, J. M. Guerrero, J.-A. Marrero-Sosa, and D. Dujic, "Input-Admittance Passivity Compliance for Grid-Connected Converters With an LCL Filter," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 2, pp. 1089–1097, 2019.
- [149] F. Hans, W. Schumacher, S.-F. Chou, and X. Wang, "Passivation of Current-Controlled Grid-Connected VSCs Using Passivity Indices," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 11, pp. 8971–8980, 2019.
- [150] J. Serrano-Delgado, S. Cobreces, M. Rizo, and E. J. Bueno, "Low-Order Passivity-Based Robust Current Control Design for Grid-Tied VSCs," *IEEE Transactions on Power Electronics*, vol. 36, no. 10, pp. 11886–11899, 2021.
- [151] S. Sanders and G. Verghese, "Lyapunov-based control for switched power converters," *IEEE Trans*actions on Power Electronics, vol. 7, no. 1, pp. 17–24, 1992.
- [152] "Railway Applications—Power Supply and Rolling Stock—Technical Criteria for the Coordination Between Power Supply (Substation) and Rolling Stock to Achieve Interoperability, CENELEC Std. EN50388, 2012."

- [153] X. Wang, F. Blaabjerg, and P. C. Loh, "Passivity-Based Stability Analysis and Damping Injection for Multiparalleled VSCs with LCL Filters," *IEEE Transactions on Power Electronics*, vol. 32, no. 11, pp. 8922–8935, 2017.
- [154] H. Yu, M. A. Awal, H. Tu, Y. Du, S. Lukic, and I. Husain, "Passivity-Oriented Discrete-Time Voltage Controller Design for Grid-Forming Inverters," in 2019 IEEE Energy Conversion Congress and Exposition (ECCE), 2019, pp. 469–475.
- [155] H. Wu and X. Wang, "Virtual-Flux-Based Passivation of Current Control for Grid-Connected VSCs," *IEEE Transactions on Power Electronics*, vol. 35, no. 12, pp. 12673–12677, 2020.
- [156] M. A. Awal, W. Yu, and I. Husain, "Passivity-Based Predictive-Resonant Current Control for Resonance Damping in LCL-Equipped VSCs," *IEEE Transactions on Industry Applications*, vol. 56, no. 2, pp. 1702–1713, 2020.
- [157] A. Akhavan, H. R. Mohammadi, J. C. Vasquez, and J. M. Guerrero, "Passivity-Based Design of Plug-and-Play Current-Controlled Grid-Connected Inverters," *IEEE Transactions on Power Electronics*, vol. 35, no. 2, pp. 2135–2150, 2020.
- [158] Z. Yang, C. Shah, T. Chen, J. Teichrib, and R. W. De Doncker, "Virtual Damping Control Design of Three-Phase Grid-Tied PV Inverters for Passivity Enhancement," *IEEE Transactions on Power Electronics*, vol. 36, no. 6, pp. 6251–6264, 2021.
- [159] M. A. Awal, L. D. Flora, and I. Husain, "Observer Based Generalized Active Damping for Voltage Source Converters With LCL Filters," *IEEE Transactions on Power Electronics*, vol. 37, no. 1, pp. 125–136, 2022.
- [160] J. Zhao, C. Xie, K. Li, J. Zou, and J. M. Guerrero, "Passivity-Oriented Design of LCL-Type Grid-Connected Inverters With Luenberger Observer-Based Active Damping," *IEEE Transactions on Power Electronics*, vol. 37, no. 3, pp. 2625–2635, 2022.
- [161] C. Xie, K. Li, J. Zou, and J. M. Guerrero, "Passivity-Based Stabilization of LCL-Type Grid-Connected Inverters via a General Admittance Model," *IEEE Transactions on Power Electronics*, vol. 35, no. 6, pp. 6636–6648, 2020.
- [162] Y. Liao, X. Wang, and F. Blaabjerg, "Passivity-Based Analysis and Design of Linear Voltage Controllers For Voltage-Source Converters," *IEEE Open Journal of the Industrial Electronics Society*, vol. 1, pp. 114–126, 2020.
- [163] S. Li and H. Lin, "A Capacitor-Current-Feedback Positive Active Damping Control Strategy for LCL-Type Grid-Connected Inverter to Achieve High Robustness," *IEEE Transactions on Power Electronics*, pp. 1–1, 2021.

- [164] X. Wang, Y. He, D. Pan, H. Zhang, Y. Ma, and X. Ruan, "Passivity Enhancement for LCL-Filtered Inverter With Grid Current Control and Capacitor Current Active Damping," *IEEE Transactions* on Power Electronics, vol. 37, no. 4, pp. 3801–3812, 2022.
- [165] X. Wang, F. Blaabjerg, and W. Wu, "Modeling and Analysis of Harmonic Stability in an AC Power-Electronics-Based Power System," *IEEE Transactions on Power Electronics*, vol. 29, no. 12, pp. 6421–6432, 2014.
- [166] X. Yue, F. Zhuo, S. Yang, Y. Pei, and H. Yi, "A Matrix-Based Multifrequency Output Impedance Model for Beat Frequency Oscillation Analysis in Distributed Power Systems," *IEEE Journal of Emerging and Selected Topics in Power Electronics*, vol. 4, no. 1, pp. 80–92, 2016.
- [167] D. Yang, X. Wang, and F. Blaabjerg, "Sideband Harmonic Instability of Paralleled Inverters With Asynchronous Carriers," *IEEE Transactions on Power Electronics*, vol. 33, no. 6, pp. 4571–4577, 2018.
- [168] Q. Liu, Y. Ying, and M. Wu, "Extended Harmonic Resonance Analysis of Grid-Connected Converters Considering the Frequency Coupling Effect," *IEEE Transactions on Industrial Electronics*, vol. 69, no. 9, pp. 9353–9363, 2022.
- [169] J. Dannehl, M. Liserre, and F. W. Fuchs, "Filter-Based Active Damping of Voltage Source Converters With LCL Filter," *IEEE Transactions on Industrial Electronics*, vol. 58, no. 8, pp. 3623–3633, 2011.
- [170] W. Yao, Y. Yang, Y. Xu, F. Blaabjerg, S. Liu, and G. Wilson, "Phase Reshaping via All-Pass Filters for Robust LCL-Filter Active Damping," *IEEE Transactions on Power Electronics*, vol. 35, no. 3, pp. 3114–3126, 2020.
- [171] A. Akhavan, J. C. Vasquez, and J. M. Guerrero, "A Simple Method for Passivity Enhancement of Current Controlled Grid-Connected Inverters," *IEEE Transactions on Power Electronics*, vol. 35, no. 8, pp. 7735–7741, 2020.
- [172] L. Yang, L. Zhao, X. Chen, Z. Zhang, H. Nian, J. Zhao, R. Deng, and L. Yan, "Robust Active Damping Control for LCL-Type Shunt Active Power Filters," *IEEE Access*, vol. 10, pp. 39456– 39470, 2022.
- [173] S. G. Parker, B. P. McGrath, and D. G. Holmes, "Regions of Active Damping Control for LCL Filters," *IEEE Transactions on Industry Applications*, vol. 50, no. 1, pp. 424–432, 2014.
- [174] M. Berg and T. Roinila, "Nonlinear Effect of Dead Time in Small-Signal Modeling of Power-Electronic System Under Low-Load Conditions," *IEEE Journal of Emerging and Selected Topics* in Power Electronics, vol. 8, no. 4, pp. 3204–3213, 2020.

[175] S. K. Mitra and Y. Kuo, Digital signal processing: a computer-based approach. McGraw-Hill New York, 2006, vol. 2.